7-11-2018

Optimal maintenance planning using reliability information for offshore wind turbines

Mark Robert Haring Industrial Engineering and Management UNIVERSITY OF TWENTE

Preface

This thesis is written in order to complete the bachelor Industrial Engineering and Management at the University of Twente. The assignment is performed under supervision of Fraunhofer IEE and the University of Twente.

I would like to express my gratitude towards my supervisors of the University of Twente, Engin Topan and Ipek Seyran Topan, for the good guidance and all the valuable feedback during the execution of my bachelor thesis. My first supervisor, Engin Topan, was always available on short term if I had any questions or concerns about my research. During the meetings, we had some good and interesting discussions concerning the content and direction of the bachelor thesis. Furthermore, during the meetings and the visits to Fraunhofer IEE in Germany, we had some fun and interesting conversations on a broad variety of subjects.

Moreover, I would like to thank Fraunhofer IEE for the possibility to execute my bachelor thesis for their company. The knowledge and expertise shared by my supervisors at Fraunhofer IEE contributed greatly to a good underpinned research. Thus, I would also like to express my gratitude towards my supervisors at Fraunhofer IEE, Stefan Faulstisch and Alexander Lutz, for their valuable feedback and comments.

Finally, I would like to thank my family and friends which supported me during the execution of the bachelor thesis. They provided me with interesting thoughts and ideas which I could potentially implement in my research. If I had any struggles during my bachelor thesis, the motivational words of my family and friends would help me to continue and overcome these struggles.

With kind regards, Mark Haring Enschede, November 7, 2018.

Management summary

Overview

Fraunhofer IEE is conducting research for the national and international transformation of energy supply systems. They develop solutions for technical and economic challenges in order to further reduce the costs of using renewable energies, to secure the supply despite volatile generation, to ensure grid stability at the usual high level and to make the business model of the energy transition a success.

As the global operational wind turbine fleet ages and a trend towards self-operations is observed by wind farm operators, research into reducing the levelized cost of electricity continues. Operation and maintenance costs have a substantial contribution to the levelized cost of electricity and 67% of the operation and maintenance costs are caused by the high maintenance costs (IRENA, Renewable Power Generation Costs in 2017, 2018). Furthermore, offshore wind turbines operate in a harsher environment which makes it harder and costlier to execute maintenance. Therefore, high downtimes and high maintenance costs can be observed which inadvertently results in a higher levelized cost of electricity.

With reliability information provided by Fraunhofer, we have analyzed and proposed an optimal maintenance policy for a single component which minimizes maintenance cost and thereby ensures reliability. The component chosen for this research purpose is the converter system. The converter system fails frequently and has a high share on the annual downtimes. Furthermore, the optimal preventive replacement time of a rotor system in an offshore wind turbine has been analyzed and proposed.

Approach

The used optimal maintenance policy is an age-based preventive maintenance policy. This policy schedules preventive maintenance after fixed operational time which minimizes the cost per unit time. The optimum can be found at the lowest value of the objective function. The objective function is determined from the expected cost per cycle and the expected cycle length. The expected cost per cycle is calculated from the cost of preventive and corrective maintenance and the reliability information of the component.

Preventive maintenance involves the maintenance activities prior to a failure. In our study, preventive maintenance will consist of the replacement of the converter system. Corrective maintenance involves the maintenance activities after a failure has been detected. Both costs of preventive and corrective maintenance are determined from existing literature.

The provided reliability information is analyzed and modeled in excel. The reliability information consists of 7 failure categories from which the reliability of the converter system can be determined. The wind speed data of FINO1 location in the North Sea has been used to determine two of the failure categories. All failure categories are modelled hourly over a 25 year-span. Thereafter, the reliability for each quarter of a year is determined. From the reliability, the unreliability is easily calculated, and the expected cycle length can be determined.

Furthermore, the optimal fixed operational time is analyzed by means of a sensitivity analysis executed on the maintenance costs. Additionally, the work is extended to the rotor systems of offshore wind turbines.

Key findings

- The designed optimal maintenance planning model can determine the optimal preventive replacement time for numerous wind turbine components, using different cost parameters inputs and reliability information. The designed model minimizes the total downtime, maintenance and logistics costs.
- For a preventive maintenance cost of 541.213€ and corrective maintenance cost of 1.230.654€ (CM/PM cost ratio of 2,27), an effectiveness of 0,24% is found of using an age-based maintenance policy to schedule the preventive replacement of the converter system. The converter should be preventively replaced after 21,75 operational years with a cost per unit time of 56.545€.
- For the rotor system, the optimal preventive replacement time is found to be after 9,5 operational years with a cost per unit time of 189.052€. The used preventive maintenance cost is 924.691€ and corrective maintenance cost of 1.849.382€ (CM/PM cost ratio of 2). Operators can save 296.575€ in rotor system maintenance costs over the lifetime of one wind turbine. E.g. for a wind farm with 50 wind turbines, operators can save approximately €15 million in rotor system maintenance costs.
- The optimal preventive replacement time is strongly dependent on the cost ratio between preventive and corrective maintenance cost. The optimums found for the converter system range between 25 years for a CM/PM cost ratio of 2 and 13,25 years for a ratio of 6.
- Determining the optimal for all major components can yield optimums that lay close to each other and yield the possibility to perform opportunistic maintenance. Combining preventive maintenance actions can even further reduce the total maintenance cost incurred over the lifetime of a wind turbine.

Recommendations

For further research:

- 1. The optimal maintenance planning model should be extended with other seasonal changes such as more variability in the cost determinations, waiting times, and capacity factor.
- 2. Further research the possibility to use k-mean clustering to find more suitable optimums for more than one component.
- 3. Execute a sensitivity analysis on the influence of the reliability of a component on the optimal fixed operational time.
- 4. Analyze different optimums of major components to seek the opportunity to perform opportunistic maintenance.
- 5. Research the possibility to implement condition-based maintenance using condition monitoring systems in wind turbines.

The implementations of the above recommendations will further improve the accuracy of the optimal maintenance planning model and consequently the value derived from it. Furthermore,

condition-based maintenance should be further researched as it can minimize the cost of preventive and corrective maintenance actions.

Management samenvatting

Overzicht

Fraunhofer IEE doet onderzoek voor de nationale en internationale transformatie van energievoorzieningssystemen. Ze ontwikkelen oplossingen voor technische en economische uitdagingen om zodoende de kosten van het gebruik van hernieuwbare energiebronnen verder te verlagen, het aanbod veilig te stellen ondanks volatiele opwekking, om de netstabiliteit op het gebruikelijke hoge niveau te verzekeren en om het bedrijfsmodel van de energietransitie tot een succes te maken.

Naarmate de wereldwijde operationele windturbinevloot ouder wordt en er een zelfoperatie trend wordt waargenomen door exploitanten van windparken, wordt het onderzoek naar de verlaging van de genivelleerde kosten van elektriciteit voortgezet. Exploitatie- en onderhoudskosten dragen substantieel bij aan de genivelleerde kosten van elektriciteit en 67% van de exploitatie- en onderhoudskosten worden veroorzaakt door de hoge onderhoudskosten (IRENA, hernieuwbare stroomproductiekosten in 2017, 2018). Bovendien werken offshore windturbines in een ruwere omgeving, waardoor het moeilijker en duurder wordt om onderhoud uit te voeren. Daarom kunnen hoge stilstand tijden en hoge onderhoudskosten worden waargenomen die onbedoeld leiden tot hogere genivelleerde elektriciteitskosten.

Met betrouwbaarheidsinformatie van Fraunhofer hebben we een optimaal onderhoudsbeleid voor één onderdeel geanalyseerd en ontwikkeld, waardoor de onderhoudskosten tot een minimum worden beperkt en de betrouwbaarheid wordt gegarandeerd. De gekozen component voor dit onderzoeksdoel is het elektrisch omzettersysteem. Het elektrisch omzettersysteem faalt regelmatig en heeft een hoog aandeel in de stilstand tijden.

Aanpak

Het gebruikte optimale onderhoudsbeleid is een, op leeftijd gebaseerd, preventief onderhoudsbeleid. Het beleid plant preventief onderhoud na vaste operationele tijd, waardoor de kosten per tijdseenheid tot een minimum worden beperkt. Het optimum kan worden gevonden bij de laagste waarde van de doelfunctie. De doelfunctie wordt bepaald aan de hand van de verwachte kosten per cyclus en de verwachte cycluslengte. De verwachte kosten per cyclus worden berekend op basis van de kosten van preventief en correctief onderhoud en de betrouwbaarheidsinformatie van het onderdeel.

Preventief onderhoud omvat de onderhoudswerkzaamheden voorafgaand aan het falen van een onderdeel. In onze studie zal preventief onderhoud bestaan uit het vervangen van het elektrisch omzettersysteem. Correctief onderhoud omvat de onderhoudswerkzaamheden nadat het falen van een component is gedetecteerd. Beide kosten van preventief en correctief onderhoud worden bepaald op basis van bestaande literatuur.

De verstrekte betrouwbaarheidsinformatie wordt geanalyseerd en gemodelleerd in Excel. De betrouwbaarheidsinformatie bestaat uit 7 faalcategorieën waaruit de betrouwbaarheid van het elektrisch omzettersysteem kan worden bepaald. De windsnelheid gegevens van de FINO1-locatie in de Noordzee zijn gebruikt om twee van de faalcategorieën te bepalen. Alle faalcategorieën zijn per uur gemodelleerd over een periode van 25 jaar. Daarna wordt de betrouwbaarheid voor elk kwartaal van een jaar bepaald. Van de betrouwbaarheid kan de onbetrouwbaarheid gemakkelijk worden berekend en kan de verwachte cycluslengte worden bepaald. Verder wordt de optimale vaste operationele tijd om een component te vervangen geanalyseerd doormiddel van een gevoeligheidsanalyse uitgevoerd op de onderhoudskosten. Daarnaast is het onderzoek verder uitgebreid om de optimale vervangingstijd van het rotorsysteem te bepalen.

Belangrijkste bevindingen

- Het ontworpen optimale onderhoudsplanningsmodel kan de optimale preventieve vervangingstijd bepalen voor windturbinecomponenten, doormiddel van verschillende kostenparameters en betrouwbaarheidsinformatie. Het ontworpen model minimaliseert de totale stilstand-, onderhouds- en logistieke kosten.
- Voor een preventieve onderhoudskosten van 541.213 € en correctieve onderhoudskosten van 1.230.654 € (CM/PM-kostenratio van 2,27), is een effectiviteit van 0,24% gevonden in het gebruiken van een, op leeftijd gebaseerd, onderhoudsbeleid. Het elektrisch omzettersysteem moet preventief worden vervangen na 21,75 operationele jaren met een kosten per tijdseenheid van 56,545,77 €.
- Voor het rotorsysteem blijkt de optimale preventieve vervangingstijd na 9,5 operationele jaren te zijn met een kosten per tijdseenheid van 189.052 €. De gebruikte preventieve onderhoudskosten bedragen € 924.691 en de kosten voor correctief onderhoud bedragen 1.849.382 € (CM/PM-kostenratio van 2). Exploitanten van windparken kunnen € 296.575 besparen op onderhoudskosten van het rotorsysteem gedurende de levensduur van één windturbine. E.g. voor een windpark met 50 windturbines, kunnen de exploitanten ongeveer € 15 miljoen aan onderhoudskosten van het rotorsysteem besparen.
- De optimale preventieve vervangingstijd is sterk afhankelijk van de kostenratio tussen preventieve en correctieve onderhoudskosten. De gevonden optima voor het elektrisch omzettersysteem variëren tussen 25 jaar voor een CM/PM-kostenratio van 2 en 13,25 jaar voor een ratio van 6.
- Het bepalen van de optimale vervangingstijden voor alle belangrijke componenten kan meerdere optima opleveren die dicht bij elkaar liggen en de mogelijkheid bieden om opportunistisch onderhoud uit te voeren. Het combineren van preventieve onderhoudsacties kan de totale onderhoudskosten, die tijdens de levensduur van een windturbine worden gemaakt, nog verder verminderen.

Aanbevelingen

Voor verder onderzoek:

- 1. Het optimale onderhoudsplanningsmodel moet worden uitgebreid met andere seizoensgebonden veranderingen, zoals meer variabiliteit in de kostenbepalingen, wachttijden en capaciteitsfactor.
- 2. Meer onderzoeken moet worden uitgevoerd naar de mogelijkheid om k-mean clustering te gebruiken om meer geschikte optima voor meer dan één component te vinden.

- 3. Een gevoeligheidsanalyse moet worden uitgevoerd op de invloed van de betrouwbaarheidsinformatie van een component op de optimale preventieve vervangingstijd.
- 4. Analyseer de verschillende optima van alle belangrijke componenten om zodoende een mogelijkheid te vinden om opportunistisch onderhoud uit te voeren.
- 5. Onderzoek de mogelijkheid om, op toestand gebaseerd, onderhoud te implementeren met behulp van conditiebewakingssystemen in windturbines.

De implementaties van de bovenstaande aanbevelingen zullen de nauwkeurigheid van het optimale onderhoudsplanningsmodel verbeteren en daarmee de waarde die eruit wordt afgeleid. Bovendien zal, op toestand gebaseerd, onderhoud nader worden onderzocht, omdat het de kosten van preventieve en correctief onderhoudsacties tot een minimum kan beperken.

Table of Contents

Prefacei							
Ma	nagei	ment	summary	ii			
Ma	nageı	ment	samenvatting	V			
List	of fig	gures		X			
List	of ta	bles .		xi			
List	ofat	brevi	ations	xii			
1 Introduction							
1	L.1	Com	pany summary	1			
1	1.2 Pro		olem context	1			
1	L.3	Curr	ent situation	3			
1	1.4 Res		earch goals	4			
1	L.5	Scop	be and demarcations	5			
2	The	oretio	al framework	6			
2	2.1	Com	ponent choice	6			
2	2.2	Maiı	ntenance policy	8			
	2.2.1 Te		Terminology	8			
	2.2.2		Lifetime probability distributions and statistics	9			
	2.2.	3	Exponential and Weibull distribution	. 12			
	2.2.4		Optimal age-based preventive maintenance policy				
2	2.3 Mai		ntenance cost	. 19			
	2.3.1		Maintenance categorization	. 19			
	2.3.	2	Vessel types	. 21			
	2.3.	3	Downtime cost formula	. 22			
	2.3.	4	Capacity factor	. 23			
	2.3.	5	Assumptions and limitations of the maintenance planning model	. 25			
3	Solution		methodology	. 27			
3	3.1	Failu	ıre behavior	. 27			
3	3.2	Relia	ability	. 30			
3	3.3	Prev	entive and corrective costs	. 31			
	3.3.1		Vessel cost				
	3.3.2		Downtimes	. 32			
	3.3.3		Downtime cost	. 32			
	3.3.4		Labor cost	. 33			
	3.3.5		Overview PM and CM costs	. 33			
3	3.4	Opti	mal time interval calculation	. 34			

	3.5	Ben	Benchmark policy					
	3.6	Sensitivity analysis approach						
4	Res	Research results						
	4.1	4.1 Optimal fixed operational time interval						
	4.2	Sens	sitivity analysis	39				
	4.2.	1	Different configurations of maintenance costs	39				
	4.2.	2	Optimal fixed operational time interval (major repair & replacement)	41				
	4.3	Case	e study: rotor system	44				
	4.3.	1	Optimal fixed operational time interval of the rotor system (replacement)	44				
	4.3.	2	Optimal fixed operational time interval of the rotor system (repair & replacement)	48				
5	Con	Conclusion						
6	Rec	Recommendations						
7	Ref	References						
8	Арр	Appendix						
	A: Technical data 5MW WT							
	B: Calculations Excel							

List of figures

Figure 1-1: O&M cost breakdown
Figure 1-2: Problem cluster
Figure 2-1: Main components of a WT6
Figure 2-2: Share on annual failure rate, MTTR and downtime (M.Lutz, 2017)
Figure 2-3: Percentage of downtime per component for two wind power plants in China (Zhao et al,
2017) 8
Figure 2-4: Bath-tub curve (Scheu et al., 2017)9
Figure 2-5: PDF and CDF of a discrete random variable (Reddy, 2011)
Figure 2-6: PDF of a continuous random variable (Reddy, 2011)
Figure 2-7: CDF of a continuous random variable (Reddy, 2011)
Figure 2-8: Reliability calculated from the PDF 12
Figure 2-9: Exponential probability distribution functions13
Figure 2-10: Exponential reliability functions13
Figure 2-11: Weibull PDF
Figure 2-12: Weibull reliability
Figure 2-13: Weibull hazard function 15
Figure 2-14: Schematic overview maintenance policies
Figure 2-15: Scheme of time-based maintenance policies
Figure 2-16: Trapezoidal rule
Figure 2-17: Typical behavior of the objective function g(T)
Figure 2-18: FINO1 location
Figure 2-19: Power curve of the SMW WI
Figure 2-20: Power curve and the increase in power
Figure 2-21: German offshore wind capacity factor (Andrew, 2018)
Figure 2-22: Monthly national capacity factor (load factor) of offshore wind in 2017 (Pineda, 2017) 26
Figure 3-1: Yearly energy produced in GWN
Figure 3-2: Average failure falles per year of the converter system
Figure 3-5. Reliability of the converter system
Figure 2-4. Overview maintenance costs and ratios of total PW of CW cost
Figure 2-5. Average failure fates per year of the converter system (major repair & replacement)
Figure 3-0. Reliability of the converter system (repair & replacement)
Figure 4-1. Optimal fixed operational time to schedule FW for a converter system
Figure 4-2: Optimal objective function $g(t)$ for unreferit comparations
replacement)
Figure 4-4: Optimal objective function $g(t)$ for different CM/PM cost ratios (repair & replacement) 43
Figure 4-5: Reliability of the rotor system (only considering replacement)
Figure 4-6: Ontimal fixed operational time to schedule PM of the rotor system (replacement) 46
Figure 4-7: Optimal objective function g(t) for different CM/PM cost ratios (replacement of rotor
system)
Figure 4-8: Reliability of the rotor system (considering major repair & replacement) 48
Figure 4-9: Optimal fixed operational time to schedule PM of the rotor system (major repair &
replacement)
Figure 4-10:Optimal objective function g(t) for different CM/PM cost ratios (major repair &
replacement of rotor system)

List of tables

Table 2-1: Information of converter system: maintenance actions, repair duration, technicians, vessel
type and material costs (Fraunhofer, 2017) 20
Table 2-2: Failure and maintenance data (Sperstad, McAuliffe, Kolstad & Sjormark, 2016) 20
Table 2-3: Overview maintenance cost for major repair and replacement
Table 2-4: Costs of different vessel types (IEA Wind task 26, 2016) 22
Table 2-5: Weather condition for each season and waiting time (Santos, Teixeira & Soares, 2013) 25
Table 3-1: Share of maintenance actions and failure categories on overall failure behavior
(Fraunhofer, 2017)
Table 3-2: Data of pre-inspection (IEA Wind task 26, 2016)
Table 3-3: CM and PM mobilization cost inputs for each category (optimistic, baseline and
pessimistic)
Table 3-4: CM mobilization time inputs 35
Table 3-5: Configuration inputs for the sensitivity analysis 36
Table 4-1: PM & CM cost for the different configurations
Table 4-2: Optimal operational time t & cost per unit time for each configuration
Table 4-3: Difference in maintenance cost & ratio between PM and CM cost for each configuration 41
Table 4-4: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios 41
Table 4-5: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios (repair
& replacement)
Table 4-6: Share of maintenance actions and failure categories on overall failure behavior rotor
system (Fraunhofer, 2017)
Table 4-7: Information of rotor system: maintenance actions, repair duration, technicians, vessel type
and material costs (Fraunhofer, 2017)45
Table 4-8: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios
(replacement rotor system)
Table 4-9: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios (major
repair & replacement rotor system) 49

List of abbreviations

Abbreviation	Meaning
WT	Wind turbine
0&M	Operations & Maintenance
OECD	Organization for Economic Co-operation and
	Development
LCOE	Levelized cost of electricity
СМ	Corrective maintenance
PM	Preventive maintenance
CBM	Condition-based maintenance
TBM	Time-based maintenance
OEM	Original equipment manufacturer
SCADA	Supervisory control and data acquisition
MTBF	Mean time between failure
MTTR	Mean time to repair
MTTF	Mean time to failure
ECC	Expected cycle cost
ECL	Expected cycle length
MW	Mega watt
kW	Kilo watt
kWh	Kilo watt hour
GW	Giga watt
GWh	Giga watt hour
m/s	Meter per second
СТV	Crew transfer vessel
UL	Jack up vessel
FIT	Feed-in-tariff

1 Introduction

In the following chapter, we will give an introduction for our research. We will first give a company summary for whom the research has been executed. We will then give the problem context followed by a description of the current situation. Thereafter, the proposed research questions are formulated. We will end the chapter with the scope and demarcation set for our research.

1.1 Company summary

Fraunhofer IEE is an independent institute of Fraunhofer and is situated in Kassel, Germany. IEE stands for institute for energy economics and energy system technology. Previously it was one institute with Fraunhofer IWES which focuses on validation of wind turbine technology. Fraunhofer is the biggest research institute for applied research in Europe with around 20.000 researchers and a budget of 1,8 billion euro. Fraunhofer IEE is conducting research for the national and international transformation of energy supply systems. They develop solutions for technical and economic challenges in order to further reduce the costs of using renewable energies, to secure the supply despite volatile generation, to ensure grid stability at the usual high level and to make the business model of the energy transition a success.

Fraunhofer has been gathering and analyzing data concerning failure behaviors of wind turbines for several years. They have been working on a project named "WInD-Pool". This is a joint database of operators in the wind energy industry. The collected data is structured using industry standards after which performance assessment and benchmarking of the industry is made. For this purpose, advice concerning the performance of wind farms can be given to the operators. Furthermore, it enables the determination of reliability characteristics and failure behavior of the wind farms, wind turbines (WTs), and its components.

The executed research will focus on the possibilities in the wind farm industry to reduce the maintenance costs of WT. Operation and maintenance costs have a big share on the total costs incurred to build and operate a power plant, such as a WT. This results in a higher price that must be obtained per kilowatt hour produced, over the lifetime of the WT. With the use of the extensive database and the failure behavior of components, maintenance of wind turbines can be optimally planned by using reliability of its components.

1.2 Problem context

Over the past few years the wind power industry has encountered a large growth. The growth is mainly focused on the development of larger WTs, maximum capacity and the size of offshore wind farms. The maximum capacity generated by a WT has grown as big as 9MW, but next generation turbines will fall between 12 and 15 MW (Merchant, 2018). Bigger turbines are mostly situated farther from shore to access higher wind speeds. However, the reliability and availability of WTs involve high level of uncertainty, which require extensive planning effort. Due to accessibility of the turbine, maintenance services are often difficult and expensive. Most operators have full-service contracts with the original equipment manufacturer. During the duration of these contracts or warranty periods, manufacturers are obliged to take on the service costs of the WT. These contracts tend to be expensive and large wind farm operators increasingly bring more service elements inhouse (Global wind services market, 2017). As the trend towards self-operation by operators increases (IRENA, Renewable Power Generation Costs in 2017, 2018), an influx of researches into the reliability and maintenance management of WTs can be observed.

Whether the operator chooses for a service contract or to conduct the maintenance themselves, the maintenance costs are part of the operation and maintenance costs (O&M costs). The O&M costs consist of variable and fixed costs. The fixed O&M costs typically include insurance, administration, fixed grid access fees and service contracts for scheduled maintenance. Variable O&M costs typically include scheduled and unscheduled maintenance not covered by fixed contracts, as well as replacement parts and materials, and other labor costs (IRENA, Wind Power, renewable energy technologies: cost analysis series, 2012).



Figure 1-1: O&M cost breakdown

The costs of maintenance consist of around 67% of the total O&M costs (IRENA, Renewable Power Generation Costs in 2017, 2018). Maintenance costs are followed by labor cost, which have a share of 14% of the total O&M costs, and materials at 7%. These numbers are depicted in Figure 1-1. For OECD countries, the average O&M costs for onshore WTs tend to be between 0,02 USD/kWh and USD 0,03/kWh (approximately 30 to 47 USD/kW/year). However, O&M costs for offshore WTs are significantly higher than those of onshore WTs. This is mainly the consequence of the higher cost of accessing the site and performing maintenance actions. Crew and vessels used for these maintenance actions are expensive and tend to have high day rate, and a high mobilization cost to access the site. This is due to the marine environment which is a much harsher environment to operate in compared to dry land. In Europe the O&M costs for offshore WTs are estimated to be between 109 USD/kW/year and as high as 140 USD/kW/year (IRENA, Renewable Power Generation Costs in 2017, 2018). Therefore, our research will merely focus on offshore WTs because it is to be expected that more opportunity to decrease the high maintenance costs is to be found here.

Furthermore, the O&M costs in turn tend to have a share of 20% to 30% of the levelized cost of electricity (LCOE) (IRENA, Renewable Power Generation Costs in 2017, 2018). The levelized cost of electricity is the average price that a generating asset must receive in a market to break even over its

lifetime. In this case, the generating asset is the WT and the average lifetime is between 20 and 25 years. In other words, the LCOE measures all the costs incurred during the lifetime of the WT divided by the returns obtained from the energy produced.

It is important to realize that not only the costs of maintenance actions have a big influence on the LCOE but the effects of the maintenance actions too. Implementing a good maintenance policy will prevent failures of components or subsystems which in turn will reduce the downtimes of WTs. Consequently, the WT will have a higher availability or up-time which will provide more revenue and thus achieving a lower LCOE. In conclusion, both the maintenance costs, as well as the loss in revenues play an important role in reducing the LCOE.

Different types of maintenance actions are being executed on WTs. Corrective maintenance (CM) is executed after a failure of a component or subsystem and is intended to put the item into a state in which it can perform its required function. Also, preventive maintenance (PM) is carried out at predetermined intervals or according to prescribed criteria and is intended to reduce the probability of a failure or the degradation of the functioning item (BSI Standards Publication: Maintenance - Maintenance terminology, 2010). Additionally, preventive maintenance can sometimes avoid highly expensive corrective maintenance actions and avoid downtimes of the WT. Some maintenance may be small and frequent (e.g. replacement of small parts, periodic inspections, etc.) while other maintenance may be large and infrequent (e.g. unscheduled repair of significant damage or the replacement of principal components).

1.3 Current situation

As previously discussed, the maintenance costs have a big share in the O&M costs. This is due to the maintenance policy which is currently maintained. According to Fraunhofer a mix of PM and CM is used which can be described as an experience-based maintenance policy. Currently, maintenance actions are executed according to pre-determined time intervals based on previous failure behavior of components or specified by the manufacturer. However, the time intervals between maintenance actions can be optimized which will result in higher availability of the WT and possibly a decrease of the total maintenance costs.

There are different categories which can cause a failure. In the following referred to as failure category. Different failure categories exist e.g.: wind gusts, icing, lighting, overload, wear out, random, early or aging failure. Some failure categories are due to external modes e.g.: weather conditions, which we cannot influence. However, other failure categories can be caused by internal modes e.g.: aging failure, early failure, and wear-out failure. These can be influenced by the maintenance policy that the service provider currently maintains. By conducting PM, the WT and the components can be preserved and restored before a failure actually occurs. Not all failures can be avoided by using a PM policy and some CM actions will still take place.

When a failure occurs, maintenance actions are conducted to repair the WT. Commonly maintenance actions can be categorized in four groups: reset, minor, major and replacement (IEA wind task 26, 2016). These different categories will be explained in section 2.3.1. The total downtime with a reset is usually a few hours, while the total downtime with a replacement can take up to a few weeks. These differences should be taken into account while planning PM and determining the optimal policy. Preferably, the maintenance with long downtime of the WT is to be minimized because they are responsible for the majority of the maintenance costs.

In conclusion, the core problem is that the maintenance policy does not consider the failure behavior of components which may affect the LCOE significantly. An overview of all the problems and their relationships is described in Figure 1-2: Problem clusterFigure 1-2. The core problem is tinted in green.





1.4 Research goals

As previously discussed, the core problem is that the maintenance policy does not consider the failure behavior of components and therefore long downtimes of wind turbines with high maintenance costs are observed which inadvertently results in a higher LCOE. The main research question follows from this core problem.

How can maintenance of wind turbines be planned by using reliability information of components such that total downtime, maintenance, and logistics costs are minimized?

To be able to answer this main research question, sub questions and knowledge questions need to be formulated.

- 1. How can the maintenance of wind turbines be characterized? Can it be modeled by using a well-known maintenance policy? If so, how can the optimal parameters of the selected policy be calculated with the use of failure behaviors?
 - a. What component of the WT should we focus on?
 - b. What are the different types of maintenance actions and what are their corresponding costs?
- 2. What is the influence of different maintenance cost configurations on the optimal maintenance policy?

1.5 Scope and demarcations

To ensure that the research is feasible within the given time restriction, it is necessary to define the scope and the demarcations of the research. We define the scope such that the research can be done within 10 weeks. In this chapter the scope of the research and the set demarcations shall be discussed.

In our research, we will merely focus on offshore WTs. Offshore WTs can be much bigger in size compared to on-shore WTs because they are not limited by space constraints. Those WTs can generate more power thanks to the higher wind speeds measured offshore and the size of the blades. Compared to onshore WTs, offshore WTs usually tend to have a lower reliability and higher maintenance costs due to accessibility of the turbine (IRENA, Wind Power, renewable energy technologies: cost analysis series, 2012). In our research, we will analyze a 5MW offshore WT. The specification of the 5 MW WT can be found in appendix A.

To conduct our research, we have used existing wind speed data of an offshore location situated in the North Sea. In January 2002, the federal government of Germany decided on the construction of three research platforms (FINO1, FINO2, FINO3). These sites were potential sites for the development of wind farms. The wind speed data used are those of the FINO1 location. We will use the FINO1 location to estimate the costs of PM and CM in section 3.3.

Furthermore, while calculating the optimal maintenance policy, we will only focus on one component of the WT. The component choice is discussed in section 2.1. We will use an approximate evaluation model based on discretizing time to be able to calculate the optimal maintenance schedule of this particular part. This model can eventually be expended to other components or sub systems by adjusting the failure behaviors of the component.

Moreover, the found optimal fixed operational time to schedule PM will be compared to the benchmark policy operators currently adhere to. Currently, operators do not use PM to replace components prior to a failure. Therefore, the maintenance policy they adhere can be considered to be solely CM and thus components are only replaced upon failure. The benchmark policy will be determined using the MTTF of components and will be explained in section 3.5.

The LCOE and the total O&M costs will not be determined. There are too many unknowns to amount to a good approximation and therefore these costs indicators will be left out of the research. However, the total maintenance costs and the total downtime costs will be used to give an impression of the improvements the optimal policy yields.

2 Theoretical framework

In the following chapter, we will use existing literature to answer some of our research questions and to underpin several choices that we have made during the execution of the research. We will first discuss what component to focus on. We will then explain some basic statistical knowledge and the two most important distributions, followed by the explanation of the optimal maintenance policy. Moreover, we discuss the maintenance cost structure and make a distinction between preventive and corrective maintenance. We will end this chapter with the assumptions and limitations of the optimal maintenance planning model.

2.1 Component choice

There are different types of WTs with varying designs, advantages and disadvantages. The most common type is the horizontal axis WT as depicted in Figure 2-1. A typical WT will contain up to 8000 components. The main components are the generator system, yaw system, blade/pitch system, gearbox system, converter system, rotor system and several other systems. In Figure 2-1, the main components are indicated with numbers. We will briefly explain some components and their functions. The converter (number 13) and transformer (number 14) indicated below can also be installed in the tower instead of in the nacelle.





The main components of a WT are shown in Figure 2-1and their functions (EWEA, 2017) are:

- 1. Blade
- 2. Blade support
- **3.** Pitch angle actuator adjusts the angle of the blades to make the best use of the prevailing wind
- 4. Rotor hub is made from cast iron and it holds the blades in position as they turn
- 5. Spinner
- 6. Main support
- 7. Main shaft transfers the rotational force of the rotor to the generator
- 8. Aircraft warning lights

- **9. Gearbox**, the gears increase the low rotational speed of the rotor shaft in several stages to the high speed needed to drive the generator
- **10.** Mechanical brakes, disc brakes bring the turbine to a halt when required
- **11. Hydraulic cooling devices**
- **12. Generator** converts mechanical energy into electrical energy. Both synchronous and asynchronous generator are used
- **13.** Power converter and electrical control, protection and disconnection devices. Power converter converts direct current from the generator into alternating current to be exported to the grid network
- 14. Transformer converts the electricity from the turbine to higher voltage required by the grid
- **15. Anemometers** measures the wind speeds
- 16. Frame of the nacelle
- 17. Supporting tower
- 18. Yaw driving device, mechanisms that rotates the nacelle to face the changing wind direction

A failure of the WT can be caused by one or several components. Some components cause more failures of the WT then others. According to Fraunhofer (M.Lutz, 2017), and shown in Figure 2-2, the rotor system, converter and control system have the biggest share on the failures. These components also have a high share on the mean time to repair. Perhaps the most important and interesting part is the fact that the converter and rotor system have the biggest share on the annual downtime.



Figure 2-2: Share on annual failure rate, MTTR and downtime (M.Lutz, 2017)

According to Zhao, Li, Dong, Kang, Lv and Shang (2017), as shown in Figure 2-3, the generator and converter system have the biggest share on the annual downtime of WT. There is however a difference in the data used by Fraunhofer and Zhao which explains the discrepancy between the results. Zhao used data collected from the widely available supervisory control and data acquisition (SCADA) system of two different wind farms in China. On the contrary, Fraunhofer has a broader categorization of components compared to Zhao et al. (2017) and other researches. It can be seen that Fraunhofer makes a distinction in several generator components and systems whereas this is not the case in Zhao research. Nevertheless, we can see that the converter system has one of the biggest shares on the annual downtime of the WT.



Figure 2-3: Percentage of downtime per component for two wind power plants in China (Zhao et al, 2017)

As discussed in the introduction, not only do O&M costs have an influence on the LCOE but also the observed downtime of WTs. We will make a distinction between the total PM and CM costs in section 3.3. However, for now, it is important to note that the downtime costs will be accounted for in the total PM and CM costs. Obviously, components with a big impact on the failures and long downtime of the WT are the most interesting for our research. Conducting PM activities can avoid unnecessary failures and thus decrease the downtime of the WT. Combining the above and after consultation with Fraunhofer, we have chosen to research the optimal maintenance policy for the converter system.

2.2 Maintenance policy

Maintenance is defined "as the combination of all technical, administrative and managerial actions during the life cycle of an item intended to retain it in, or restore it to, a state in which it can perform the required function" (BSI standards publication, 2010). Maintenance is crucial for heavy and capital-intensive industries to keep their machinery and equipment in good operating condition. Many different types of maintenance policies are available with each their own benefits and drawbacks. Some maintenance policies are better suited for some components and others for the overall system. After consultation with my supervisor and Fraunhofer, we have chosen to use the optimal aged-based replacement (type 1 policy) which will be explained in section 2.2.4.

2.2.1 Terminology

First, three important definitions frequently used in maintenance concepts and methodology are to be explained.

Availability is defined as the probability that an item will perform its required function under given conditions at a stated instant of time (Scheu, Kolios, Fischer & Brennan, 2017). Logically, positive financial turnover can only be encountered in periods of availability. It must be noted that availability rests on other important factors such as reliability, maintainability and accessibility. Availability can be expressed with an equation if the mean time between failures (MTBF) and mean time to repair (MTTR) is known. We will not use this specific equation because we do not possess actual data of the up and downtimes of WT's. We would like to refer the reader to Ostachowicz, McGugan, HInrichs & Luczak (2016) and Tinga (2013) for the equation of availability.

Maintainability refers to the relative ease and efficiency of performing tasks associated with machine maintenance, including both routine service and unplanned repairs (Walford, 2006). Maintainability can also be expressed with the probability of achieving an objective (repair, replacement, etc.) within a given time.

In the English vocabulary, accessibility is defined as the fact of being reached or obtained easily. As previously explained, the accessibility of offshore WTs are more difficult than onshore WTs because of the harsher marine environment, and the vessels needed to access the site.

In our study, we will not make use of the three definitions and equations above. However, these definitions are often used in maintenance practices and concepts and it is thus important to know their definitions.

2.2.2 Lifetime probability distributions and statistics

To be able to understand how our maintenance policy works, we first need to explain some distributions and statistical concepts. This will help us understand the different failure modes addressed in section 3.1.

A failure is the termination of the ability of an item to perform a required function (BSI Standards Publication: Maintenance - Maintenance terminology, 2010). With adequate data it is possible to show that, on the average, a component fails after a certain given time. This is called the failure rate and it is often used in reliability engineering. It can be expressed with the Greek letter, λ (lambda). Probability distributions are used to model the failure behavior of e.g. components. Failure rates commonly depend on time and vary during the life cycle of the component or system. Thus, the failure rates are usually not constant and are mostly shaped as the bathtub curve (Scheu et al., 2017) as depicted in Figure 2-4. This implies that in the early stage components encounter a decrease in the failure rate. However, an increase in the failure rate is observed in a later stage, which is named aging failure.



Figure 2-4: Bath-tub curve (Scheu et al., 2017)

A random variable is a numerical description of the outcome of an experiment whose value depends on chance, i.e., whose outcome is not entirely predictable (Reddy, 2011). There are mainly two types of random variables. A discrete random variable is a variable that can only take a finite or countable number of values (e.g. a coin with heads and tails). On the contrary, a continuous random variable is a variable that may take on any value in an interval (e.g. temperature). The probability of a variable taking on different values is represented by the distribution functions of the random variable. We will discuss two examples and make a distinction between a probability density function and a cumulative distribution function (resp. PDF and CDF). Depending on whether a random variable is discrete or continuous, one will get a discrete or continuous probability distribution.

In Figure 2-5, we see the probability functions of a discrete random variable involving the outcomes of a rolling dice. The first histogram (Figure 2-5a) represents the probability of the dice taking on a certain value which is called the probability density function (PDF) and denoted with f(x). The Y-axis represents the probability and the X-axis represents the value of the dice. Thus, the probability of a dice taking on the value "3" is 1/6.



Figure 2-5: PDF and CDF of a discrete random variable (Reddy, 2011)

In Figure 2-6, we see the probability functions of a continuous random variable. However, for a continuous random variable it is implausible to determine the probability that the outcome will be a certain value e.g. 57,5°F. The probability that the value will be within a certain range, e.g. between 55°-60°F, can be determined by calculating the area under the PDF curve, as shown in Figure 2-6b. It is for the continuous random variables that the CDF becomes useful, denoted with F(x). The CDF is simply the area under the PDF curve starting from the lowest value up to the desired value. This is illustrated in Figure 2-7. With such a CDF plot, it is easily to determine the probability that the temperature will be less than 60°F.



Figure 2-6: PDF of a continuous random variable (Reddy, 2011)



Figure 2-7: CDF of a continuous random variable (Reddy, 2011)

Provided a derivative exists, the inverse relationship between f(x) and F(x) is as followed.

$$f(x) = \frac{dF(x)}{dx}$$
(1)

Asset reliability is defined as the "The ability of an item to perform a required function under stated conditions for a given time interval" (Igba, Alemzadeh, Durugbo & Henningsen, 2015). Reliability is expressed as a probability value (a value between 0 and 1) and is denoted with R(x) or $\bar{F}(x)$. The reliability can be calculated with the use of the PDF. The sum of all the possible outcomes of the PDF must be equal to 1 and cannot be negative.

With the use of the PDF, the probability that an item will fail within a certain time interval can easily be calculated with the knowledge above. Now, the continuous random variable is the variable time. So, calculating the probability that an item will fail up to a given point in time is the area under the PDF curve. Knowing that the sum of all possible outcomes must be equal to 1, the probability that an item will survive up to that given point in time, is simply 1 subtracted by the probability of a failure. The equation of reliability is as followed and illustrated in Figure 2-8.

$$R(t) = 1 - F(t)$$
 (2)

Note that the "x" in the equation has been replaced with "t" due to the random variable now being time, which is denoted by t. The reliability can also be calculated from the failure rates with the following equation:

$$R(t) = e^{-\int_0^t \lambda(t)dt}$$
(3)



Figure 2-8: Reliability calculated from the PDF

Where R(t) represents the reliability, F(t) can also be described as the unreliability of a component. However, what is more commonly used is the failure rate function or the so-called hazard function denoted with λ (t). Given that the component has survived until time t, the hazard function represents the probability of a failure per unit time t. It can be expressed with the following equation:

$$\lambda(t) = \frac{f(t)}{R(t)} \tag{4}$$

Finally, the MTTF describes the mean time expected until the first failure for a non-repairable system. MTTF can be determined from the reliability using the following equation:

$$MTTF = \int_0^\infty R(t)dt$$
 (5)

2.2.3 Exponential and Weibull distribution

Consequently, we need to discuss two types of distribution often used in reliability engineering. As discussed above, we can calculate the failure rate and the probabilities of a failure after a given time t. Depending on the failure rate, one distribution is more suitable then another.

2.2.3.1 Exponential distribution

Assuming we are dealing with a constant failure rate, we can use the exponential distributions to determine the probability of a failure. Components with a constant failure rate are simply components that do not degrade as time continues. In fact, the exponential distribution is a special case of the Weibull distribution with a shape parameter of 1 (β =1). We will discuss the Weibull distribution and its characteristics later on. The exponential distribution is useful to calculate the remaining useful life of a component. The PDF of the exponential distribution is as followed.

$$f(t) = \lambda e^{-\lambda t} \quad \text{for } 0 \le t \le \infty$$
 (6)

In Figure 2-9, we can see the exponential probability distribution function for various values of λ .



Figure 2-9: Exponential probability distribution functions

The CDF of the exponential distribution is as followed.

$$F(t) = 1 - e^{-\lambda t}$$
(7)

The reliability can now be calculated and is as followed. The exponential reliability functions for various values of λ are depicted in Figure 2-10.



$$R(t) = 1 - F(t) = e^{-\lambda t}$$
 (8)

Figure 2-10: Exponential reliability functions

Combining equations (6) and (8), the Hazard function will yield the following:

$$\lambda(t) = \frac{f(t)}{R(t)} \qquad (9)$$

2.2.3.2 Weibull distribution

The Weibull distribution has wide range of applications in reliability calculation due to its flexibility in modeling different distribution shapes. In addition to being the most useful distribution function in reliability analysis, it is also useful in classifying failure types, trouble shooting, scheduling preventive

maintenance and inspection activities (Vermat, Ajit, Karanki, 2016). The Weibull distribution has two parameters, α and β (scale and shape parameters). Thanks to the scale and shape parameters, several other distributions can be modelled. The PDF of the Weibull distribution is as followed.

$$f(t) = \frac{\beta}{\alpha} \left(\frac{t}{\alpha}\right)^{\beta - 1} e^{-\left(\frac{t}{\alpha}\right)^{\beta}}$$
(10)

In Figure 2-11 we can see that the Weibull PDF takes on different shapes according to the shape parameter β .



Figure 2-11: Weibull PDF

The Weibull CDF and reliability function are as followed. The Weibull reliability functions for various shape parameter β are depicted in Figure 2-12



Figure 2-12: Weibull reliability

The Weibull hazard function is as followed and is shown in Figure 2-13 for various shape β parameters.



$$\lambda(t) = \frac{\beta t^{\beta-1}}{\alpha^{\beta}} \quad (13)$$

Figure 2-13: Weibull hazard function

Taking into consideration the bath-tub curve discussed at the beginning of this chapter, the three bath-tub regions can be represented with the Weibull distribution and varying β values.

- β < 1 results in decreasing failure rate (burn-in period)
- β = 1 results in constant failure rate (useful life period)
- $\beta > 1$ results in increasing failure rate (Wear-out period)

2.2.4 Optimal age-based preventive maintenance policy

Two distinctive maintenance strategies can broadly be classified as corrective and preventive maintenance (Galambos, Galambasova, Rataj & Kavka , 2017). CM is executed only when a system or component has failed and needs replacement or repair. In some cases, systems can continue working when a failure of a component has occurred due to installed back-up components or due to the importance of the component itself. WTs are very complex systems where a failure of a component can have severe consequences such as damage to other components or subsystems. But for us the most important part is that the failure of a component can also have economic consequences such as expensive maintenance costs due to accessibility of the WT, labor hours, costs of spare parts or loss in revenue caused by the downtime.

PM involves maintenance activities prior to the failure of the system or component. However, the optimal moment of replacement is often hard to determine. One of the main objectives is to reduce the failure frequency or the downtime of the system or component. PM contributes to minimizing failure costs and machine downtime (Galambos, Galambasova, Rataj & Kavka , 2017). This strategy

can be based on the Original Equipment Manufacturer (OEM) recommendations. This is suitable for some PM activities such as lubrication, oil changes, filter cleaning and so on. In our case, PM will involve the maintenance action of replacing a component prior to a failure, even though the component has not reached the end of its life cycle. However, we will first explain the different types of PM policies.



Figure 2-14: Schematic overview maintenance policies

Generally, there are three types of PM policies: time-based maintenance (TBM), condition-based maintenance (CBM), and opportunistic maintenance. These are depicted in Figure 2-14. Opportunistic maintenance is a form of PM and it is based on the opportunity to perform PM actions by taking advantage of planned or unplanned maintenance. Thus, while CM or PM is performed, and suitable maintenance resources are already at location, opportunistic maintenance can be performed on other components or systems. As opportunistic maintenance is not a maintenance policy which can be used to determine the scheduling of PM actions, we will further look into CBM and TBM policies.

CBM is typically used when there is a higher degree of uncertainty in the deterioration models, thus in the lifetime of the component or system (Wind farm data collection and reliability assessment for O&M optimization, 2017). Maintenance decisions are based on information of the actual condition or health of the component or system. This information can be obtained from health or condition monitoring systems. However, applying CBM, is only possible if there are conditions that are related to the moment of failure and if it is technically possible to monitor these conditions (Jonge, 2017). Due to time restriction and availability of data, we have decided not to focus on CBM but on TBM policies.

TBM is carried out in accordance to a pre-determined time schedule. We can make a distinction between two types of TBM policies, namely age-based and block-based maintenance. Under blockbased maintenance policy, PM is scheduled after a fixed interval with length T, depicted in Figure 2-15b. If a failure occurs during this fixed interval, CM is performed. However, these failures do not influence the PM schedule thus the fixed interval will remain the same. The disadvantage of this policy is that PM is sometimes scheduled shortly after an expensive CM has been performed on the system or component. The main advantages are the easier planning of PM and the clustered maintenance actions when the same block-based maintenance is applied to multiple units (Jonge, 2017). As there is a substantial cost difference between PM and CM actions, block-based maintenance might be very costly over the lifetime of a WT. Therefore, we have decided to focus on an age-based policy.



Figure 2-15: Scheme of time-based maintenance policies

As explained in section 2.1, we will only focus on one component of the WT. A commonly used PM policy to determine the optimal PM interval for a single component is the age-based preventive maintenance policy (type-1 policy). Additionally, this policy yields the possibility to be extended and to take into account several components so that an optimal PM interval can be determined for more than one component. Thus, we will use an age-based preventive maintenance policy to analyze our problem and determine the optimal maintenance policy. An age-based preventive maintenance policy is defined as a maintenance policy which performs PM after fixed operational time T (e.g. hours, days, months, years, etc.) without a failure. T can be finite or infinite. In case that T is infinite, no PM will be scheduled. If the system fails prior to operational time T, CM will take place, and the PM is rescheduled after T operational time. In this policy, the system is assumed to be "as good as new" after PM or CM (Sultana & Karim, 2015). The policy is illustrated in Figure 2-15a.

The age-based policy is useful when the failure rate function of the system or component is increasing. Additionally, the costs of PM must be (much) smaller than the costs of CM. If the costs of PM and CM would be equal or close to each other, the policy would yield to not do any PM and wait until a failure occurs.

The objective function g(T) of the age-based policy is the expected cost per unit time. It can be obtained from the ratio between the expected cycle cost (ECC) and the expected cycle length (ECL).

$$g(T) = \frac{Total expected cost per cylce (ECC)}{Expected cycle length (ECL)}$$
(14)

The total ECC can be determined with the costs of CM and PM and the probabilities of a failure. The ECC is calculated as shown below.

$$ECC = Co\overline{F}(T) + C1F(T)$$
(15)

T represents the fixed operational time as previously discussed. Furthermore, the theory stated above uses $\overline{F}(T)$ to represent the reliability. However, the reliability can also be denoted by R(T). C₀ are the costs of PM and are incurred if the system does not fail before operational time T. The probability of this happening is equal to the reliability at time T. Thus, the probability of C₀ is 1-F(T)= $\overline{F}(T)$. If the system does fail before time T, the costs of CM will be C1. The probability that the system fails prior to time T is F(T).

The ECL is calculated as followed.

$$ECL = \int_0^T tf(t) + T\bar{F}(T)$$
 (16)

The time until a failure is denoted with the random continuous variable X and distribution function F(x). PM is executed when $X \ge T$, thus when the time until a failure exceeds the fixed operational time T. In other words, no failure will occur before the fixed operational time T. The cycle length is then T with probability $\overline{F}(T)$. CM is executed when X <T in which case a failure will occur before the fixed operational time T. The cycle length will then be X with probability density f(t). ECL can also be calculated with partial integration which yields the following formula:

$$ECL = \int_0^T \bar{F}(t) dt \qquad (17)$$

The ECL is actually the area under the reliability curve from 0 to time T. To ensure that we come to a good approximation of the ECL we use the trapezoidal rule to compute the area. This implies that we compute the total area by calculating small areas under the curve and finally sum these up. An example of the trapezoidal rule is depicted in Figure 2-16. The equation is as followed.



 $\int_{0}^{T} h(t)dt \approx \sum_{i=1}^{n} \frac{1}{2} (ti - t(i-1)) \{h(t(i-1)) + h(ti)\}$ (18)

Figure 2-16: Trapezoidal rule

The typical behavior of the objective function is shown in Figure 2-17. The optimal fixed operational time to plan PM can be found at point T* where the total costs per unit time are minimal. At point $g(\infty)$, no PM is performed and the system continues working until a failure occurs. The effectiveness of using an age-based replacement policy compared to a CM policy can be determined by the following equation:



Figure 2-17: Typical behavior of the objective function g(T)

2.3 Maintenance cost

In the following section, we will determine the difference in PM and CM cost. Due to confidentiality, these costs were not provided by manufacturers or operators of WTs and assumptions have been underpinned using existing literature.

We have used wind speed data of FINO1 location situated in the North Sea to be able to calculate the wear-out and fatigue failure of a WT. We will further explain this in section 3.1. In the following sections, we will use the FINO1 location as the location of our WT where we want to perform PM and CM. The FINO1 site is located approximately 60 km of the shore of Germany, illustrated in Figure 2-18. We will use this information to determine the travel costs for different maintenance vessels in section 3.3.



Figure 2-18: FINO1 location

2.3.1 Maintenance categorization

Maintenance of offshore WTs are more difficult and expensive due to accessibility of the site, bad weather conditions, equipment and furthermore. The costs of maintenance actions are dependent on the different vessels needed, labor costs, downtime costs, etc. In the previous section, we have made a distinction between PM and CM. CM is executed once a failure has occurred, on the contrary PM is executed before the occurrence of a failure. However, when a failure occurs different maintenance actions can be needed. The maintenance actions can commonly be categorized in four groups: reset, minor, major and replacement (IEA Wind Task 26, 2016).

- Manual resets are brief inspection or resets of the WT which sometimes involve routine replacements or some consumables. Commonly two technicians are sent to the turbine to execute these maintenance actions.
 In some cases, the maintenance action will involve a remote reset. A remote reset is where the WT is reset remotely from the control room. The consequence is only some downtime and no technicians are needed.
- Minor repairs are relatively small repairs of the WT needing small spare parts. These maintenance actions involve small crew of technicians 2 to 3 technicians.
- Major repairs are large repairs and small component replacements needing relatively small spare parts. These replacement operations take longer to execute and involve a larger crew of around 6 technicians.

• Major replacements are the replacements of large components involving a vessel with lifting capacity e.g. jack-up (JU) vessel. For these types of maintenance actions, a large crew and technicians are needed.

Reset and minor repairs usually only take a few hours. On the contrary major and replacement repairs usually take longer than a day. For the converter system, the time needed for the different maintenance actions, technicians needed, duration and vessel type are summed up in Table 2-1.

Table 2-1: Information of converter system: maintenance actions, repair duration, technicians, vessel type and material costs (Fraunhofer, 2017)

Komponente	Ausfall- konsequenz	Reparaturdauer s (h)	Techniker ত	Schiffs- typ	Material- kosten ଢ (€)
		e (,	-	576	
	Reset	3	2	CTV	
Convertor System (-MSE)	Minor Repair	7	2	CTV	240
Converter system (=wise)	Major Repair	14	6	Crane	5300
	Replacement	57	6	JU	13000

On average, reset and minor maintenance are responsible for 75% of the maintenance activities but only have a share of 5% on the total downtime observed due to maintenance activities. On their turn, major and replacement repairs are responsible for 25% of the maintenance activities but have a share of 95% on the total downtime (Milborrow, 2014). Here, the same classification of maintenance actions is used as previously described. Because downtime has a big contribution to the maintenance costs, we will merely focus on the major and replacement repairs.

According to (Sperstad, McAuliffe, Kolstad & Sjormark, 2016) and (IEA Wind task 26, 2016) the vessels needed, to perform different maintenance actions are summed up in Table 2-2. This information differs from the information Fraunhofer possesses. Additionally, according to (Stalhane, Vefsnmo, Halvorsen-Weare, Hvattum & Nonas, 2016), only gearbox major repairs will require lifting capacity. Sperstad et al. state that major repair actions can be executed with the use of CTV's. Wind farm operators usually have long term agreements concerning CTV's and it can be assumed that these are always available and fairly cheap.

Maintenance task	Failure rate [per turbine per year]	Active maintenance time [h]	Spare part lead time [days]	Number of technicians required	Vessel required
Manual resets	5	3	0	1	CTV
Minor repair	3	7.5	0	3	CTV
Major repair	0.3	22	10	5	CTV
Major replacement	0.11	34	60	n/a	Jack-up vessel
Annual service	n/a	70	0	3	CTV

Table 2-2: Failure and maintenance data (Sperstad, McAuliffe, Kolstad & Sjormark, 2016)

The theory we use to determine the optimal fixed operational time to schedule PM states that the failure rate must be increasing over time, the costs of PM must be much smaller than CM, and that the system can be assumed to be "as good as new" after maintenance actions. The failure rate of the major repairs is indeed increasing over time. However, according to our calculations, the costs of CM and PM for major repairs will only differ slightly. The difference between both will consist of the cost of a pre-inspection to assess which maintenance action should take place, and the additional downtime cost of the pre-inspection. As discussed in section 2.3.2, crew transfer vessels (CTV's) have a day rate cost of 3500€ and the pre-inspection duration is approximately 8 hours. Depending on the capacity factor, which will be explained in section 2.3.4, the downtime cost due to pre-inspection will be between 1000€ and 1600€ (resp. capacity factor of 0,25 and 0,4). In case that a crane is indeed

needed to undertake the major repair, a crane ship can be hired and will cost approximately 6666 €/hour (Puglia, 2013). However, the 5MW we analyze in our research has a service crane installed in its nacelle and can lift small spare parts up to 2000kg (RePower, 2004) which makes hiring a crane ship redundant.

We can use the failure rate function to determine the probability of a failure and the probability that major repairs will be needed. We could decide to schedule major repairs in advance but, in case that the failure has not yet taken place, it would be unnecessary to undertake any maintenance actions. Furthermore, the system cannot be assumed to be "as good as new" after major repairs. We could decide to aggregate the failure rates of the major repairs and major replacements. This would imply that we would always replace the component even though it could possibly be repaired with major repairs. In section 2.3.2, we show that the vessel needed for replacement is very expensive compared to a CTV or a crane ship used for major repairs.

To conclude, we have decided to focus merely on major replacements because the difference between the PM and CM costs is substantial. The difference in PM and CM for major repairs will only differ slightly and the system cannot be assumed to be "as good as new". The difference between CM and PM will consist of a pre-inspection with a CTV at a cost of 3500€, and the downtime cost during the pre-inspection which is between 1000€ and 1600€. Furthermore, the labor cost is expected to be higher during CM due to higher labor rates for unscheduled labor. All the other costs of PM and CM for major repairs will be the same. The difference between PM and CM costs for major repair and replacement are summed up in Table 2-3. The calculations of Table 2-3 can be found in section 3.3. In addition, if we decided to replace a component instead of executing major repair, the maintenance cost of replacing the component would be higher compared to performing corrective major repairs.

	PM (€)	CM (€)
Major repair	67.076	73.175
Replacement	541.213	1.230.654

Table 2-3: Overview maintenance cost for major repair and replacement

2.3.2 Vessel types

Depending on the maintenance action, a certain type of vessel is needed. For a reset and minor repair, CTV's are used. A standard CTV can reach speeds up to 40 km/h (22 knots). For major repairs and replacements, a CTV, crane or JU are commonly used depending on the component. JU vessels are big ships with heavy lifting equipment and enough space to transport large spare parts. However, CTVs are needed for pre-inspections, and an 8-hour pre-inspection is assumed for major repairs and major replacements (Sperstad, McAuliffe, Kolstad & Sjormark, 2016).

The costs of the different vessel types are summed up in Table 2-4. We see that the JU has a high mobilization time and costs. Mobilization time here is understood as the total length of time from when the need for the vessel is identified to the ordered vessel becoming available in the wind farm, including waiting for a vessel to become available on the market, travel time to the maintenance base of the wind farm, waiting for it to be re-equipped, etc. (IEA Wind task 26, 2016). In reality, the mobilization time has a high degree of variability and depends on location of the wind farm, vessel type, time of the year, charter strategy (buy or rent), etc. Mobilization time of 60 days is considered optimistic but not unreasonable for short-term-on-demand charters in the European market (IEA

Wind task 26, 2016). The mobilization cost are the costs incurred for the JU vessel to travel to the wind farm, being re-equipped, planning and preparing the marine operation, and the transportation method. The mobilization cost also has a high degree of variability. The travel distance may differ dependent on the previous location of the JU. Also, the JU vessel can be mobilized to the wind farm using different transportation methods e.g. being towed, heavy lift or self-propelled. A mobilization cost of 500.000€ will be incurred for CM and is based on (Dinwoodie, Enderud, Hofmann, Martin & Sperstad, 2014) and (Maples, Saur & Hand, 2013).

Furthermore, it is assumed that the JU vessel will operate 24 hours a day while at site. The labor cost of the technicians needed for major replacements and the fuel cost are included in the day rate of the JU. However, for CTV's the cost of technicians are not included in the day rate.

Maintenance type	Vessel speed [knots]	Transfer time [minutes]	Technician space	Day rate [€]	Mob. time" [days]	Mobilisation cost [€]	Signifi- cant wave height limit [m]	Wind speed limit [m/s]
Standard crew transfer vessel (CTV)	22	22	12	3,500	n/a	n/a	1.8	16
Jack-up vessel	7	n/a	n/a	140,000	60	500,000	2	11
Diving support vessel (DSV)	16	n/a	n/a	75,000	15	225,000	2	n/a
Cable laying vessel (CLV)	14	n/a	n/a	100,000	30	550,000	1	n/a

Table 2-4: Costs of different vessel types (IEA Wind task 26, 2016)

 Mobilisation time shown includes lead time waiting for a vessel to be available on the market for the sort of short-term on-demand charters assumed here.

2.3.3 Downtime cost formula

When a major component fails, the WT will stop working and will not produce any electricity. The operator occurs a loss in revenue during the standstill, which is referred to as downtime cost (Kirkeby & Mikkelsen, 2016). The cost for downtime is calculated as followed (Puglia, 2013).

Cost of downtime $(\mathbf{E}) =$ Number of WT * rated power * capacity factor * cost of electricity * downtime (20)

In our case, the number of WT is equal to 1. The rated power is the maximum power that the WT can generate per hour. This will be equal to 5MW (5000 kW). The cost of electricity is the price obtained per kWh. The price of electricity fluctuates depending on the wind speeds and produced energy. However, in consultation with Fraunhofer, we have decided to use a feed-in tariff (FIT) of $0,10 \in kWh$. A FIT protects producers of renewable energy for some of the inherent risks of renewable energy production. These are typically achieved by offering long-term agreements for the pricing of the energy. The goal of a FIT is to provide price certainty and to help finance investments in renewable energy. The capacity factor will be discussed in section 2.3.4. The downtimes will be calculated in section 3.3.2. Finally, the downtime cost will be calculated in section 3.3.3.

2.3.4 Capacity factor

The capacity factor (Cf) of a WT is the ratio of the average power produced and the rated power. The equation of the capacity factor is as followed.

$$Cf(\%) = \frac{Average \ power}{Rated \ power}$$
(21)

As previously stated, the maximum power that the WT can generate is 5MW. To determine the actual power output at a given point in time we need to know the measured wind speed at that point in time. Measured wind speed can be converted to average power produced using the power curve of a WT. The power curve shows the power generated by the WT at different wind speeds, see Figure 2-19.



Figure 2-19: Power curve of the 5MW WT

On the x-axis we can see the different wind speeds. On the y-axis we see the generated power for the corresponding wind speed. Notice that the curve starts at a wind speed of 3,5 m/s and stops at 30 m/s. These correspond with the cut-in and cut-out speed of the WT. The turbine will not generate power at wind speeds under 3,5 m/s and above 30 m/s.

As stated at the beginning of section 2.3, we have used wind speed measurement of the FINO1 location to analyze our problem. The wind speed data of the FINO1 location is measured from 1-1-2014 until 1-1-2018 and is measured at a height of 100 meters. This is equivalent with the hub height of our WT. The wind speed is measured every 10 minutes and the date, hour, and time interval are precisely logged. The average is determined from a sequence of 6 consecutive measurements that do not overlap each other e.g. 1 to 6, 7 to 12, etc. Thus, each hour consists of 6 wind speed measurements from which the average wind speed is calculated. From the hourly average wind speeds, we can determine the hourly average power of the WT. From the data follows that for a 25 year span the average wind speed is equal to 9,2 m/s which is equivalent to an average power of somewhat near 2500 kW. This result is not exactly correct because the WT may have had some standstills due to too low or to high wind speeds (cut-in and cut-out wind speeds). Therefore, we have calculated the hourly average power for 25 years using the wind speed data of the FINO1 location. This calculation yield that the average power of the 5MW WT is equal to 2630 kW.

Even though we now have taken into account the standstill due to the cut-in and cut-out wind speeds the average generated power is higher than before. This is because the power curve of the WT isn't linear. As wind speed increases, the generated power also increases however it proportionally increases more. Thus, accessing higher wind speeds will result in substantial more
power generated. The power curve and the increase in generated power compared to the wind speed are depicted in Figure 2-20.



Figure 2-20: Power curve and the increase in power

According to the data and information above, the capacity factor of the WT would be equal to 52,6%. According to existing literature, baseline measurements and actual data of German wind farms, the average lifetime capacity factor is 41,1%, see Figure 2-21. We have decided to use the average capacity factor of German wind farms in our downtime cost calculation for the following reason. The data of the FINO1 location consists of wind speed measurements at 100 meters. From this the power generated by the WT can be determined. However, in addition to the standstill caused by the cut-in and cut-out wind speeds, the WT experiences standstills during maintenance actions. These maintenance actions are not included in the FINO1 data and therefore we will use the average capacity factor of 41,1%.

All numbers are to May 2018. Analysis by EnergyNumbers.info.	Latest rolling 12-month capacity factor	Lifetime capacity factor	Age ^(y) \$	Installed capacity (MW) \$	Total elec. gen. (GWh) ♦	Power per unit area spanned (W/m²) 🖨
Amrumbank West	41.7%	44.1%	2.6	302	2 872	4.0
Bard Offshore 1	37.0%	37.8%	1.1	400	1460	2.6
<u>Borkum Riffgrund I</u>	36.8%	40.5%	2.6	312	2 881	3.5
<u>DanTysk</u>	48.0%	50.0%	3.1	288	3 178	2.2
Gode Wind I		40.3%	1.0	330	1 1 2 7	
Gode Wind II		39.9%	1.0	252	852	
Nordsee One		32.0%	0.4	331	354	
Nordsee Ost 1	33.9%	36.1%	3.0	144	1 368	2.9
Nordsee Ost 2	35.8%	36.8%	3.0	144	1 402	2.9
Sandbank	46.2%	47.1%	1.3	288	1 552	2.9
Windpark Baltic 1 & 2	46.7%	46.8%	2.6	336	3 516	4.3
Total	41.4%	41.1%		3128	20 563	3.0

Figure 2-21: German offshore wind capacity factor (Andrew, 2018)

2.3.5 Assumptions and limitations of the maintenance planning model

It is important to note, that in the theoretical framework discussed above, we have made some important assumptions.

• Wave and wind speed limits, as waiting times due to bad weather are not incorporated in the maintenance planning model

The reader may have noticed in Table 2-4 that different types of vessel have different wave height limits and wind speed limits. In case of bad weather and rough sea, most vessels cannot leave port and need to wait until the weather improves to execute the maintenance actions at sea. As an example, if wave heights exceeding 2 meters or wind speeds exceeding 11 m/s are experienced, the JU cannot travel to the wind farm and must stay at port.

Weather commonly fluctuates during the year. In the winter, the weather is usually worse with higher wind speeds and wave heights. On the contrary, in the summer, the weather is usually better. Also, the weather conditions during the autumn and the spring differs. The probability of good weather and sea conditions is lower during winter months compared to e.g. the summer months, see Table 2-5. Good weather conditions refer to favorable wave and wind speed conditions which lay within the limits of the JU vessel. In addition, the waiting times of vessels in port due to weather conditions is shown in Table 2-5. This implies, that executing maintenance actions during the summer is more favorable because the probability of waiting in the port, due to bad weather, is smaller. If a vessel must wait in port, no maintenance actions can be executed on the WT. In case of CM, the WT is at a standstill and the operator will suffer additional downtime cost. For both PM and CM, if a vessel must wait in port, the vessel cost increases due to the day rate cost.

Year Seasons	Probability of having good sea conditions	Waiting time (days)
Winter	0,3	10
Autumn	0,5	7
Spring	0,6	5
Summer	0,8	2

Table 2-5: Weather condition for each season and waiting time (Santos, Teixeira & Soares, 2013)

• Average day rate of the JU vessel is used instead of seasonal day rates

As previously discussed, summer months are more favorable for executing maintenance action. Therefore, the demand during these months is higher and the day rate increases. During winter months, the demand is lower and thus the day rate decreases. During winter months the daily rate can be as low as 130.000 €/day and during the summer months as high as 192.000 €/day (Dalgic, Lazakis, Turan & Judah, 2015). JU vessels are commonly hired for a month from which the day rate is calculated. During the hired month, the JU is used to execute several maintenance actions on different WTs. In our research we focus on only one WT and therefore we have decided to only use the day rate in the vessel cost calculations.

• Fluctuation in capacity factor due to seasonal weather conditions are not incorporated in the maintenance planning model

Furthermore, the capacity factor also fluctuates during the year. As one can imagine, higher wind speed can yield more power generation. Thus, during the winter months, the capacity factor of an average WT at different locations is expected to be higher compared to the summer months, see Figure 2-22.



Figure 2-22: Monthly national capacity factor (load factor) of offshore wind in 2017 (Pineda, 2017)

• Lead time of spare parts are assumed to be 0 or incorporated in the mobilization time of the JU vessel

Finally, not all spare parts are always available on stock. The spare parts are commonly ordered once the part is needed. A lead time of a few days up to a several weeks is not uncommon. In our research we have decided to not incorporate the lead times for the following reason. In case of PM, the maintenance action is known in advance and the spare part can be ordered in time. In case of CM, the travel time of the JU is expected to be 60 days. In the meantime, it can be assumed that the spare part is ordered and delivered before the JU becomes available.

Thus, we have decided to make several assumptions concerning the literature discussed above. We will not make use of the seasonal price changes in the JU vessel day rate. This is because this information is made confidential by operators and in case it is available it is highly inconsistent. Furthermore, possible waiting times due to bad weather conditions and vessel limits will not be incorporated in our maintenance planning model. The wave heights at FINO1 location were not available be able to incorporate this into the model. The fluctuations in the capacity factor for different months of the year are also not included, as we do not incorporate the seasonal vessel pricing. Finally, the lead time for spare parts are assumed to be 0.

3 Solution methodology

In the following chapter, we will explain how we have applied the theory discussed in section 2. Furthermore, the equations presented in section 2 will be applied in the following chapter. We will first explain how we have determined the failure rates of the converter. From the failure rates, we can determine the reliability which we use as an input to calculate the optimal fixed operational time to schedule PM. Finally, the parameters used for the sensitivity analysis will be discussed.

We have made 3 different excel files. The first will be used to calculate the failure rates and the reliability of the converter system. The second will be used to calculate the objective function from which the optimal fixed operational time to schedule PM can be found. Finally, the third excel file will be used to apply the sensitivity analysis. All the details of the calculations can be found in Appendix B.

3.1 Failure behavior

The failure behavior of the converter system can be represented with the use of six failure categories determined by Fraunhofer. Some failure categories are time dependent (e.g. early failure, aging failure), however several failure categories can be assumed to be random and thus constant over time (e.g. icing, lightning). Other failure categories are dependent on external factors such as wind speeds, climatic conditions. The six different categories are explained below (Faulstich, Berkhout, Mayer & Siebenlist, 2016).

- **Early failures**: failures that occur at the beginning of the start-up phase, for example, due to installation or manufacturing errors of individual components.
- Random failures: failures which are constant risk based
- Aging failures: failures of a component that occur due to aging effects, e.g. corrosion.
- Fatigue failures: failures that occur because of fatigue or wear.
- **Overload failures**: failures of a component due to an overload of a component.
- Component specific failures: this category allows modelling additional effects. This could also comprise natural events such as lightning strikes, earthquakes and damage due to bird strike

Below, the failure rates determined by Fraunhofer of the different failure categories are shown. The early and aging failure are time dependent, but the random, icing, and lighting are constant. Furthermore, the overload failure is dependent on the wind speeds and the wear-out failure is dependent on the total energy produced until a given point in time.

Early failure:
$$\lambda(t) = 0.16668 * \left(\frac{t}{1.7999}\right)^{-0.7}$$
 (22)

Aging failure:
$$\lambda(t) = 0.1884875 * \left(\frac{t}{13,2635}\right)^{1.5}$$
 (23)

- Random failure: $\lambda(t) = 0.08$ (24) Lightning failure: $\lambda(t) = 0.05$ (25)
- *Icing failure*: $\lambda(t) = 0.05$

Overload failure: $\lambda(Vwind) = 0,01171838 * (Vwind - 3,5)$ (27)

Wear out failure: $\lambda(t) = 0.015504753 * (\frac{E}{154.7912416})^{1,4}$ (28)

(26)

We have modelled the different failure categories for 25 years to be able to determine the reliability of the converter system. As stated above, some failure categories are dependent on external factors such as wind. The used wind speed data is collected from the FINO1 location. The wind speeds are measured every 10 minutes. We could determine the failure rate and probabilities for every 10 minutes over the 25-year span, however this would lead to very slow calculation times in excel. Furthermore, it will not be necessary to know exactly at which time of day the maintenance actions should be executed. Nevertheless, to come to a good approximation of the overload and wear-out failure, we have determined the hourly average wind speed. The determination of the hourly average wind speeds is discussed in section 2.3.4. This yields hourly (overload and wear-out) failure rates over the life span of the WT. This also implies that we should use the same interval (hours) to calculate the other failure rates.

As it can be seen in equations (22) and (23), the early and aging failure rates are dependent on time. The "t" in the equation represents years. Because we have decided to calculate the failure rates in hours, we need to convert hours to years. One year exists of 8760 hours, so one hour is exactly 0,000114155 years. To determine the failure rate at a certain point in time, we simply need to fill in the same number of hours (for that point in time) in the equation. Say we want to know the aging failure rate after half a year, this would imply that we fill in 0,000114155 * $\binom{8760}{2} = 0,5$ in equations (22) and (23).

For equations (24), (25), and (26) it is fairly simple. Because the rates are not time dependent, the failure rate at any given point in time will be equal to its own failure rate. Thus, icing and lightning are equal to 0,05 and random is equal to 0,08 per year.

As previously discussed, the overload failure is dependent on the wind speed. Thus, to determine equation (27), we need to fill in the measured wind speed into the equation. The FINO1 wind speed measurements are extrapolated to simulate wind speeds for 25 years. So, to determine the overload failure at a certain point in time, we fill in the measured wind speed in equation (27). Note that 3,5 is subtracted in the equation. This is due to the cut-in wind speed of the WT, explained in section 2.3.4.

Finally, the wear-out failure is dependent on the energy produced. The order of magnitude in equation (28) is measured in GWh (1.000.000.000 Wh). Since it is a value for the produced energy it is measured in Wh. The energy produced can be determined from the wind speeds using the WT power curve. No power will be produced at wind speeds lower than 3,5 m/s or higher than 30 m/s, see section 2.3.4. Wear-out failure represents the deterioration of the turbine as time passes and more energy is produced. Therefore, to calculate the failure rate at a certain point in time, we fill in the cumulative energy produced, until that given point in time, in equation (28). The cumulative energy produced is measured in kWh, so we must divide it with 1.000.000 to convert it to GWh. This yields a failure rate per GWh. For our analysis we need to convert this to a failure rate which is time dependent. To achieve the previous, we need to multiply the failure rate (per GWh) with the energy produced per year. The energy produced per year is the summation of all the energy produced in that corresponding year. The energy produced over a 25 year-span is 23,04 GWh.



Figure 3-1: Yearly energy produced in GWh

With the calculations above, we determine the failure behavior of the converter system. As explained in the theoretical framework, not all failures will lead to a replacement of the component. Both the failure categories as well as the maintenance actions contribute to the overall failure behavior of the converter system, as depicted in Table 3-1. The failure categories are respectively: early, aging, random, lightning, icing, overload, and wear-out.

	Early (22)	Aging (23)	Random (24)	Lightning (25)	lcing (26)	Overload (27)	Wear- out (28)
Reset	5	4	4.5	4	4	3.5	5
Minor	5.7	8.3	7.3	4.7	4.7	5	11.3
Major	1	4	4	1	1	3	6
Replacement	0.3	0.7	0.2	0.3	0.3	0.5	0.7

Table 3-1: Share of maintenance actions and failure categories on overall failure behavior (Fraunhofer, 2017)

These shares must be incorporated in the failure rate calculations of the different failure categories. This implies that equation (22) is multiplied with a factor of $\frac{0,3}{12}$, equation (23) with $\frac{0,7}{17}$, equation (24) with $\frac{0,2}{16}$, equation (25) with $\frac{0,3}{10}$, equation (26) with $\frac{0,3}{10}$, equation (27) with $\frac{0,5}{12}$, and equation (28) with $\frac{0,7}{23}$.



Figure 3-2: Average failure rates per year of the converter system

3.2 Reliability

We have now determined the failure rates for the different failure categories for each hour over a 25-year span. Using equation (3) and the calculated failure rates, we can determine the reliability of the converter system.

To determine the reliability at a given point in time, all the failure rates until that given point in time should be added together and filled in into equation (3). Thus, $\lambda(t)$ represents all the different hourly failures rates added up until time t.

$$\lambda(t) = \lambda eary(t) + \lambda aging(t) + \lambda random(t) + \lambda icing(t) + \lambda lightning(t) + \lambda wearout(E) + \lambda overload(V wind)$$

Subsequently the $\lambda(t)$ is inserted into equation (3). The "dt" from the equation should be consistent with the time increments of the failure rates thus dt = 0,000114155. The number "e" from the equation is also called the Napier's constant and e = 2,718281828459.

The time unit used in our maintenance planning model is years. Thus, we have determined the reliability for every 0,25 years for a 25 year-span. The reliability of the converter system is depicted in Figure 3-3.



Figure 3-3: Reliability of the converter system

3.3 Preventive and corrective costs

3.3.1 Vessel cost

With the information from section 2.3.2, the vessel cost for major replacements can be determined. For PM no pre-inspections are needed because the maintenance is scheduled in advance and the maintenance action is known. In addition, the mobilization time can be assumed to be almost 0 because the JU vessel is scheduled in advance and will be available at the moment of PM.

The mobilization cost is amongst others, the cost associated with planning and preparing a marine operation before the vessel arrives at the wind farm (Dinwoodie, Enderud, Hofmann, Martin & Sperstad, 2014). In case of scheduled PM, the mobilization costs will be smaller, and it is estimated to be between 57 000€ and 114 000€ (Steenbergen, Gelder, Miraglia & Vrouwenvelder, 2014). We have used a mobilization cost of 114.000€ for PM.

In case of CM, the vessel cost will be much higher. A pre-inspection with a CTV, at a day rate of 3500€, will be needed which takes approximately 8 hours. During this time the WT is at a standstill. After the pre-inspection, the failure cause is assumed to be found and the planning and preparation of the marine operation can start. Due to the mobilization time, the WT will have a standstill for an additional 60 days until the JU arrives at the site location and starts the maintenance action. Additionally, a mobilization costs of 500 000€ will be added to the CM cost, see Table 2-4.

For both PM and CM, the day rate of the JU vessel will be estimated at 140 000€ (5833,33 € per hour). According to Fraunhofer, the average duration of the maintenance action at site location will be 57 hours. However, the JU vessel needs to travel to the site from shore and eventually travel back. The JU can travel at a speed of 7 knots. 1 knot is equal to 1,852 km/h and thus the total travel time from shore to site will be as followed.

Total travel time from shore to site
$$(h) = \frac{60}{7 * 1,852} = 4,6$$
 hours

In the above formula, the numerator represents the distance from shore to the center of the wind farm. We will round up the travel time up to 6 hours because the WTs are not situated next to each other but have a distance of 3-10 rotor diameter (378- 1260 meter) between each other. Thus, the travel time cost (from shore to the site and back) will cost $12 * 5833,33 = 70000 \in$.

In conclusion the PM and CM vessel costs are as followed.

Vessel cost (€)

= cost inspection vessel (CTV) + mobilization cost + (hourly rate JU * maintenance time) + (hourly rate JU * travel time)

PM vessel cost = 0 + 114000 + (5833,33 * 57) + (5833,33 * 12) = 516500€

CM vessel cost = 3500 + 500000 + (5833,33 * 57) + (5833,33 * 12) = 906000€

As one may have noticed, the WT does not work when CM is needed, or when maintenance actions are being executed. During this time, the wind farm operator is losing money. In section 3.3.2 we will discuss the downtimes incurred during PM and CM.

3.3.2 Downtimes

We will now determine the downtime of the WT in case of PM and CM. In case of PM, the WT will only have a standstill during the maintenance operation time. Thus, the downtime during PM will be equal to 57 hours. During the travel time to the wind farm, the WT is still operational. Once the component has been replaced, the WT is working again.

However, during the CM more downtime is observed. CM is executed once the WT has stopped working. A CTV with two technicians on board is sent to the WT to inspect and determine the failure cause. The total travel time of the CTV from shore to the wind farm and back is calculated in section 3.3.4 and is equal to 3 hours. After the inspection, the operator will contact maintenance providers and search for an available JU vessel. As described in section 3.3.1, the mobilization time of a JU vessel is approximately 60 days (1440 hours). As mentioned in section 2.3.2, the mobilization time here is understood as the total length of time from when the need for the vessel is identified to the ordered vessel becoming available in the wind farm, including waiting for a vessel to become available on the market, travel time to the maintenance base of the wind farm, waiting for it to be reequipped, etc. (IEA Wind task 26, 2016). After the CM, the WT is assumed to be operational again.

Thus, the downtime during CM can be calculated as followed.

CM downtime = *travel time CTV* + *inspection downtime* + *mobilization downtime*+ maintenance downtime

$$CM \ downtime = 3 + 8 + 1440 + 57 = 1508 \ hours$$

3.3.3 Downtime cost

With the information of section 2.3.3 and 2.3.4, and the information gathered above, we can now determine the downtime cost for PM and CM using equation (20).

In both CM and PM, the number of WT is equal to 1, the maximum power generated is 5000kW, the capacity factor is equal to 41,1% and the cost of electricity is 0,10€ kW/h. Downtime of PM is equal to 57 hours and downtime due to CM is equal to 1508 hours.

This yields the following.

PM downtime cost(€) = 1 * 5000 * 0,414 * 0,10 * 57 = 11799€*CM* downtime cost(€) = 1 * 5000 * 0414 * 0,10 * 1508 = 312156€ It is important to note that downtime cost estimations can substantially deviate in the real world. As we have explained in section 2.3.4, the capacity factor is dependent on the average power produced, which in turn is determined from the measured wind speeds. As one may imagine, it is possible that lower wind speeds are measured during the downtime of the WT. Consequently, the average power produced will be lower which will result in a lower capacity factor and thus yield a lower downtime cost. On the contrary, it is also possible that higher wind speeds are measured during the downtime which will result in a higher capacity factor and thus yield a lower downtime which will result in a higher capacity factor and thus yield a higher downtime cost.

3.3.4 Labor cost

For the JU vessel, all the labor cost of technicians is included in the day rate. This is not the case for the CTV day rate. For a pre-inspection, a CTV is hired, and 2 technicians are sent to the WT to determine the failure cause. The duration of a pre-inspection takes approximately 8 hours, see Table 3-2. These technicians are not scheduled in advance and they are more expensive than scheduled man hours. One unscheduled man hour cost $80 \notin h$ (Puglia, 2013). Additionally, the technicians need to travel to the WT and back to shore. A CTV can travel at a speed of 22 knots (41 km/h). The travel time from shore to the site will then be $\frac{60}{22*1,852} = 1,47$ hours. Thus, traveling from shore to the site and back will take around 2,94 hours. We will round this up to 3 hours as we did in estimating the travel time of the JU in section Vessel cost3.3.1. We have multiplied the distance to the wind farm by two because the CTV must head back to shore after the inspection.

These costs are only incurred during CM because no inspection is needed with PM. Thus, the labor cost for a pre-inspection will be as followed

Labor cost (€) = number of technicians * (travel time + duration of inspection) * labor cost per hour

Labor cost =
$$2 * (6 + 8) * 80 = 2240 \in$$

Maintenance type	Expected number of annual events	Repair time [h]	Number of technicians required	Cost (€)	Vessel requirement
Pre-inspection for major repair	0.3	7.5	2	-	сту
Pre-inspection for major replacement	0.11	7.5	2	-	сти

Table 3-2: Data of pre-inspection (IEA Wind task 26, 2016)

3.3.5 Overview PM and CM costs

Previously, we have determined all the costs incurred during PM and CM replacement of the converter system. The only cost which must be added to both PM and CM cost is the spare part cost which is 13.000€. Below is an overview of all the costs for each maintenance strategy, depicted in Figure 3-4. The total cost of PM is 541.213€ and CM is 1.230.654€. This is also consistent with reports that state that the costs of CM are a factor of two higher than for PM costs (Rademakers, Braam, Zaaijer & Bussel, 2003). We see that the vessel cost is responsible for 95% of the total PM cost. In case of CM, the vessel cost is responsible for 74% and the downtime cost is responsible for 25% of the total CM cost.



Overview maintenance costs & ratios on total cost

Figure 3-4: Overview maintenance costs and ratios of total PM or CM cost

3.4 Optimal time interval calculation

We have almost all the necessary data to determine the optimal fixed operational time to schedule PM. The PM and CM cost are determined in section 3.3, the reliability is calculated in section 3.2. To determine the optimal fixed operational time t to schedule PM, we first need to determine the unreliability. Once the unreliability is known, we can determine the ECC using equation (15). Furthermore, we will use equations (16) or (17) to compute the ECL.

The unreliability F(T) is simply 1 minus the reliability $\overline{F}(T)$ as explained in section 2.2.2.

To determine the ECL we use equation (17). Here we have an integral form 0 to T (wanted point in time) over the reliability function, which we have previously determined. Our used time unit is years. Using equation (18), the trapezoidal rule, we determine small areas under the reliability curve. Each area is a quarter of a year which implies that "dt" must also be equal to 0,25.

Now we have all the needed variables to compute the objective function g(t). C_0 is the corrective cost, C1 the corrective cost, $\overline{F}(T)$ is the reliability, F(t) the unreliability, and ECL the expected cycle length. Inserting these values into the equation, the objective function is determined using equation (14).

3.5 Benchmark policy

Currently operators perform PM for some maintenance actions, but they do not use PM to replace components prior to a failure. However, CM is used to replace components upon failure. As stated in section 2.2.2, MTTF describes the mean time expected until the first failure for a non-repairable system. In our case, we can consider the components as non-repairable systems because the components are replaced and not repaired. Thus, determining the MTTF of the converter system will provide us with a benchmark policy which solely performs CM to replace a component.

Using equation (5), the MTTF can be determined by calculating the area under the reliability curve from 0 to infinity. When time reaches infinity, the reliability of a component should be equal to 0. As

can be seen in Figure 3-3, the reliability of the converter system is expected to be 32,37% after 25 years. Since we have not calculated the reliability of the converter system after 25 years, it is not possible to determine the MTTF using equation (5) in this specific case.

3.6 Sensitivity analysis approach

The optimal age interval to replace a component is strongly dependent on the costs of PM and CM. Therefore, we have executed a sensitivity analysis for different cost parameters and analyzed what the influence is on the optimal fixed operational time to schedule a replacement of the converter system. The costs inputs with the most variability and influence on the total cost are the most interesting to analyze. Thus, we have analyzed different cost inputs for the mobilization time and mobilization cost.

The used inputs for the CM mobilization times are 30, 45 and 60 days. As explained in the theoretical framework, 60 days is considered optimistic but not unreasonable for short-term-on-demand charters in the European market (IEA Wind task 26, 2016). Therefore, we will use a mobilization time of 60 days as the upper limit in our sensitivity analysis. This value may be lower, and 30 to 45 days is also considered reasonable for the European market. We will not use any other mobilization time inputs for PM because these maintenance actions are planned in advance and it can be assumed that the mobilization time negligible. Note that decreasing the mobilization time will have a big influence on the total CM cost, namely the downtime cost will simultaneous decrease.

We used three different inputs for the CM mobilization cost, $250.000 \in$, $500.000 \in$, $750.000 \in$. The mobilization cost is dependent on the planning and preparation of the marine operation, travel distance to the wind farm and transport method. It has a high degree of variability and therefore we have used 3 different inputs categories and named them: optimistic, baseline, and pessimistic. For PM the mobilization cost inputs are respectively: $57.000 \in$, $114.000 \in$, and $171.000 \in$. All the different inputs used are summed up in Table 3-3 and Table 3-4.

Inputs (€)	Optimistic	Baseline	Pessimistic
CM mobilization cost	250.000	500.000	750.000
PM mobilization cost	54.000	114.000	171.000

Table 3-3: CM and PM mobilization cost inputs for each category (optimistic, baseline and pessimistic)

Table 3-4: CM mobilization time inputs

Inputs (days)			
CM mobilization time	30	45	60

In conclusion, we use three different mobilization times in combination with the three different mobilization cost categories. This yields 9 configurations to determine the optimal fixed operational times to schedule PM. The inputs are inserted in the excel file and the output of the objective function is plotted in a graph. The 9 different configurations are summed up in Table 3-5.

Table 3-5: Configuration inputs for the sensitivity analysis

Configuration	CM mobilization time (days)	CM mobilization cost (€)	PM mobilization cost (€)
30 Optimistic	30	250.000	54.000
45 Optimistic	45	250.000	54.000
60 Optimistic	60	250.000	54.000
30 Baseline	30	500.000	114.000
45 Baseline	45	500.000	114.000
60 Baseline	60	500.000	114.000
30 Pessimistic	30	750.000	171.000
45 Pessimistic	45	750.000	171.000
60 Pessimistic	60	750.000	171.000

We have decided not to analyze the day rate of the JU vessel for the following reason. The same day rate is used for PM as for CM. When changing the day rates, it should be done for both maintenance actions. Dependent on the day rates they will simultaneous increase or decrease but the ratio between PM cost and CM cost will stay the same. The influence on the objective function is that it will shift to the right if the day rates are increased and shift to the left if they are decreased. However, the curve of the objective function will be the same but respectively with higher or lower costs.

In addition to a cost sensitivity analysis, we have analyzed one of the assumptions we have made in section 2.3.5. We decided not to include major repairs in the planning maintenance model. However, JU vessels are usually hired for a month and used to execute several other maintenance actions on the same or different WTs. JU vessels are hired several times during the lifetime of the WT. Therefore, it is interesting to analyze what the optimal fixed operational time to replace the converter system would be if we include major repair into our maintenance planning model. This optimum can possibly align with other moments when a JU vessel is hired, and (other) components are needed to be replaced. This yields the possibility to perform opportunistic maintenance on the converter system as suitable maintenance resources are already located at site.

Thus, we have added the contribution of the major repairs to the failure rates after which the new overall reliability of the converter system is determined. From Table 3-1 we can determine the new shares with which the failure rates are multiplied. This implies that equation (22) is multiplied with a factor of $\frac{1,3}{12}$, equation (23) with $\frac{4,7}{17}$, equation (24) with $\frac{4,2}{16}$, equation (25) with $\frac{1,3}{10}$, equation (26) with $\frac{1,3}{10}$, equation (27) with $\frac{3,5}{12}$, and equation (28) with $\frac{6,7}{23}$. The new average failure rates per year for each failure categories are depicted in Figure 3-5 and the new reliability of the converter system is depicted in Figure 3-6.



Figure 3-5: Average failure rates per year of the converter system (major repair & replacement)



Figure 3-6: Reliability of the converter system (repair & replacement)

Research results 4

In the following chapter, the results of the optimal maintenance planning model are discussed. First, the optimal fixed operational time to schedule PM for the converter system is presented and based on the reliability and maintenance cost determined in section 3.2 & 3.3. Thereafter, we will analyze the results from the sensitivity analysis. We will end this chapter with another case study which is executed on the rotor system of a WT.

Optimal fixed operational time interval 4.1

With the parameters determined in section 3, the objective function is calculated and plotted in excel, as depicted in Figure 4-1. The parameters used are a PM cost of 541.213,5€ and CM cost of 1.230.654€, combined with the reliability and the ECL of the converter system.



Optimal fixed operational time for PM

Figure 4-1: Optimal fixed operational time to schedule PM for a converter system

As shown in Figure 4-1, the optimal fixed operational time to schedule PM is found at 21,75 years with a cost per unit time of 56.545,77€. At this point in time, the costs are minimized after which the cost function starts to increase again. If the converter system has not failed prior to 21,75 years, PM actions are executed, and the converter system is to be replaced. If the converter system breaks before the found optimum (21,75), CM actions are executed, and the next PM is scheduled in 21,75 years. We expect that WTs have a lifetime of 20-25 years, thus the PM action will probably only be scheduled once during the WT lifetime. This is due to the high reliability of the converter system. The reliability of the converter system at 21,75 years is 42,69%.

Moreover, we see from Figure 4-1 that the objective function is relatively flat while we would expect a bigger increase at the end of the objective function. This is because the ECC has not yet reached its maximum. When the objective function reaches infinity, we expect the ECC to be equal to the CM cost and the ECL to be equal the MTTF. The reasoning behind this is that while time progresses, reliability decreases and eventually will become 0. By then, we expect a failure of the component and CM actions will take place.

Using equation (19), the effectiveness of using an age-based maintenance policy can be determined. However, to be able to compute equation (19), the g (∞) must be known. As stated previously, when the objective function reaches infinity, the ECC must be equal to the CM cost and the ECL must be equal to the MTTF. The ECC after 25 years is 1.007.499€ and thus not equal to the CM cost (1.230.654€). Furthermore, the ECL at 25 years is also not equal to the MTTF. Nevertheless, we will use the cost per unit time at 25 years to estimate the effectiveness of using an age-based maintenance policy. However, this estimation is an underestimation of the effectiveness of using an age-based maintenance policy. Equation (19) yields us an effectiveness of 1,0024 (equivalent of 0,24%). The low effectiveness of using an age-based maintenance policy is due to the high expected lifetime of the converter system.

4.2 Sensitivity analysis

4.2.1 Different configurations of maintenance costs

For our sensitivity analysis we have inserted 9 different configurations into our optimal maintenance planning model and plotted the results in a graph. The total PM and CM costs for the 9 different configurations are summed up in Table 4-1.

Configuration	PM cost (€)	CM cost (€)
30 Optimistic	481.213,5	832.694
45 Optimistic	481.213,5	906.674
60 Optimistic	481.213,5	980.654
30 Baseline	541.213,5	1.082.694
45 Baseline	541.213,5	1.156.674
60 Baseline	541.213,5	1.230.654
30 Pessimistic	595.213,5	1.332.694
45 Pessimistic	595.213,5	1.406.674
60 Pessimistic	595.213,5	1.480.654

Table 4-1: PM & CM cost for the different configurations

The objective functions for the 9 different configurations are depicted in Figure 4-2. However, it is hard to see where exactly the minimum is found and therefore we have summed them up in Table 4-2.



Figure 4-2: Optimal objective function g(t) for different configurations

Table 1-2. Ontimal	operational	time t &	cost ner	unit time	for each	configuration
Tuble 4-2. Optimul	operational	linela	cost per	unit time	joi eucii	conjiguration

Configuration	Optimal fixed operational time t	Cost per time unit (€)
30 Optimistic	25	40.449,60
45 Optimistic	25	43.264,72
60 Optimistic	25	46.079,86
30 Baseline	25	51055,41
45 Baseline	25	53.870,54
60 Baseline	21,75	56.545,77
30 Pessimistic	22,75	61.450,17
45 Pessimistic	21,25	64.020,10
60 Pessimistic	21	66.514,55

For the first five configurations, we see that the optimal fixed operational time to schedule PM is found at 25 years. However, their objective functions are still decreasing which implies that the optimum will be more than 25 years. As WTs are expected to have a lifetime of 20-25 years, it would be unnecessary to schedule PM for the converter system.

The last four configurations (60 Baseline, 30/45/60 Pessimistic) are the most interesting to analyze. From Table 4-2, we see that the cost per unit time increases and that the optimal fixed operational time t decreases, with the exception of one outlier. For the 60 Baseline, the optimum is found at 21,75 years at a cost per unit time of 56.545,77€. However, for the 30 Pessimistic we see an increase of the optimal fixed operational time namely at 22,75 years. Hereafter, the optimal fixed operational time decreases again.

To be able to understand why the previous happens, we must look at the difference between PM and CM costs and the ratio between them, summed up in Table 4-3. The ratio between PM and CM cost is further referred to as CM/PM cost ratio.

Configuration	Difference in maintenance	CM/PM cost ratio
	cost (€)	
30 Optimistic	351480,5	1,730404488
45 Optimistic	425460,5	1,884140823
60 Optimistic	499440,5	2,037877158
30 Baseline	541480,5	2,000493336
45 Baseline	615460,5	2,137186157
60 Baseline	689440,5	2,273878978
30 Pessimistic	737480,5	2,239018436
45 Pessimistic	811460,5	2,363309972
60 Pessimistic	885440,5	2,487601508

 Table 4-3: Difference in maintenance cost & ratio between PM and CM cost for each configuration

As it can be seen in Table 4-3, there is a bigger difference in maintenance cost between the 60 Baseline configuration and the 30 Pessimistic configuration. However, the CM/PM cost ratio is lower for the 30 Pessimistic compared to the 60 Baseline. The lower the ratio is, the longer we should wait to schedule PM. On the contrary, when the ratio is higher, it is more beneficial to schedule PM earlier to avoid an expensive CM. Thus, we see that the objective function is not dependent on the difference between PM and CM cost, but on the CM/PM cost ratio. We have run the optimal maintenance planning model several times for the 60 Baseline configuration with a CM cost of 1.230.654€ and adjusted the PM cost according to different CM/PM cost ratios. The results are shown in Table 4-4.

Additionally, using equation (19) and the information at the end of section 4.1, we can determine the effectiveness of using an age-based maintenance policy for each CM/PM cost ratio. The results have been converted to percentages and are depicted in Table 4-4 as well. As the CM/PM cost ratio increase, so does the effectiveness of using an age-based maintenance policy.

Ratio	2	2,5	3	3,5	4	4,5	5	5,5	6
CM €	1.230. 654								
PM €	615.32 7	492. 261	410.21 8	351.61 5	307.66 3	273.47 8	246.13 0	223.75 5	205.10 9
Optimal T* (years)	25	21	17,75	16	15	14	14	13,75	13,25
Cost/ time (€)	58.035	55.2 15	52.315	49.764	47.585	45.747	44.166	42.856	41.740
Effectiveness %	0	1	3,8	7	10,2	13,2	16,2	18,8	21,1

Table 4-4: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios

4.2.2 Optimal fixed operational time interval (major repair & replacement)

The parameters used for our model are the failure rates and reliability determined in section 3.6. Furthermore, we have used the 60 Baseline configuration to determine the optimal fixed operational time to schedule PM. We have a PM cost of 541.213,5€ and CM cost of 1.230.654€ (CM/PM cost ratio of 2,27). This yields the following objective function, depicted in Figure 4-3.



Figure 4-3: Optimal fixed operational time to schedule PM of a converter system (repair & replacement)

We find an optimum at 9,5 years with a cost per unit time of 161.679,47€. Thus, after 9,5 operational years we should schedule PM and replace the converter system. However, the reliability used in this model represents the major repair and replacement failures. Thus, the component can still be in good working condition or in need of major repair when it is replaced.

If the system fails prior to 9,5 years a CTV should be sent to the WT to assess the failure cause and to determine what type of maintenance actions are needed. As explained in section 2.3.1, major repairs can be executed with the use of a CTV, which is cheaper compared to a replacement with a JU vessel. This is due to high mobilization time, mobilization cost, and day rate of the JU vessel. Thus, if the component can be repaired, a CTV with 4-6 technicians should be sent to the WT to repair the component

Nevertheless, it is useful to model the combined reliability of major repairs and replacement for the following reason. WTs consist of many different components with each having their own reliability and maintenance cost. Running the optimal maintenance planning model for these other components can yield an optimum which is also approximately 9,5 years. Consequently, if both components are to be found operational for 9,5 years, PM actions can be performed simultaneously on both components. Thus, the converter system can possibly be replaced and reconditioned offsite using opportunistic maintenance. The reconditioned component can be considered to be "as good as new" and can eventually be used for PM on another WT.

We have run the optimal maintenance planning model for different CM/PM cost ratios and used the 60 Baseline configuration to determine the CM cost. The PM cost is calculated from the different CM/PM cost ratios. The results are summed up in Table 4-5. From Figure 4-4, we can see that the objective functions resemble the cost function more than we would expect. This is due to the higher CM/PM cost ratio but especially because of the lower reliability of the converter system. At 9,5 years the reliability is expected to be 29,47% and we expect a reliability of 0,01% after 25 years.

As the ECC is approximately equal to the CM cost (1.230.541€ versus 1.230.654€) and the reliability approaches 0 at 25 years, we can assume that objective function has reached infinity. Therefore, using equation (5), we can determine the MTTF which is estimated to be 7,3 years. Thus, when the objective function reaches infinity, the cost per unit time is estimated to be 168.356€. Using equation

(19), and the information above, we can determine the effectiveness of using an age-based maintenance policy. The effectiveness in % of the CM/PM cost ratios are summed up in Table 4-5.

Ratio	2	2,5	3	3,5	4	4,5	5	5,5	6
CM €	1.230.65 4								
PM €	615.327	492.2 61	410.2 18	351.6 15	307.6 63	273.4 78	246.1 30	223.7 55	205.1 09
Optimal T* (years)	11	9	7,75	7	6,75	6,5	5,75	5,5	5
Cost/time (€)	164.295	159.0 77	153.3 75	148.0 87	143.6 77	140.0 58	136.4 59	133.2 10	130.2 69
Effectiveness %	2,5	5,8	9,8	13,7	17,2	20,2	23,7	26,4	29,2

 Table 4-5: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios (repair & replacement)



Figure 4-4: Optimal objective function g(t) for different CM/PM cost ratios (repair & replacement)

From Figure 4-4, we see an abrupt change in the objective function at around 5 years. This same abrupt change can be noted in the reliability curve of the converter system, depicted in Figure 3-6. As reliability is determined from the different failure rates categories, the abrupt change in the reliability curve can be traced back to fluctuations in failure rates, depicted in Figure 3-5. We see these fluctuations especially with the wear-out as well as the overload failure rate. Since wind speed fluctuates over time, so will the failure rates which are dependent on wind speeds. Furthermore, the wear-out failure rate is also dependent on the energy produced each year, depicted in Figure 3-1. Thus, if less energy is produced compared to a previous year, a slight decrease in the failure rate can be seen.

To conclude, assuming the reliability is unchanged, the objective function is strongly dependent on the CM/PM cost ratios. As CM/PM cost ratio increases, the optimal fixed operational time to schedule PM decreases. Thus, the lower the ratio is, the longer we should wait to schedule PM. On

the contrary, when the ratio is higher, it is more beneficial to schedule PM earlier to avoid expensive CM. Using the reliability information based on solely replacement actions, PM should be scheduled to replace the converter system. Depending on the CM/PM cost ratios, the found optimums range in between 25 and 13,25 operational years. Furthermore, the usage of an age-based maintenance policy becomes more favorable as the CM/PM cost ratios increases.

However, when we determine the optimums based on the reliability information which considers major repairs and replacements, we find optimums in between 11 and 5 operational years. It is important to note that replacing the converter system based on the reliability information considering major repair and replacement, the converter system may still be in good working condition. Thus, the found optimums should be used to perform opportunistic maintenance when maintenance resources are already allocated at site.

4.3 Case study: rotor system

With the use of our model, we have analyzed the optimal fixed operational time to schedule PM for a rotor system of a WT. Only the most important steps and calculations will be discussed. We have determined the optimum for two different cases of the rotor system. First, the optimum is determined using the reliability information based on replacement maintenance actions only. Thereafter, the optimum is determined using the reliability information based on major repairs and replacement.

4.3.1 Optimal fixed operational time interval of the rotor system (replacement)

We first need to determine the reliability of the component to be able to compute the ECC and ECL from which the objective function is calculated. The reliability can be determined from the failure rates of the rotor system. Below, the failure rates determined by Fraunhofer of the different failure categories are shown.

Early failure: $\lambda(t) = 0,29169 *$	$\left(\frac{t}{1,028459149}\right)^{-1}$	-0,7	(33)
		. 1 5	

Aging failure: $\lambda(t) = 0,4545875 * \left(\frac{t}{5,499491297}\right)^{1,3}$	(34)
Random failure: $\lambda(t) = 0,105$	(35)
Lightning failure: $\lambda(t) = 0.05$	(36)
Icing failure: $\lambda(t) = 0.05$	(37)
Overload failure: λ (Vwind) = 0,015624506 * (Vwind - 3,5)	(38)
<i>Wear out failure</i> : $\lambda(t) = 0,054603696 * (\frac{E}{43,95306859})^{1,4}$	(39)

Just like the converter system, the failure rates must be multiplied with a certain share. These shares represent the probability that one of the maintenance actions are needed when a failure occurs, depicted in Table 4-6. We are only interested in replacing the rotor system for the same reasons as the converter system. Therefore we must multiply equation (33) with a factor of $\frac{0.1}{10.5}$, equation (34) with $\frac{0.5}{20.5}$, equation (35) with $\frac{0.1}{10.5}$, equation (36) with $\frac{0.1}{5}$, and equation (37) with $\frac{0.1}{5}$, equation (38) with $\frac{0.1}{8}$, and equation (39) with $\frac{2}{40.5}$.

	Early (33)	Aging (34)	Random (35)	Lightning (36)	lcing (37)	Overload (38)	Wear-out (39)
Reset	2	5,5	2	1.4	1.4	2.9	8.8
Minor	5.9	11	5.9	2	2.5	4.5	23.3
Major	2.5	3.5	2.5	1.5	1	0.5	6.5
Replacement	0.1	0.5	0.1	0.1	0.1	0.1	2

Table 4-6: Share of maintenance actions and failure categories on overall failure behavior rotor system (Fraunhofer, 2017)

Using equation (14), the model calculates the reliability for time t over a 25 year-span. The reliability of the rotor system is depicted in Figure 4-5.



Figure 4-5: Reliability of the rotor system (only considering replacement)

We now also know the unreliability of the component, which is 1 minus the reliability. Using equations (17) and (18), the ECL is calculated. The last two unknowns to be able to compute the objective function are the PM and CM costs. Similarly to the converter system, the rotor system uses a JU vessel to replace the component. Thus, the day rate, mobilization time, and mobilization cost are unchanged. However, we need to adjust several other inputs. From Table 4-7 we see that we must change the reparation duration, which currently is 153 hours. This will also change the downtime cost for PM and CM. Furthermore, we must adjust the spare part cost to 52.000€. This yields a PM cost of 1.159.941,5€ and CM cost of 1.849.382€ (CM/PM cost ratio of 1,594375). Note that the CM/PM cost ratio is lower compared to the converter system.

Table 4-7: Information of rotor system: maintenance actions, repair duration, technicians, vessel type and material costs (Fraunhofer, 2017)

Rotor System (=MDA)	Reset	3	2	сти	
	Minor Repair	9	2	сти	196
	Major Repair	19	3	Crane	1879
	Replacement	153	12	JU	52000

The objective function is now calculated in the optimal maintenance planning model and is depicted in Figure 4-6. With a PM cost of 1.159.941,5€ and CM cost of 1.849.382€ (CM/PM cost ratio of

1,594375), we find the optimal fixed operational time to schedule PM at 13 years. The cost per unit time is equal to 197.198€. Thus, if the system is still operational after 13 years, the rotor system should be replaced with PM. If the system does fail prior to 13 years, CM will take place and the next PM is scheduled in 13 years.



Figure 4-6: Optimal fixed operational time to schedule PM of the rotor system (replacement)

According to the parameters used, the optimum we found is correct. However, according to Rademakers et al. the CM/PM cost ratio for offshore WTs is estimated to be 2 or higher. The CM/PM cost ratio from the above found optimum is 1,594375. We have run the model several times for the same CM cost and a PM which is determined from different CM/PM cost ratios. The results are summed up in Table 4-8 and depicted in Figure 4-7. In Figure 4-7, we see an abrupt change in the objective functions. This is caused by the abrupt change in the reliability curve at around 5 years, which can be traced back to fluctuations in the failure rates, as explained in section 4.2.2.

Additionally, as the reliability approaches 0 at 25 years, we can assume that the reliability has reached infinity. Thus, we can use equation (5) to determine the MTTF of the rotor system. The MTTF of the rotor system is estimated to be 9,2 years. As explained in section 4.1, as time reaches infinity, the ECC should be equal to the CM cost and the ECL equal to the MTTF. The ECC at 25 years is 1.849.066€ and the ECL is equal to 9,2 years, which indicates the objective function has reached infinity. The cost per unit time is estimated to be 200.915€. Consequently, using equation (19), we can determine the effectiveness of using an age-based maintenance policy. The effectiveness in % of the CM/PM cost ratios are summed up in Table 4-8.

Ratio	2	2,5	3	3,5	4	4,5	5	5,5	6
CM €	1.849.38 2								
PM €	924.691	739.7 52	616.4 60	528.3 94	462.3 45	410.9 73	369.8 76	336.8 76	308.2 30
Optimal T* (years)	9,5	7,75	7	5,75	5,25	5	5	5	5
Cost/time (€)	189.052	175.4 40	162.4 58	151.6 83	141.1 33	131.8 90	124.3 36	118.1 54	113.0 04
Effectiveness %	6,3	14,5	23,7	32,4	42,3	52,3	61,6	70	77,8

Table 4-8: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios (replacement rotor system)



Figure 4-7: Optimal objective function g(t) for different CM/PM cost ratios (replacement of rotor system)

To conclude, using a CM/PM cost ratio of 2 which is underpinned by literature, we find the optimal fixed operational time to schedule PM at 9,5 year with a cost per unit time of 189.052€. If operators would adhere the current maintenance policy, which is a CM policy, the cost per unit time would be estimated to be 200.915€. The effectiveness of using an age-based maintenance policy compared to a CM policy is 6,3%. Moreover, using an age-based maintenance policy instead of a CM policy, 296.575€ in maintenance cost can be saved over the lifetime of one WT. Thus, e.g. for a wind farm with 50 WTs, approximately € 15 million can be saved in rotor system maintenance cost.

Therefore, we can conclude to schedule the PM replacement of the rotor system after 9,5 operational years. Moreover, from section 4.2.2, we also find an optimum at 9,5 years. If PM actions are undertaken for a rotor system and another converter system is found to be operational for 9,5 years, it would be economically sound to replace the converter system as well and recondition it offsite. Combining these two maintenance actions decreases the total logistic cost incurred over the lifetime of the WT and consequently decreases the total maintenance costs. In conclusion, while performing PM replacement on a rotor system, opportunistic maintenance can be used to replace a converter system which has also been operational for 9,5 years.

4.3.2 Optimal fixed operational time interval of the rotor system (repair & replacement) As explained in section 4.3.1, the failure rates must be multiplied with a certain share. These shares represent the probability that one of the maintenance actions are needed when a failure occurs, depicted in Table 4-6. We are now interested in major repair and replacement of the rotor system. Therefore we must multiply equation (33) with a factor of $\frac{2,6}{10,5}$, equation (34) with $\frac{4}{20,5}$, equation (35) with $\frac{2,6}{10,5}$, equation (36) with $\frac{1,6}{5}$, and equation (37) with $\frac{1,1}{5}$, equation (38) with $\frac{0,6}{8}$, and equation (39) with $\frac{8,5}{40,5}$.

Using equation (14), the model calculates the reliability for time t over a 25 year-span. The reliability of the rotor system is depicted in Figure 4-8.



Figure 4-8: Reliability of the rotor system (considering major repair & replacement)

We now also know the unreliability of the component, which is 1 minus the reliability. Using equations (17) and (18), the ECL is calculated. In section 4.3.1, we have determined the PM cost to be 1.159.941,5€ and a CM cost of 1.849.382€ (CM/PM cost ratio of 1,594375).

The objective function is now calculated in the optimal maintenance planning model and depicted in Figure 4-9. With the PM and CM cost stated above, we find the optimal fixed operational time to schedule PM at 16,5 years. The cost per unit time is equal to 371.406€. The effectiveness of using an age-based maintenance policy is 1,0004 (0,04%). As it has been done in previous sections, the optimal fixed operational time to schedule PM is determined for several CM/PM cost ratios. The results are summed up in Table 4-9 and depicted in Figure 4-10.

Additionally, as the reliability approaches 0 at 25 years, we can assume that the reliability has reached infinity. Thus, we can use equation (5) to determine the MTTF of the rotor system. The MTTF of the rotor system is estimated to be 4,97 years. As explained in section 4.1, as time reaches infinity, the ECC should be equal to the CM cost and the ECL equal to the MTTF. The ECC at 25 years is 1.849.360€ and the ECL equal to 4,97 years, which indicates the objective function has reached infinity. The cost per unit time is estimated to be 371.555€. Consequently, using equation (19), we can determine the effectiveness of using an age-based maintenance policy. The effectiveness in % of the CM/PM cost ratios are summed up in Table 4-9 as well.



Optimal fixed operational time for PM

Figure 4-9: Optimal fixed operational time to schedule PM of the rotor system (major repair & replacement)

Ratio	2	2,5	3	3,5	4	4,5	5	5,5	6
CM €	1.849.38 2								
PM €	924.691	739.7 52	616.4 60	528.3 94	462.3 45	410.9 73	369.8 76	336.8 76	308.2 30
Optimal T* (years)	13,5	11,5	10,5	9,5	8,75	8,5	8,5	7,75	7,5
Cost/time (€)	370.125	367.4 16	364.4 54	361.4 44	358.9 45	356.4 32	354.3 44	352.6 67	350.9 00
Effectiveness %	0,4	1,1	1,9	2,8	3,5	4,2	4,9	5,4	5,9

Table 4-9: Optimal fixed operation time t & cost per unit time for different CM/PM cost ratios (major repair & replacement rotor system)



Figure 4-10:Optimal objective function g(t) for different CM/PM cost ratios (major repair & replacement of rotor system)

To conclude, for the cost parameter used (CM/PM cost ratio 1,594375) we find an optimum at 16,5 years with a cost per unit time of 371.406€. However, the effectiveness of using an age-based maintenance policy is low, namely 0,04%. Using a more reasonable CM/PM cost ratio of 2, we find the optimum at 13,5 years with an effectiveness of 0,4%. As the CM/PM cost ratios increases, so does the effectiveness of using an age-based maintenance policy.

However, the reliability used in this model represents the major repair and replacement failures. Thus, the component can still be in good working condition or in need of major repair when it is replaced. Nevertheless, it is useful to model the combined reliability of major repairs and replacement. As explained, WTs consist of many different components. Other components might have an optimum which is close to 13,5 years and which gives the opportunity for operators to replace the rotor system using opportunistic maintenance.

5 Conclusion

In the final chapter, we will conclude the research and answer the research questions set at the beginning of the report. We will first answer the sub questions before answering the main research question.

How can the maintenance of wind turbines be characterized? Can it be modeled by using a wellknown maintenance policy? If so, how can the optimal parameters of the selected policy be calculated with the use of failure behaviors?

The maintenance of wind turbines can mainly be characterized by preventive and corrective maintenance. Corrective maintenance involves the activities undertaken after a failure of the system or component. In some cases, the system can continue working even though a failure has occurred. This is due to installed back-up components or the importance of the component itself. Nevertheless, for the majority of components the system will stop working once a failure has occurred. A failure has economic consequences such as expensive maintenance cost due to difficult accessibility of the wind turbine, labor hours, logistic cost, cost of spare parts or loss in revenue caused by the downtime.

On the other hand, preventive maintenance involves the maintenance activities prior to the failure of the system or component. This strategy can be based on the Original Equipment Manufacturer recommendations. This is suitable for some preventive maintenance activities such as lubrication, oil changes, filter cleaning and so on. However, the focus of the research is on preventive maintenance activities which involve replacing a component prior to a failure even though it hasn't reached the end of its useful life cycle. One of the main objectives of preventive maintenance is to reduce the failure frequency of the system or component. Therefore, it contributes to minimizing failure costs and machine downtime.

The maintenance activities can mainly be classified as: reset, minor repair, major repair, and replacement. A manual reset is a brief inspection or reset of the wind turbine which sometimes involve routine replacements or some consumables. A minor repair is a relatively small repair of the wind turbine needing small spare parts. Reset and minor repairs take less than a day to execute. Major repairs are large repairs and small component replacements needing relatively small spare parts. As the name suggests, replacements are the replacements of large components involving a large crew and a vessel with lifting capacity. These maintenance activities can take up to a few days or weeks to execute. During this time high downtime cost is incurred by the operators of wind turbines.

Therefore, our component of choice is the converter system of an offshore wind turbine. Converter systems converts direct current from the generator into alternating current to be exported to the grid network. It has a high share on the annual downtime of wind turbines which results in high downtime costs. The preventive maintenance cost to replace a converter system on an offshore wind turbine is 541.213,5€. Corrective maintenance cost of the converter system is estimated at 1.230.654€. Furthermore, we have analyzed a second wind turbine component, namely the rotor system, to check the usage of our designed optimal maintenance planning model. The preventive maintenance cost to replace a cost of the convertine is 1.159.941,5€. Corrective maintenance cost of the system.

Thus, to avoid unnecessary failures and high costs, preventive maintenance should be scheduled prior to a failure. This can be accomplished by modeling the planning of preventive maintenance with

an age-based policy, also known as type-1 policy. This policy is especially useful to model the preventive maintenance of one or multiple components. To be able to use this policy to model the preventive maintenance, the failure rate function of the component must be increasing over time. Furthermore, the difference between preventive and corrective maintenance cost should be substantial for the policy to be useful. An age-based policy is defined as a maintenance policy which performs preventive maintenance after an optimal fixed operational time (hours, days, months, years). If a failure occurs prior to the optimal fixed operational time, corrective maintenance will be executed, and the next preventive maintenance will again be scheduled after the optimal fixed operational time. The age-based policy considers the system to be "as good as new" after the maintenance activities.

The optimal fixed operational time to replace a component can be derived from the objective function, which is a function over time. The objective function minimizes the maintenance cost and is found from the expected cost per cycle and the expected cycle length. As the name suggests, the expected cost per cycle is the maintenance cost expected during the life cycle of the component. It is determined from the preventive and corrective maintenance cost and the reliability and unreliability of the component. The reliability and unreliability can be determined from the failure behaviors of a component. Failure behaviors are commonly expressed with failure rate functions from which the reliability can easily be calculated. Finally, the expected cycle length is the cycle length of the component until it is expected to fail. It can be determined from the reliability and unreliability of the component.

What is the influence of different maintenance cost configurations on the optimal maintenance policy?

Using different cost configurations has a big influence on the optimal fixed operational time to schedule preventive maintenance. The optimal fixed operational time in our research is found at 21,75 years at a cost per unit time of 56.545,77€. Using different cost configurations, the optimal shifts between an optimal time of 25 years and 13,25 years. The optimal strongly depends on the reliability of the component, and the ratio between preventive and corrective maintenance cost (CM/PM cost ratio). E.g. a CM/PM cost ratio of 2 implies that corrective maintenance cost are two times bigger than the preventive maintenance cost. Assuming the reliability is unchanged, a CM/PM cost ratio of 2 will yield an optimal at 25 years which will make the scheduling of preventive maintenance redundant. On average, the expected lifetime of a wind turbine is estimated to be 20-25 years. A slight increase of the CM/PM cost ratio to 2,5 already has a significant impact on the optimal which shifts to 21 years. Increasing the CM/PM cost ratio even further to 3 yields an optimal at 17,75 years. At a CM/PM cost ratio of 6, we find an optimal fixed operational time to schedule preventive maintenance of the converter system at 13,25 years. Furthermore, the usage of an agebased maintenance policy becomes more effective as the CM/PM cost ratios increases. The effectiveness is estimated to be 1 (0%) for a CM/PM cost ratio of 2 and estimated to be 1,211 (21,1%) for a CM/PM cost ratio of 6.

Finally, our main research question will be answered:

How can maintenance of wind turbines be planned by using reliability information of components such that total downtime, maintenance, and logistics costs are minimized?

Thus, to conclude, the maintenance of wind turbines can be planned using an age-based maintenance policy. This policy schedules preventive maintenance of one or multiple components after fixed operational time (hours, days, months, years). The objective function of the age-based maintenance policy finds an optimal time which minimizes the total maintenance costs and ensures enough reliability of the system or component. The total maintenance cost includes all the downtime, maintenance, and logistics costs incurred during execution of preventive or corrective maintenance.

Furthermore, we have designed an optimal maintenance planning model which can determine the optimal preventive replacement time for numerous wind turbine components, using different cost parameters inputs and reliability information. The designed model minimizes the total downtime, maintenance and logistics costs

6 Recommendations

In the following chapter, we will discuss the final recommendations for the proceedings of future research. From the research and section 2.3.5 a few recommendations arise.

The optimal maintenance planning model should be extended with other seasonal changes such as more variability in the cost determinations, waiting times, and capacity factor.

As previously explained, the day rates of the used JU vessels fluctuate per season. This is due to the more favorable weather conditions during certain months of the year to execute maintenance on offshore wind turbines. During the more favorable months, a higher demand for jack-up vessels can be observed. Consequently, a higher day rate must be paid compared to lower demand months such as during the winter.

The higher demand during the more favorable weather conditions is a direct impact of the wave and wind speeds limits of the JU vessel. Long waiting times in port are not uncommon during the winter months compared to e.g. the summer months. Waiting times in port have a big influence on the total vessel cost and the downtime cost. Consequently, the preventive or corrective maintenance cost can dramatically increase.

Furthermore, adding other seasonal changes to the model would also imply to implement this for the capacity factor. An average capacity factor has been used to determine the downtime cost. However, seasonal capacity factor would be more suitable once the model is adjusted.

Further research the possibility to use k-mean clustering to find more suitable optimums for more than one component.

The model currently determines the optimal fixed operational time to schedule preventive maintenance for one single component. First, the model should be used to analyze the optimal of all other major components. Hereafter, the age-based maintenance policy can be extended with e.g. k-mean clustering. K-mean clustering is where different optimums of several components are grouped together in clusters and a new optimum is determined for the newly defined clusters (Teknomo, 2018). In fact, k-mean clustering is an optimization of the found optimums of each component in a cluster. Due to time restriction and insufficient available data, we were not able to determine the optimums of all major components in an offshore wind turbine.

Execute a sensitivity analysis on the influence of the reliability of a component on the optimal fixed operational time to schedule preventive maintenance.

In the research a sensitivity analysis on the maintenance costs has been carried out. However, the optimal fixed operational time is also strongly dependent on the reliability of a component. Therefore, it would be useful to analyze the influence of a change in reliability and the influence of variability on the optimal time to schedule preventive maintenance. As explained in the report, reliability is dependent on different failure categories which can be caused by internal or external factors. External factors can be factors such as wind speeds which influence the wear-out and overload failure rates. Moreover, failure categories which are dependent on internal factors can change due to e.g. improved manufactured components. Therefore, the overall reliability of the

component can possibly increase or decrease which in terms has an influence on the objective function. For example, it would be useful to analyze and determine when to schedule preventive maintenance if the operators have experienced one or several years with higher measured wind speeds. As information about the parameter's limits of the of the different failure categories were not available to us, we were not able to determine the possible changes in reliability of the converter or rotor system.

Analyze different optimums of major components to seek the opportunity to perform opportunistic maintenance.

Using the optimal maintenance planning model, optimums for all major components should be determined based on reliability information which solely considers major repair and replacement. Replacing components requires expensive equipment with high logistic costs. Thus, while performing corrective or preventive replacement actions, needed maintenance resources are already allocated at site. This yields the opportunity to use the maintenance resources to replace other components which are expected to fail with a short period of time after the allocation of maintenance resources. Thus, using the determined optimums, components can possibly be replaced preventively using opportunistic maintenance.

Research the possibility to implement condition-based maintenance using condition monitoring systems in wind turbines.

As wind turbines increases in size, operation and maintenance procedures need to be optimized to increase reliability, safety and maximize cost effectiveness (A. May & D. McMilan, 2018). The installation of condition monitoring systems into wind turbines becomes more widely spread adopted by operators. This allows operators to possibly adopt condition-based maintenance and to reduce the cost of preventive as well as corrective maintenance actions. Maintenance decisions are based on information of the actual condition monitoring systems. Most wind turbines have an installed condition monitoring system, called SCADA system. The SCADA system measures temperatures, pitch angle, electrical parameters, rotor speed, etc. However, applying CBM, is only possible if there are conditions that are related to the moment of failure and if it is technically possible to monitor these maintenance with the use of condition monitoring systems.

7 References

- 26, I. W. (2016). Offshore wind farm baseline documentation. National Renewable Energy Laboratory.
- A. May & D. McMilan. (2018). Condition Based Maintenance for Offshore Wind Turbines: The Effects of False Alarms from Condition Monitoring Systems.
- (1995). About the Gamma Function.
- Andrew. (2018). *Germany's offshore wind capacity factors*. Retrieved from Energy numbers: http://energynumbers.info/germanys-offshore-wind-capacity-factors
- (2010). BSI Standards Publication: Maintenance Maintenance terminology. BSI.
- Dalgic, L. &. (2013). Vessel charter rate estimation for offshore wind O&M activities .
- Dalgic, Lazakis, Turan & Judah. (2015). *Investigation of optimum Jack-Up vessel chartering strategy* for offshore wind farm O&M activities.
- Dinwoodie, Enderud, Hofmann, Martin & Sperstad. (2014). *References cases for verification of operation and maintenance simulation models for offshore wind farms.* Wind Engineering.
- Faulstich, Berkhout, Mayer & Siebenlist. (2016). Modelling the failure behaviour of wind turbines.
- Galambos, Galambasova, Rataj & Kavka . (2017). *Design of condition-based decision support system for preventive maintenance.* De Gruyter.
- (2017). Global wind services market. Tayor Wessing and Frost & Sullivan.
- (2017). Global wind turbine O&M. MAKE Consulting.
- Guerreiro, J. (n.d.). *Financial assessement and simulation of O&M costs of offshore wind energy projects*. Técnico Lisboa.
- Igba, Alemzadeh, Durugbo & Henningsen. (2015). *Performance assessment of wind turbine gearboxes using in-service data: Current approaches and future trends.* Elsevier.
- IRENA. (2012). *Wind Power, renewable energy technologies: cost analysis series.* International Renewable Energy Agency.
- IRENA. (2018). *Renewable Power Generation Costs in 2017.* Abu Dhabi: International renewable Energy Agency.
- Jonge, B. d. (2017). Maintenance optimization based on mathematical modeling.
- Kaiser & Snyder. (2010). Offshore Wind Energy Installation and Decommissioning Cost Estimation in the U.S. Outer Continental Shelf. Energy research group.
- Kirkeby & Mikkelsen. (2016). *Optimizing Jack-up vessel chartering strategies for offshore wind farms.* Norwegian University of Science and Technology.
- M.Lutz. (2017). Using different data sources to determine the state of the turbine and its reliability. *Presentation.* Prague.
- Maples, Saur & Hand. (2013). *Installation, operation and maintenance strategies to reduce the cost of offshore wind energy*. National renewable energy laboratory.

- Merchant, E. F. (2018, 07 16). Wind Turbines to See 'Unprecedented' Growth in Size and Capacity . Retrieved from Green tech media: https://www.greentechmedia.com/articles/read/windturbine-sizes-to-see-unprecedented-growth#gs.f0oPzO4
- Milborrow, D. (2014, September 30). *The tide turns on offshore maintenance costs*. Retrieved from Wind power offshore: https://www.windpoweroffshore.com/article/1314299/tide-turns-offshore-maintenance-costs
- Nilsson, J. (2005). Maintenance management of wind power systems.
- Ostachowicz, McGugan, Hlnrichs & Luczak . (2016). MARE-WINT New Materials and Reliability in Offshore Wind Turbine Technology. Springer Open.
- Pfaffel, Faulstich & Rohrig. (2017). Performance and reliability of wind trubines: A review.
- Pineda, I. (2017). *Offshore Wind in Europe, Key trends and statistics 2017*. Wind Europe.
- Puglia, G. (2013). Life cycle cost analysis on wind turbines.
- R. Steenbergen, P. van Gelder, S. Miraglia & A. ton Vrouwenvelder. (2014). *Safety, reliability and risk analysis: beyond the horizon.* CRC Press/Balkema.
- Rademakers, Braam, Zaaijer & Bussel. (2003). Assesment and optimisation of operation and maintenance of offshore wind turbines.
- Reddy, T. (2011). Applied Data Analysis and Modeling for Energy.
- RePower. (2004). 5M proven technology in new dimensions. RePower.
- Ribrant, J. (2005). *Reliability performance and maintenance: A survey of failures in wind power systems.*
- Santos, Teixeira & Soares. (2013). *Maintenance planning of an offshore wind turbine using stochastic petri nets with predicates.*
- Scheu, Kolios, Fischer & Brennan . (2017). *Influence of statistical uncertainty of component reliability estimations on offshore wind farm availability.* Hamburg: Elsevier Ltd.
- Sperstad, McAuliffe, Kolstad & Sjormark. (2016). *Investigation key decision problems to optimize the operation and maintenance strategy of offshore wind farms.* Elsevier.
- Stalhane, Vefsnmo, Halvorsen-Weare, Hvattum & Nonas. (2016). *Vessel fleet optimization for maintenance operations at offshore wind farms under uncertainty.* Elsevier.
- Steenbergen, Gelder, Miraglia & Vrouwenvelder. (2014). *safety, reliability and risk analysis: Beyond the horizon.*

Sultana & Karim. (2015). Optimal Replacement Age and Maintenance Cost: A case stuy.

- Teknomo, K. (2018, 08 21). Retrieved from K-mean clustering tutorials: http://people.revoledu.com/kardi/tutorial/kMean/index.html
- Tinga, T. (2013). Principles of Loads and Failure Mechanisms. Springer.

Vermat, Ajit, Karanki. (2016). Rekiability and Safety Engineering.

- Walford, C. (2006). Wind turbine reliability: Understanding and minimizing wind turbine operation and maintenance costs.
- Wind farm data collection and reliability assessment for O&M optimization. (2017). IEA wind.
- Zhao, Li, Dong, Kang, Lv and Shang. (2017). Fault Prediction and Diagnosis of Wind Turbine Generator using SCADA Data.

8 Appendix

A: Technical data 5MW WT

Design	
Rated Power	5,000 kW
Cut-In Wind Speed	3.5 m/s
Rated Wind Speed	13 m/s
Cut-Out Wind Speed	
Offshore Version	30 m/s
Onshore Version	25 m/s
Certification acc. to	TK I, GL Offshore (extended with design for 10.5 m/s
	annual average wind speed)
Rotor	
Diameter	126 m
Speed Range, normal operation	approx. 6.9 – 12.1 rpm
Control	
Principle	Blade Angle and Speed Control, Electrical Pitch
Safety Systems	
3 Individual Fail-Safe Blade Pitch S	Systems
Rotor Locking Brake	
Fully integrated lightning protection	n system with multiple receptor principle in the rotor blades
Gearbox	
Design	Combined Planetary/Spur Wheel Gears
Transmission Ratio	i = approx. 97
Generator	
Design	Doubly Fed Asynchronous Generator, 6 Pole
Speed Range	approx. 670 – 1,170 rpm
Masses	
Rotor	approx. 120 t
Nacelle (without rotor)	approx. 290 t
Next lie (without noter)	

Status October 2004 Subject to technical changes
B: Calculations Excel

Hourly wind speeds

AutoSave 🖲 🗊 🖬 🦘 🕫 - 💷				Reliabili										
File Home Insert Page Layo	ut Formulas Data I	Review View Ad												🖻 Share
fx Insert Function	ical Text Date & Lookup & Time ~ Reference ~	Math & More Trig ~ Functions ~	Name Manager M	election Remove	ecedents ependents Arrows Fr	Show Formulas Show Formulas Show Formula Show Formula Show Formula Show Formula Show Formula	Watch Window	Calculation Calculation Options ~ Calculation	late Now late Sheet					^
811 * : × ✓ Å														^
A	8	c	D	E		F	G	н	1	J	к	L	м	
1 Import data FIN	O1 (100m)													
2 Average per hour	Average total		Days	Hours	1	2		3	4	5	6	7	8	9
3 9,76285714285714	=AVERAGE(F3:AD8762)		=ROUNDUP(E3/24;0)	1	=A3	=A8763		=A17523	=A26283	=A35043	=A43803	=A52563	=A61323	=A70083
4 11,4242857142857			=ROUNDUP(E4/24;0)	2	=A4	=A8764		=A17524	=A26284	=A35044	=A43804	=A52564	=A61324	=A70084
5 12,28			=ROUNDUP(E5/24;0)	3	=A5	=A8765		=A17525	=A26285	=A35045	=A43805	=A52565	=A61325	=A70085
6 13,4571428571429			=ROUNDUP(E6/24;0)	4	=A6	=A8766		=A17526	=A26286	=A35046	=A43806	=A52566	=A61326	=A70086
7 13,7871428571429			=ROUNDUP(E7/24;0)	5	=A7	=A8767		=A17527	=A26287	=A35047	=A43807	=A52567	=A61327	=A70087
8 14,9685714285714	0 cells		=ROUNDUP(E8/24;0)	6	=A8	=A8768		=A17528	=A26288	=A35048	=A43808	=A52568	=A61328	=A70088
9 15,7028571428571	=COUNTIF(F3:AD8762;0)		=ROUNDUP(E9/24;0)	7	=A9	=A8769		=A17529	=A26289	=A35049	=A43809	=A52569	=A61329	=A70089
10 15,7285714285714			=ROUNDUP(E10/24;0	8	=A10	=A8770		=A17530	=A26290	=A35050	=A43810	=A52570	=A61330	=A70090
11 15,2085714285714			=ROUNDUP(E11/24;0	9	=A11	=A8771		=A17531	=A26291	=A35051	=A43811	=A52571	=A61331	=A70091
12 15,61		_	=ROUNDUP(E12/24;0	10	=A12	=A8772		=A17532	=A26292	=A35052	=A43812	=A52572	=A61332	=A70092
13 14,8228571428571			=ROUNDUP(E13/24;0	11	=A13	=A8773		=A17533	=A26293	=A35053	=A43813	=A52573	=A61333	=A70093
14 14,04			=ROUNDUP(E14/24;0	12	=A14	=A8774		=A17534	=A26294	=A35054	=A43814	=A52574	=A61334	=A70094
15 13,6			=ROUNDUP(E15/24;0	13	=A15	=A8775		=A17535	=A26295	=A35055	=A43815	=A52575	=A61335	=A70095
16 13,8514285714286			=ROUNDUP(E16/24;0	14	=A16	=A8776		=A17536	=A26296	=A35056	=A43816	=A52576	=A61336	=A70096
17 14,6385714285714			=ROUNDUP(E17/24;0	15	=A17	=A8777		=A17537	=A26297	=A35057	=A43817	=A52577	=A61337	=A70097
18 14,9557142857143			=ROUNDUP(E18/24;0	16	=A18	=A8778		=A17538	=A26298	=A35058	=A43818	=A52578	=A61338	=A70098
19 14,3585714285714			=ROUNDUP(E19/24;0	17	=A19	=A8779		=A17539	=A26299	=A35059	=A43819	=A52579	=A61339	=A70099
20 14,3342857142857			=ROUNDUP(E20/24;0	18	=A20	=A8780		=A17540	=A26300	=A35060	=A43820	=A52580	=A61340	=A70100
21 14,9057142857143			=ROUNDUP(E21/24;0	19	=A21	=A8781		=A17541	=A26301	=A35061	=A43821	=A52581	=A61341	=A70101
22 15,1571428571429			=ROUNDUP(E22/24;0	20	=A22	=A8782		=A17542	=A26302	=A35062	=A43822	=A52582	=A61342	=A70102
23 15,6742857142857			=ROUNDUP(E23/24;0	21	=A23	=A8783		=A17543	=A26303	=A35063	=A43823	=A52583	=A61343	=A70103
24 15,0371428571429			=ROUNDUP(E24/24;0	22	=A24	=A8784		=A17544	=A26304	=A35064	=A43824	=A52584	=A61344	=A70104
25 14,59			=ROUNDUP(E25/24;0	23	=A25	=A8785		=A17545	=A26305	=A35065	=A43825	=A52585	=A61345	=A70105
26 14,6071428571429			=ROUNDUP(E26/24;0	24	=A26	=A8786		=A17546	=A26306	=A35066	=A43826	=A52586	=A61346	=A70106
27 13,9242857142857			=ROUNDUP(E27/24;0	25	=A27	=A8787		=A17547	=A26307	=A35067	=A43827	=A.52587	=A61347	=A70107
28 13,4957142857143			=ROUNDUP(E28/24;0	26	=A28	=A8788		=A17548	=A26308	=A35068	=A43828	=A52588	=A61348	=A70108
29 13,5642857142857			=ROUNDUP(E29/24;0	27	=A29	=A8789		=A17549	=A26309	=A35069	=A43829	=A52589	=A61349	=A70109
30 14,1			=ROUNDUP(E30/24;0	28	=A30	=A8790		=A17550	=A26310	=A35070	=A43830	=A52590	=A61350	=A70110
31 14,0442857142857			=ROUNDUP(E31/24;0	29	=A31	=A8791		=A17551	=A26311	=A35071	=A43831	=A52591	=A61351	=A70111
32 12,8542857142857			=ROUNDUP(E32/24;0	30	=A32	=A8792		=A17552	=A26312	=A35072	=A43832	=A52592	=A61352	=A70112
33 13,0385714285714			=ROUNDUP(E33/24;0	31	=A33	=A8793		=A17553	=A26313	=A35073	=A43833	=A52593	=A61353	=A70113
34 11,2442857142857			=ROUNDUP(E34/24;0	32	=A34	=A8794		=A17554	=A26314	=A35074	=A43834	=A52594	=A61354	=A70114
35 9,45			=ROUNDUP(E35/24;0	33	=A35	=A8795		=A17555	=A26315	=A35075	=A43835	=A52595	=A61355	=A70115
36 8,89			=ROUNDUP(E36/24;0	34	=A36	=A8796		=A17556	=A26316	=A35076	=A43836	=A52596	=A61356	=A70116
• Random Failure rate	Lightning Failure rate	Icing Failure rate	Overload Failure rate	Wearout Failure n	ate Y	earsDaysHours V	(earout ene	ergy input converter	Wearout energ	y input Average w	ind speed per hour	+	: •	Þ
Ready Calculate												#	E	+ 100%

Hours in a year

AND CONTRACTOR OF CALL					Reliability convert	er system using failure b	ehaviors (replacer	nent) - Excel				Mark Haring		
File Home Insert Page La	yout Formulas D	ata Review		, Tell me what										
fx AutoSum Recently Financial L Used - F	ogical Text Date & L · Time - Re function Library	ookup & Math & ference - Trig - F	More unctions ~ Manager = 0 Defi	lefine Name ~ be in Formula ~ reate from Selection ned Names	So Trace Precedents of Trace Dependents Remove Arrows	Show Formulas Show Formulas Show Formula Evaluate Formula armula Auditing	Watch Cal Window Op	culation Calculate Now tions - Calculation						
* × × β														
A	8	c	D	E	F	G	. H	Formula Bar	J	ĸ	ι	м	N	1
loout			Davs	Hours	1	2	3	4	5	6	7	9	9	_
lour coruser	265#24		-POUNDUD/52/24/01	Hours	-00049053	-563+68649653	-063-6864865		-103400044003	-163468644653		-1 63+60647653	-4403-400044003	-
tour -	1/83		=ROUNDUR(E4/34-0)1		-CBCA*CEA	-FC2+CBC4+CE4	-002+0034 30		-10740864+014	=102+0804+0+4	-KC2+CBC4+CF4	-1 6346864*6*4	-4407408044044	
our I	4/ 5/2		-00000000000000000000000000000000000000		-00048000	-1 44 100 48 004	-0,270000-50							
			=ROUNDUP(E3/24;0)3		-0004 303	-F02T0004 0C0	-05275054 SE		-10270034 003	=J3273834 3E3	=K52+5854 565	*L32*3034 3C3	-140340004000	
			=ROONDOP(E0/24,0)4		-3034 300	=F02+0004 000	-032+3834 56	1152+5054 500	10210004 000	100210804 000	=K02+3034 300	=L32+3B34 3C0	1013243034 300	
			-ROUNDUP(E7/24,0)5	-	-3034 307	-132+3034 307	-032+3834 3E	-1132+3034 327	-132+3034 307	-332+3034 327	-K32+3034 307		-1413243034 327	
			=ROUNDUP(E6/24;0)6		-5054*550	=F52+5854 368	-052+5854 561		-102+383+ 368	-10240804368	=K32+3834 368	=L32+3D34 3EB	-1452+5834 568	
			=ROUNDUP(E3/24,0)7		-3034 303	-024000490010	-032+3834 3E		-132+3834 363	-102+0804 303	-K32+3834 3E3	-L3273034 303	-1413243834 353	
			-ROUNDUR/E11/24-09		-0004*0011	-F02+0804 0E10	-GC3+CBC4=CC	-ucatopca=cc11	-IC2+0804 0E10	-IC2+CBC4*CC11	-K02+3834 3E10	-L02+3034 3E10	-640340004*0011	1
			-ROUNDUR/E13/34/30		-0004 5011	-503+686486513	-002-0004 00.		-102+0004*0012	-1032080480013		-1 03+000470013	-84034000440011	-
			=PO(IND)(D)(512/24)(10		-0004 0012	-502-5054 5012	-002-0004 00.	12	-102-0004 0012	-1034000440013	-***************	-1 \$346064=6512	-8407+0004+0011	2
			=ROUNDUR/E14/24:012		-6864*6814	-F\$2+\$8\$4*\$E14	-G\$2+5054 5C		-IC2+CBC4+CE14	-162+6864*6814	=KG2+GBG4+GE14	-L02+0004 0C10	-M4C7+CBC4+CE1/	
			-ROUNDUR/E15/24:013		-5854*5E15	-E\$7+\$R\$4*\$E15	-G\$2+\$B\$4*\$E	-HC2+CRC4*CE15	-102+0854+0615	-I\$2+\$8\$4*\$E15	-K\$2+5854*5E15	-1 \$2+\$8\$4*\$E15	-M\$7+\$854*\$515	5
			=ROUNDUR/E16/24/014		-0004 0010	-EC3+CBC4*CE16	-GC2+CBC4*CE	-UC24CBC4*CC16	-102+0004 0010	-162+6864+6516	-¥\$2+\$8\$4*\$516	-1 \$346864*6616	-840740004*0016	6
			-POUNDUR/E17/24-015		-5954*5517	-503+0804*0517	-003+0004*00	7 -UC24CBC4*CE17	-10740004*0017	-10240004*0017	-KC2+CBC4*CE17	-1 \$24\$0\$4*\$517	-640340004*0017	7
			=ROUNDUR/E18/24:016		=\$8\$4*\$E18	=F\$2+\$8\$4*\$F18	=G\$2+\$B\$4*\$E	IR =H\$2+\$B\$4*\$E18	=1\$2+\$8\$4*\$518	=1\$2+\$8\$4*\$E18	=K\$2+\$8\$4*\$518	=1 \$2+\$8\$4*\$518	=M\$2+\$B\$4*\$E15	8
			-POUNDUD/519/24-017		-0004*0010	-502+0004 0010	-CC3+CDC4=CC	-UC34CDC4*CC10	-107+0004*0010	-10340004*0010	-¥\$2+\$8\$4*\$519	-1 \$346064#6610	-640340004*0010	6
			=ROUNDUR/E20/24-018		=\$8\$4*\$E20	=E\$2+\$8\$4*\$E20	=G\$2+\$B\$4*\$E	N0 #H\$2+\$8\$4*\$E20	=1\$2+\$8\$4*\$E20	=1\$2+\$8\$4*\$520	=K\$2+\$8\$4*\$520	=1 \$2+\$B\$4*\$E20	=M\$2+\$8\$4*\$E30	6
			=ROUNDUR(F21/24:019	-	=\$R\$4*\$E21	=F\$2+\$B\$4*\$F21	=G\$2+\$B\$4*\$F	H\$2+\$8\$4*\$F21	=1\$2+\$8\$4*\$F21	=1\$2+\$8\$4*\$F21	=K\$2+\$8\$4*\$E21	=L\$2+\$B\$4*\$E21	+M\$2+\$8\$4*\$F21	1
			=ROUNDUP(F22/24:020		=\$8\$4*\$F22	=F\$2+\$B\$4*\$E22	=G\$2+\$B\$4*\$E	27 =H\$2+\$8\$4*\$F22	=1\$2+\$8\$4*\$F22	=1\$2+\$8\$4*\$F22	=K\$2+\$8\$4*\$F22	=1 \$2+\$B\$4*\$E22	=M\$2+\$8\$4*\$F27	2
			-ROUNDUP/E23/24-021		-\$8\$4*\$522	-E\$2+\$B\$4*\$E23	-G\$2+\$8\$4*\$E		-152+CR54+SE22	-1\$2+\$8\$4*\$523	-K\$2+\$8\$4*\$E23	-1 \$2+\$B\$4*\$E22	-MC2+CRC4+CE21	2
			=ROUNDUP(F24/24)(22		=\$8\$4*\$524	=F\$2+\$8\$4*\$F24	=G\$2+\$B\$4*\$E	=H\$2+\$B\$4*\$E24	=I\$2+\$B\$4*\$F24	=1\$2+\$8\$4*\$F24	=K\$2+\$B\$4*\$E24	=1 \$2+\$8\$4*\$524	=M\$2+\$8\$4*\$F26	á.
			=ROUNDUP(F25/24:023		=\$8\$4*\$525	=F\$2+\$8\$4*\$F25	=G\$2+\$B\$4*\$E	5 =H\$2+\$8\$4*\$E25	=I\$2+\$8\$4*\$E25	=1\$2+\$8\$4*\$F25	=K\$2+\$8\$4*\$525	=1 \$2+\$B\$4*\$E25	=M\$2+\$R\$4*\$E25	5
			=ROUNDUP(F26/24:024	-	=\$8\$4*\$£26	=F\$2+\$B\$4*\$E26	=G\$2+\$B\$4*\$E	16 =H\$2+\$B\$4*\$E26	=I\$2+\$8\$4*\$E26	=1\$2+\$8\$4*\$F26	=K\$2+\$8\$4*\$E26	=L\$2+\$8\$4*\$E26	=M\$2+SB\$4*SE26	6
			=ROUNDUP(F27/24-025		=\$8\$4*\$527	=F\$2+\$8\$4*\$F27	=G\$2+\$8\$4*\$F	7 =H\$2+\$R\$4*\$F27	=I\$2+\$R\$4*\$F27	=1\$2+\$8\$4*\$527	=K\$2+\$8\$4=\$F27	=1 \$2+\$8\$4*\$527	=M\$2+\$R\$4*\$F27	7
			=ROUNDUP(F28/24:026		=\$R\$4*\$F28	=F\$2+\$8\$4*\$F28	=G\$2+\$B\$4*\$E	R =H\$2+\$B\$4*\$F28	#I\$2+\$R\$4*\$F28	=1\$2+\$8\$4*\$F28	=K\$2+\$8\$4*\$F28	#L\$2+\$B\$4*\$E28	=M\$2+\$R\$4*\$F28	8
			=ROUNDUP(F29/24:027		=\$8\$4*\$F29	=F\$2+\$B\$4*\$E29	=6\$2+\$8\$4*\$E	9 =H\$2+\$B\$4*\$E29	=1\$2+\$8\$4*\$E29	=1\$2+\$8\$4*\$E29	=K\$2+58\$4*5E29	=1\$2+\$8\$4*\$E29	=M\$2+\$8\$4*\$E25	9
			=ROUNDUP(E30/24:028		=\$B\$4*\$E30	=F\$2+\$B\$4*\$E30	=G\$2+\$B\$4*\$E	0 =H\$2+\$B\$4*\$E30	=I\$2+\$B\$4*\$E30	=J\$2+\$B\$4*\$E30	=K\$2+\$8\$4*\$E30	=L\$2+\$B\$4*\$E30	=M\$2+\$B\$4*\$E30	0
			=ROUNDUP(E31/24:029		=58\$4*\$E31	=F\$2+\$8\$4*\$E31	=GS2+SBS4*SE	1 =H\$2+\$B\$4*\$E31	=I\$2+\$B\$4*\$E31	=J\$2+\$B\$4*\$E31	=K\$2+\$8\$4*\$E31	=L\$2+\$B\$4*\$E31	=MS2+SBS4*SE31	ĩ
			=ROUNDUP(E32/24:030		=58\$4*\$E32	=F\$2+\$B\$4*\$E32	=G\$2+\$B\$4*\$E	2 =H\$2+\$B\$4*\$E32	=I\$2+\$B\$4*\$E32	=J\$2+\$B\$4*\$E32	=K\$2+\$8\$4*\$E32	=L\$2+\$B\$4*\$E32	=M\$2+\$B\$4*\$E32	2
			=ROUNDUP(E33/24:031		=\$8\$4*\$E33	=F\$2+\$8\$4*\$E33	=G\$2+\$B\$4*\$E	13 =H\$2+\$B\$4*\$E33	=I\$2+\$B\$4*\$E33	=J\$2+\$8\$4*\$E33	=K\$2+\$8\$4*\$E33	=L\$2+\$B\$4*\$E33	=M\$2+\$8\$4*\$E3?	3
			=ROUNDUP(E34/24:032		=\$8\$4*\$E34	=F\$2+\$8\$4*\$E34	=G\$2+\$8\$4*\$E	4 =H\$2+\$B\$4*\$E34	=I\$2+\$B\$4*\$E34	=J\$2+\$8\$4*\$E34	=K\$2+\$8\$4*\$E34	=LS2+SBS4*SE34	=MS2+SBS4*SE34	4
			=ROUNDUP(E35/24:033		=\$8\$4*\$E35	=F\$2+\$8\$4*\$E35	=G\$2+\$8\$4*\$E	15 =H\$2+\$8\$4*\$F35	=I\$2+\$B\$4*\$E35	=J\$2+\$8\$4*\$E35	=K\$2+\$8\$4*\$E35	=L\$2+\$8\$4*\$E35	=M\$2+\$8\$4*\$F35	5

Wearout energy input

Autolieve (0 🗄 🐆 d - i			Reliabili	ity converter system using failure be	haviors (replacement)	+ Excel					Mark Haring	m – c	a x
	Insert Page Layout Formulas	Data Review View Ad		what you want to d	do									E Share
$\int_{\substack{\text{Insert}\\\text{Function}}} \sum_{\substack{\text{AutoSur}\\ \forall}}$	m Recently Financial Logical Text Date Used V Viet Viet Viet Viet Viet Viet Viet Vi	a Lookup & Math & More ⇒ Reference → Trig → Functions →	Name Manager Define Name Manager Defined Names	Bo Trace Pr off Trace Do tion Remove	recedents Show Formulas lependents 'S Error Checking - e Arrows - (a) Evaluate Formula Formula Auditing	Watch Window Calculati Options	Calculate Now							^
F3 *	I × √ fe =IFS('Averaj 'Wearout en energy inpu	ge wind speed per hour'IF3<='We hergy input'I\$8\$5;'Average wind s it'I\$A\$8;'Wearout energy input'I\$I	arout energy input'l\$A\$3;'W peed per hour'lF3<='Wearo 8\$8;'Average wind speed pe	'earout energy ir ut energy input'l ir hour'lF3<='We	nput'1\$B\$3;'Average wind spee 1\$A\$6;'Wearout energy input'1 arout energy input'1\$A\$9;'Wea	d per hour'IF3<='W 8\$6;'Average wind rout energy input'I	earout energy input'!\$ speed per hour'IF3<=' \$8\$9;'Average wind sp	A\$4;'Wearout er Wearout energy weed per hour'IF3	hergy input'158\$4;' input'15A57;'Wear 3<='Wearout energ	Average wind spe- out energy input'l ty input'I\$A\$10;'W	ed per hour'l ISB\$7;'Averaj earout energ	IF3<='Wearout o ge wind speed j gy input'I\$8\$10;	energy input'I\$A\$5 per hour'IF3<='We 'Average wind spe	arout eed -
	8	c	D	E	FG	н	- E	1	j	к	ι	M		N A
1	Power curve													
2 Wind	speed Power kW/h	Average Power	Days	Hours	1 2	3	4	5	6	7		8	9	
3 3	0	=AVERAGE(F3:AD8762)	=ROUNDUP(E3/24;0)1		=IFS('Average ='Average wind	spee='Average wi	nd spec='Average win	d spec='Average	wind spee='Avera	ge wind spee='Av	erage wind s	ipee='Average v	vind spee ='Averag	ge wind
4 3,5	70		=ROUNDUP(E4/24;0) 2		='Average wit ='Average win-	spec='Average wi	nd spec ='Average win	d spe∈='Average	wind spee='Avera	ge wind spee ='Av	erage wind s	pec ='Average v	vind spec ='Averag	ge wind
5 4	141	Capacity factor	=ROUNDUP(E5/24;0) 3		='Average wit ='Average win	spee='Average wi	nd spee='Average win	d spe∈='Average	wind spee='Avera	ge wind spee ='Av	erage wind s	pee='Average s	vind spee ='Averag	ge wind
6 4,5	240	=C3/5000	=ROUNDUP(E6/24;0)4		='Average wii ='Average win-	spec='Average wi	nd spec='Average win	d spec ='Average	wind spec='Avera	ge wind spec ='Av	erage wind s	pec ='Average v	vind spec ='Averag	ge wind
7 5	343		=ROUNDUP(E7/24;0)5		='Average wii ='Average win	i spe∈='Average wi	nd spee='Average win	d spe∈='Average	wind spee ='Avera	ge wind spee='Av	erage wind s	ipee='Average u	vind spee ='Averag	ge wind
8 5,5	490	Total power kW	=ROUNDUP(E8/24;0)6		='Average wii='Average win	i spe∈='Average wi	nd spec ='Average win	d spe∈='Average	wind spec='Avera	ge wind spec='Av	erage wind s	pec='Average v	vind spec ='Averag	ge wind
9 6	636	=SUM(F3:AD8762)	=ROUNDUP(E9/24;0) 7		='Average wii ='Average win	spee='Average wi	nd spec='Average win	d spee ='Average	wind spec='Avera	ge wind spee='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
10 6,5	850		=ROUNDUP(E10/24;08		='Average wii ='Average win	spec='Average wi	nd spec ='Average win	d spe∈='Average	wind spec='Avera	ge wind spee ='Av	erage wind s	pee ='Average v	vind spec ='Averag	ge wind
11 7	1067	1	=ROUNDUP(E11/24;09		='Average wii ='Average win	i spec='Average wi	nd spec='Average win	d spec='Average	wind spec='Avera	ge wind spec='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
12 7,5	1340		=ROUNDUP(E12/24;010		='Average wii ='Average win	l spe∈='Average wi	nd spee ='Average win	d spe€ ='Average	wind spec='Avera	ge wind spee ='Av	erage wind s	pee ='Average v	vind spee ='Averag	ge wind
13 8	1615		=ROUNDUP(E13/24;011		='Average wii ='Average win	l spe∈='Average wi	nd spee='Average win	d spe∈='Average	wind spee ='Avera	ge wind spee='Av	erage wind s	pee='Average v	vind spee ='Averag	ge wind
14 8,5	1950		=ROUNDUP(E14/24;012		='Average wii ='Average win	spec='Average wi	nd spec ='Average win	d spec='Average	wind spec='Avera	ge wind spec='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
15 9	2289		=ROUNDUP(E15/24;013		='Average wii ='Average win	spe∈='Average wi	nd spee ='Average win	d spe∈='Average	wind spec='Avera	ge wind spee='Av	erage wind s	per='Average v	vind spec ='Averag	ge wind
16 9,5	2720		=ROUNDUP(E16/24;014		='Average wii ='Average win-	spec='Average wi	nd spec='Average win	d spec='Average	wind spec='Avera	ge wind spec='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
17 10	3166		=ROUNDUP(E17/24;015		='Average wit ='Average wind	spec='Average wi	nd spec ='Average win	d spec='Average	wind spec ='Avera	ge wind spee='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
18 10,5	3575		=ROUNDUP(E18/24;016		='Average wii ='Average wine	spe∈='Average wi	nd spee='Average win	d spe∈='Average	wind spec='Avera	ge wind spee ='Av	erage wind s	pee «'Average »	vind spec ='Averag	ge wind
19 11	3984		=ROUNDUP(E19/24;017		='Average wii ='Average win	spec='Average wi	nd spec='Average win	d spec='Average	wind spec='Avera	ge wind spee='Av	erage wind s	spee='Average v	vind spec ='Averag	ge wind
20 11,5	4366		=ROUNDUP(E20/24;018		='Average wii ='Average wine	spec='Average wi	nd spe∈='Average win	d spe∈='Average	wind spec='Avera	ge wind spee ='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
21 12	4748		=ROUNDUP(E21/24;019		='Average wir='Average wine	spee='Average wi	nd spee='Average win	d spe∈='Average	wind spee='Avera	ge wind spee ='Av	erage wind s	pee='Average v	vind spec ='Averag	ge wind
22 12,5	4930		=ROUNDUP(E22/24;020		='Average wit='Average wind	spec='Average wi	nd spec ='Average win	d spec='Average	wind spec='Avera	ge wind spec='Av	erage wind s	per ='Average v	vind spec ='Averag	ge wind
23 13	4978	_	=ROUNDUP(E23/24;021		='Average wii='Average win	spec='Average wi	nd spec='Average win	d spe€='Average	wind spec='Avera	ge wind spec='Av	erage wind s	per='Average v	vind spee ='Averag	ge wind
24 13,5	14990		=KOUNDUP(E24/24;022		= Average wii = Average win	spec='Average wi	nd spee ='Average win	d spec='Average	wind spec ='Avera	ge wind spec='Av	erage wind s	pec='Average v	vind spec='Averag	ge wind
25 14	14999		=KOUNDUP(E25/24;023		= Average will='Average wink	spee = Average wi	no spec ='Average win	o spec='Average	wind spec='Avera	ge wind spee='Av	erage wind s	pee='Average s	vind spee = Averag	ge wind
26 14,5	15000	-	=ROUNDUP(E26/24;024		='Average wit ='Average with	spec='Average wi	nd spee = Average win	d spec='Average	wind spec = Avera	ge wind spee = Av	erage wind s	pee = Average v	vind spec = Averag	ge wind
27 15	5000		=ROUNDUP(E27/24)025		= Average wii = Average win	spee=:Average wi	nd spec = Average win	d spec = Average	wind spee = Avera	ge wind spec = AV	erage wind s	pee = Average 1	vind spec = Averag	Je wind
28 15,5	5000		ROUNDUP(E28/24)(26		= Average wit = Average with	spec=Average wi	nd spec = Average win	d spec = Average	wind spec = Avera	ge wind spec = Av	erage wind s	pee = Average v	vind spec = Averag	ge wind
20 14 1	5000		=RUUNUUP(E29/24;02/		= Average wil='Average win	spee = Average wi	no spec = Average win	o spec = Average	wind spec='Avera	ge wind spec='Av	erage wind s	per = Average v	vind spec='Averag	ge wind
30 10,5	5000		=ROUNDUP(E30/24)(28		- Average will - Average with	spec = Average wi	nd spec = Average win	d spec = Average	wind spec = Avera	ge wind spec = Av	erage wind s	pee = Average v	vind spec = Averag	ge wind
22 17 6	5000		-ROUNDUP(E31/24;029		- Average wil = Average win	spee - Average wi	nd spec - Average win	d spec - Average	wind spec = Avera	ge wind spec = AV	erage wind s	per - Average v	vinu spec = Averag	ge word
33 18	5000		=ROUNDUP(E33/24-021		='Average wit ='Average with	spec - Average wi	nd snee ='Average win	d snot ='Avorage	wind spec = Avera	ee wind snee -'Au	prace wind s	nes s'Average s	wind spec - Averag	ce wind
34 18.5	5000		=ROUNDUP(F34/24-032		s'Average wils'Average with	spee = Average wi	nd spec = Average with	d spec = Average	wind spec s'Avera	ge wind spec = Av	erage wind s	per s'Average	vind speen Averag	ge wind
35 19	5000		=ROUNDUP(F35/24-033		='Average wit ='Average win	spec ='Average wi	nd spec ='Average win	d spec = 'Average	wind spec = Avera	ge wind spec -'Av	erage wind o	ines ='Average	vind spec = Averag	ge wind
36 19.5	5000		=ROUNDUP(E36/24+034		='Average wit ='Average win	spec ='Average wi	nd spec ='Average win	d spec ='Average	wind spec='Avera	ge wind spec='Av	erage wind s	ipec ='Average	vind spec ='Average	ge wind -
	Random Failure rate Lightming Fa	alure rate Iring Failure rate	Overload Failure rate	Nearout Failure r	ate YearsDaysHours W	earout energy input	converter Wearow	energy input	Average wind on	eed per hour	(+)	Be -	4	
Ready Calculate		and the second second second				an an an ang y mpar		and by the set	and and appendix					+ 100%

Wearout energy input converter

AutoSi	err (🖲 💷)	8 5	• c ¹ - •			_	_			_	Reliability	converter sy	istem using failu	e behaviors (r	eplacement)	- Excel	_	_	_				Mark Ha	ning [11]	- 0	a x
			Page Layout	Formulas		Review	View Ad	dd-ins		Tell me what																Le? Share
$\int_{\substack{\text{Insert}\\\text{Function}}}$	AutoSum ×	Recently F Used ~	Financial Logical	Text Date 8 - Time - on Library	Lookup B Reference	Hath & Trig + F	More unctions ~	() Name Manager	Define N Create for Defined Nar	ame ~ rmula ~ om Selection nes	So Trace Prec off Trace Dep K Remove A	cedents 📱 vendents 🌱 krows - Q Farm	2 Show Formula Error Checking Evaluate Form ula Auditing	s Watch ula Windo	Calculation v Options	Calculat on Calculat Calculation	te Now te Sheet									~
63		1.		-01184/020076	2-2 Managard		out (G2)																			
03				-ouni provin	, weatou	energy in	purioaj																			
	D		E		F.	1	3		н	1		J		к		L		M		N			0		P	-
2	Davs	-	Hours	1	1	2		3		4	5		6		7		8		9			10		11		
2 1	Days		nouis	A SUBALINAL	arout one	-CUBA/CCC	0763-5Mon	-SUBAC	cépten/Maio		STEDUNION PELL	14(0)00763	1211001-51124/5	100763-34401	-CUBRICUC	0763/34/0260	0	101000000000000000000000000000000000000	Trout or -FUR	1101100752	Weatout	-C) (8.4/ Chic)	1762/3Monrow	* 0 -51184/200	9763-34/01	A A
4 1		12		-SUM WE	aroutene	-SUM SPS	9762/1M/03	-SUM(S	CC0762/14/6	-sumption	8762, Wea-SU	141 (100762	Wear-Sund	100762/3Mor		0762/Wearo	ut en clit	(\$1,\$8762, We	sout or still	ALCAACO767-	Wearout	-SUM(\$N0)	1762: Wearou	t e =SUM(505	0762-3Mas	arout e -
2 1		2		-SUM WE	aroutene	=SUM(SPS	8762; wea	=50M(9	038702; Wei	=SUM(\$H\$I	8762; wea = SU	INI \$1\$8702	; wear = sum(s	150762; Wei	I=SUM(SKS	8762; Wearo	ut er=SUN	1(5L58762; We	rout er =SUR	ALCA ACO 762.	Wearoute	=SUNI(SNS	5762; wearou	t e =SUM(SUS	8762; wez	arout e =
6 1		3		-SUM WE	aroutene	-5010(323	0702, wea	-SUM(3	100767. We	-SUM(SHO	8762, Wea - 50	INI (\$1\$6702 BALCICOTES	Wear-SUM(3	100762-3Mai		0762, Wearo	ut er-SUN	13L36702, VVE	rout er -Suit	1(\$N1\$6702,	Wearoute	-50101(5145	1762, Wearou	t e =SUM(505	0702, Wea	aroute -
7 1		5		-SUM WE	aroutene	-CLINATCEC	9762-1Moo	-50116/6	CC0767-1Ma	-cutikaréuéi	8762; Wea -50	BALCICO762	Wear-Sum(2	102762-3Mor		8762-'Moaro	ut or -SUB	101.00702, We	rout or -SUB	#(\$M\$6702,	Wearoute	-SUM (SNO	762; Wearou	t o -SUM(505	8762-3Mor	arouto
1 1		5		-SUM WE	aroutene	=5UM(5F5	0702; Wea	-SUM(S	000702; We	-FURNISHO	5762; Wea = 50	101 0100702	; wear sowig	100702; Wea	E-SUM(SKS	every wearo	ut et=SUN	101.50702; We	arout er =SUP	n(5M 50702;	Wearoute	-FURACENE	5702; wearou	it e =50101(503	8762; Wee	arout e -
0 1		0		SUM WE	aroutene	=SUM(SFS	oroz, wea	-SUM S	000702, We	- CUBRICHIC	broz, wea su	1111 2120702	, wear =som(s	100702; Wee	E SUM SKS	oroz, wearo	ut er son	(3L36702, We	router -sor	n(\$10,00702)	wearoute	CURACENCE	s762, wearou	it e =50141(505	0702; wea	arout e -
9 1		1		=SUM WE	aroutene	=SUM(SFS	8702; Wes	=50M(5	cónzez-hale	=======================================	8762; Wea = 50	101(5158702	; wear = sum(s	156702; wea	=SUM(SKS	o/oz; wearo	IUL EI =SUN	(SL58762; We	srout er =SUI	n(5M56702;	Wearoute	-SUM(SNS	5762; wearou	it e =SUM(SUS	6702; Wea	aroute =
11 1		0		-SUM WE	aroutene	=SUM(\$P\$	0763-5Mon	SUMP	CC0762/14/0	-CUMPTON	8762; Wea = 50	INI \$150702	Wear = SUM(s	159752/3Mor	- SUM(SKS	8762; Wearo	ut er=SUN	1(5L58762; We	rout or -SU	1(\$M\$676767.	Wearoute	-SUM(\$N\$	5762; wearou	t e =SUM(505	8762; wea	arout e =
12 4		10		-SUM WE	aroutene	-30/0(323	0762-04/ee	-50/01/2	contra-line	======================================	5702, wea -50	101(3136702	, wear -soni(s	130702, wea		oroz, wearo	ut er-Sun	1(3c38702, we	nouter -sor	1(3W136702,	wearoute	-5000(505	s702, wearou	it e -sulvi(303	0702, WE2	aroute -
12 1		10		=SUM We	aroutene	=SUM(\$P\$	8762; wea	=SUM(S	038702; Wes	=SUM(SHS	8762; wea = 50	1011 \$1\$8702	; wear = sum(s	150762-3Mar	=SUM(SKS	0762; Wearo	ut er=SUN	1(5L58762; We	rout er =SUR	N(\$N\$8702;	Wearoute	-SUM(SNS	s762; wearou	it e =SUM(SUS	8762; wez	arout e
15 1		12		-SUM We	aroutene	-SUMISES	0762, Wea	-50/10/0	100702, We	-SUM(SHO	8762, Wea-50	111 5136702	Wear-SUM()	169763/3Mer		0702, Wearo	iut er-SUN	13L36702, We	arout er -Suit	ALCA4CO767.	Wearoute	-50101(5105	1762, wearou	10 = SUM(505	8763-3Mar	aroute
14 1		12		=SUM We	aroutene	SUMISES	8762; wea	SUMS	058702; Wei	-CUM(SHSI	8762; Wea = SU	IMI \$1\$8702	; wear =SUM(s	158702; Wei	I SUM(SKS	8702; Wearo	ut er=SUN	(SL\$8762; We	arout er "SUI	n(5M58702;	wearoute	SUMISINS	s762; wearou	it e =SUM(SUS	8702; Wei	aroute
10 1		10		-FURACINA	aroutene	=50/W[5F5	0702; Wea	-FUNCO	000702; We	-FURACEUE	5762; Wea = 50	101(5150702	, wear = sum(c	158762; Wea	======================================	8763-34/earo	ut et -50%	(3L36762; VVe	arout er =50	n(5M 56702;	Wearoute	-CUNICING	5762; wearou	te =50101(505	8702; wee	arout e =
10 1		1.0		-SUM WE	aroutene	=SUM(SPS	oroz, wea	SUNC	000702, We	s subs(cuto	b702, wea=50	1111 2120702	, wear =sowijs	100702, Wet	- CUNICKS	o702, wearo	ut el son	(3L38762, We	router sor	n(\$M\$6702;	wearoute	CURACENCE	5702; wearou	10 0 - 50101(505	6702; wea	aroute =
10 1		15		=SUM We	aroutene	=SUM(SFS	areadarea	=SUM(5	058702; Wei	= SUM SHO	5762; Wea = 50	101(\$1\$8702	; wear =sum(s	158762; Wea	I=SUM(SKS	o/oz; wearo	IUL EL SUN	1(5L58762; We	srout er =SUI	1(SMS8/02;	Wearoute	-5UNI(5N5	5762; wearou	it e =SUM(SUS	8702; wea	arout e =
10 1		10		=SUMI WE	aroutene	=SUM(SFS	aroz, wea	SUNTO	cóozca hur	=SUM(\$H\$I	b/b2; wea=su	1111 5158702	; wear =sum(s	138702; Wei	=SUM(SKS	a762; wearo	ut er=sun	(3L38762; We	HOUL EF =SUP	n(\$M\$8702;	wearoute	SUNICIPAL	s762; wearou	it e =suivi(sus	6702; wea	arout e =
19 1		17		=SUM We	aroutene	=SUM(SFS	8762; Wea	=50M(9	658702; Wei	=SUM[SHS	8762; wea=SU	INI \$1\$8702	; wear = SUM(\$	158702; Wea	I=SUM(SKS	8762; Wearo	lut el =SUN	(\$L\$8762; We	srout er =SUI	N(\$M\$8702;	wearoute	SUNISINS	s762; wearou	it e =SUM(SUS	8762; wea	arout e =
20 1		10		=SUM WE	aroutene	=5UM[\$P\$	8762; vvea	=SUM(S	038702; Wei	=SUM(SHSI	8702; wea=50	INIT \$158702	; wear =sum(s	158702; wei	==========	a762; wearo	lut et=50%	(3L38702; We	rout er = SUR	n(\$M\$8702;	wearoute	=SOM(SNS	s762; wearou	it e =50101(505	8702; wez	arout e =
20.1		15		-SUM We	aroutene	-50/01/31/3	oroz, wea	-50///5	0.30702, We	-SUM(SHO	broz, wea-su	INIT 2120102	, wear -som(a	130702, wee		over a vearo	lut ei -son	1(3L36702, We	nouter-sur	n(5m 56702,	wearoute	-50101(5145	s702, wearou	it e -sulvi(sus	6702, wee	aroute -
22 1		20		-SUM We	arout ene	=SUM(SFS	8762; wea	SUMS	058702; Wei	I SUM(SHSI	8762; wea =50	IMI(\$1\$8702	; wear =sum(s	158702; Wei	I SUM(SKS	8762; Wearo	lut el SUN	(5L58762; We	arout er =SUM	n(\$M\$8702;	wearoute	SUMISINS	5762; wearou	it e =SUM(SUS	8702; wea	arout e =
25 1		21		=SUM('WE	aroutene	=SUM(SFS	0702, Wea	-SUM(S	upo/o2; Wei	s=SUNI(SHSI	oroz; wea su	nnt 3138762	; wear =SUM(S	100/02; Wea	=SUM(SKS	o/oz; Wearo	ut et ISUN	1 2L20 /02; We	arout ef =SUI	n(\$20,98762);	wearout e	-sum(SNS	5/02; wearou	ne =sUM(SUS	0/02; Wea	arout e =
25 1		22		=SUM['We	arout ene	-SUM(SPS	0702, Wea	-SUM(S	céezez-halo	-cunit cuci	erea viela esu	INIT \$1\$8762	Wear SUM(S	169762/Wei	======================================	erez, Wearo	ut er ISUN	1 pt pt /62; We	mout or -SUB	n(pm p8/02)	wearout e	-CLIBA/CNIC	767/Wearou	t e =SUM(SUS	9763-Wei	arout e =
26 1		23		-SUM WE	aroutene	-3010(000	0702, Wes	-SUM(S	cée762:'Wei	====================	0702, wed =SU	INALCICO762	Wear SUM(S	100702; Web		9762 Wearo	ut er=SUN	10L00702; We	arout or still	n(20128/02; si ćasće 762/	Wearout a		1762: Wearou	t a _\$UM(\$U\$	9762-Wea	arout e -
27 2		24		-SUM('Ma	aroutene	-51184/626	8762-"Maa	-SUM(S	G\$9767-'We		9762-1Mag-SU	IM/ CICO762	"Mear-SUM(S	1\$9762-Wei		9762-Wearo	ut as -SUN	1 ¢1 ¢9762-11/0	rout er -SUB	# CMC2767-	Wearoute	-CI IM(\$143	762-"Mearou	t a -SUM/\$03	8762-3Mas	arout e -
28 2		26		=SUM(144	aroutene	=SLINALCER	8762-"M/~~	SUMP	G\$8762-W	-SUM Cući	8762-'Wea	MICICE763	"Wear=SUM(3	158762-344	==SUM/CVC	8762-'Wearo	ut an Si it	1 CI \$8762-14/-	arout or south	MCMC8767-	Wearout	-SUM(\$NS	1762-'Mearco	t e =\$11M(\$03	8762-34/0	arout e a
20 2		20		-SUM W	aroutene	-51184/656	9762-14(on	-SUM()	CC0762-14/m	-SUBACCUC	8762-'Miga -SU	BALCICO762	"Minor -ClinA/C	1\$9762-34/01		9762-'Wearo	ut er-Sith	10100762-14/0	mout or -SUB	#1 CAACO762-	Mearoute	-CLIBALCHIC	762; Wearou	t e -SUM(\$05	8762-3Mos	arout e -
20 3		120		-FLIBACIAN	aroutene	-SUBALCED	0762-34/es	-CLIBACO	céonen-haie	-custicué	8762, Wea-50	184(0100702	Stiest -Sunds	100702, 1101	-FUANOVO	8763-34/earo	ut er-SUM	10.00702, 140	sout er -50	4/04.400702,	Wearoute	-CLIBATÓRIÓ	1762, Wearou	t e -SUM(505	9763-3Mer	aroute -
21 2		20		SUMACHAI	arout one	-SUBALCEC	9762-14(oz	-CLIBALC	COTET-NAIS		2753: 1Alog =50	BA(C1C0767	Thight SUM	102762-340		9762 Wearo	aut on slin	10100702, 440	rout or -SUI	#/ CAACO767-1	Wearout	-CURA/CNIC	762: Moarou	to -5UNA(\$05	2762-3Mon	arout o -
32 2		25		-SUM WE	aroutene	-501106/010	8762-1A/	-50M(5	céanez-lui-	-SUM SUM	1762: Wed = 50	10.01 Cice 74 3	Mear -SUM(S	100702; Wee		8762-Wearo	ut an stat	10.00702; We	arout or still	ai da.edu.7/02;	Wearout -	-SUNIQUE	1762: Wearou	t a sum/cod	8762-Wee	arout e -
22 2		30		-SUM ING	arout ene	-SUMISES	9763-1M(00	-SUM(S	cégara halo	-JumpHSI	P767/Mico =SU	10.4/ CICO752	wear =suM(s	169763-5Mor	-SUMISKS	erez Wearo	wit or -SUN	101.00702; We	mout or -SUB	*(carconen)	Wearoute	-SUMI SNS	2767/3Moorou	t o -FUM(\$0\$	9763-3Wor	arout e -
34 2		31		-SUM WE	arout ene	-30W(\$P\$	8762-3A/ea		cée767/14/6	======================================	0702, wed =50	INN \$155702	Wear SUM(2	100762-3MA		8762-Wearo	ut au SUN	1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	arout or -SUB	nt 2m 28/02; #(carconco.)	Wearoute	-30ml(\$N\$	1762: Wearou	t e =5UM(\$U\$	8762-Wea	arout e -
25 2		32		-SUM('Ma	arout ene	-51184/000	9762-"Mea	-SUM(\$	C\$9767-'M/e	-SUM CLICI	8762-'Mias-SU	ILAI CI CE 762	-"Maar-CLIM/C	1\$9762-Wet		8762-Wearo	ut er=SUN	1 ¢1 ¢9762-'M/a	rout or -SUB	#(\$M\$9762;	Mearoute	-CLIMI (\$143	762-Mearou	t a -SUM/SOS	8762-Wes	arout e -
26 2		34		-SUM('We	arout ene	=SUM(\$P\$	8762-"M/ea	-SUM(S	G\$8762-'W/e	-SUM CHC 242 MILE	8762-'Wea =SU	IM/ \$1\$8762	"Mear =SUM(\$	1\$8762-Wei		8762-Wearo	at er=SUN	1 \$1 \$8762-'We	rout er still	ALCMC8762	Wearoute		1762-'Mearou	t a =\$UM(\$0\$	8762-3Mer	arout e a
100 A		34			ar out ente	-soundara	or one, vytea	-sond a	0,00706, 998				,			cros, weard	-30h	100,007,02, 100					riva, ritearius			
	F	Random Fa	ailure rate	Lightning Faile	ire rate	Icing Fail	ure rate	Overlo	ad Failure ra	te Wear	rout Failure rate	e Vears	sDaysHours	Wearout er	ergy input	converter	Wearout	energy input	Average	vind speed p	per hour	۲		1		Þ
Ready C	Calculate																						III III	巴	-	+ 100%

Wearout failure rate

AutoSave 💿 🖬 👆 🗸 🖒 -	- F			Reliability converter syst	em using failure behaviors (rej	olacement) - Excel						Mark Harin	, 🖽	- a ×
	Layout Formulas Data Review Vie													년 Share
$ \begin{array}{c c} f_{\mathcal{X}} & \sum_{\substack{\text{Insert} \\ \text{Function}}} & \underbrace{\sum_{\substack{\text{AutoSum} \\ \vee \\ \text{Used}}} & \underbrace{\text{Insert} \\ \vdots \\ \end{array} \right) $	Logical Test Date & Lockup & Math & Time * Reference * Trig * Fun Function Library	More ctions - Manager Creat Defined	n Formula ~ 850 n Formula ~ 455 te from Selection 855 Names	Trace Precedents Trace Dependents Remove Arrows ~ Formula	Show Formulas Error Checking – Evaluate Formula a Auditing	Calculation Calculation Options ~ Calculation	late Now late Sheet							^
C9 * I × √ J	fx													^
A	8	С	D	E	F	G	н	1		1	к		L.	N -
2 GW converter 10	000000		Days	Hours	1	2	3	4	5		6	7		8
3			=ROUNDUP(E3/24;0	1	=0,015504753*((('Wearou	t =0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
4 Year En	nergy produced GW		=ROUNDUP(E4/24;0	2	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((("We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504
5 1 =5	SUM('Wearout energy input'!F3:F8762)/100		=ROUNDUP(E5/24;0	3	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
6 2 =S	SUM['Wearout energy input'!G3:G8762]/10		=ROUNDUP(E6/24;0	4	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504
7 3 =5	SUM('Wearout energy input'IH3:H8762)/10		=ROUNDUP(E7/24;0	5	=0,015504753*((('Wearou	t =0,015504753*(((*	V∈=0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We=0	,015504753*	(('W∈=0,015504'
8 4 =S	SUM('Wearout energy input'II3:18762)/1000		=ROUNDUP(E8/24;0	6	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne=0,015504753*(((We =0,015504753	*((('We=0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
9 5 =5	SUM('Wearout energy input'IJ3:J8762)/100		=ROUNDUP(E9/24;0	7	=0,015504753*((('Wearou	t =0,015504753*((('	Ne=0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We=0	,015504753*	(('We =0,015504'
10 6 =5	SUM['Wearout energy input'!K3:K8762]/100		=ROUNDUP(E10/24;	8	=0,015504753*((('Wearou	t+=0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
11 7 =5	SUM('Wearout energy input'!L3:L8762)/100		=ROUNDUP(E11/24;	9	=0,015504753*((('Wearou	it =0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((("We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
12 8 =5	SUM['Wearout energy input'IM11:M8770]/:		=ROUNDUP(E12/24;	10	=0,015504753*((('Wearou	it+=0,015504753*(((1	Ne =0,015504753*(((We =0,015504753	*((('We=0,015	5504753*((('We	=0,0155047	53*((('We =0	,015504753*	(('We =0,015504)
13 9 =5	SUM['Wearout energy input'!N3:N8/62]/10		=ROUNDUP(E13/24;	11	=0,015504753*((('Wearou	it =0,015504753*(((1	We =0,015504753*(((We =0,015504753	*((('We=0,015	504753*((('We	=0,0155047:	53*((('We =0	,015504753*	(('We =0,01550
14 10 5	SUM(wearout energy input 103:08762)/10		=ROUNDUP(E14/24;	12	=0,015504753*((('Wearou	t =0,015504753*(((1	We =0,015504753*(((We =0,015504753	*((("We =0,015	504753*((('We	=0,01550473	53*((('We=0	0155047537	(('We =0,015504
16 12	SUM(wearout energy input (PS.P8762)/10		-ROUNDUP(E15/24;	14	-0,015504753 ([[wearou	-0.015504755 (((1	We =0,015504755 ([[We -0,013304733	(((We =0,013	5504755 ([[W	-0,01330473	55 (((WE-0	015504753	([WE =0,015504
17 12 -5	SUM/Wearout opermy input (25:02/02)/10		=ROUNDUP(E10/24;	15	=0.015504752*((('Wearou	+ -0.015504753*(((1	Ne =0,015504753 ([[We =0,015504753	111 We =0,015	504753 (((W	-0.01550475	53*((('We=0	015504752*	(('We=0,015504'
18 14	SUM/Wearout energy input INS.NO702//100		=ROUNDUP(E18/24;	16	=0.015504753*((('Wearou	t -0.015504753*((()	Ne =0.015504753*(((We =0.015504753	*((('We=0.015	504753*((('W	-0.01550475	53*///We=0	015504753*	(('We=0,015504'
19 15	SUM/Wearout energy input 153:56762)/100		-ROUNDUP(E19/24)	17	-0.015504753*((('Wearou	t -0.015504753*((()	Ne -0 015504753*(((We -0.015504753	*((('We -0.015	504753*((('W	-0.01550475	53*((('We-0	015504753*	(('We -0.015504'
20 16 =5	SUM('Wearout energy input'IU3:U8762)/10		=ROUNDUP(F20/24)	18	=0.015504753*((('Wearou	t =0.015504753*((()	Ne =0.015504753*(((We =0.015504753	*((('We =0.015	504753*((('W	=0.0155047	53*((('We =0	015504753*	(('We =0.015504'
21 17 =5	SUM('Wearout energy input'IV3:V8762)/10		=ROUNDUP(F21/24)	19	=0.015504753*((('Wearou	t =0.015504753*(((*)	Ne =0.015504753*(()	We =0.015504753	*((("We =0.015	504753*((('W	=0.0155047	53*((('We =0	015504753*	(('We =0.015504'
22 18 =5	SUM('Wearout energy input'IW3:W8762)/1		=ROUNDUP(E22/24:	20	=0.015504753*((('Wearou	t =0.015504753*(((*)	Ne =0.015504753*(((We =0.015504753	*((("We =0.015	504753*((('W	=0.0155047	53*((('We =0	.015504753*	(('We =0.015504'
23 19 =5	SUM('Wearout energy input'!X3:X8762)/100		=ROUNDUP(E23/24;	21	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0.015504753	*((('We =0.015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0.015504'
24 20 =5	SUM('Wearout energy input'IY3:Y8762)/100		=ROUNDUP(E24/24;	22	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((("We =0,015	504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
25 21 =5	SUM('Wearout energy input'IZ3:Z8762)/100		=ROUNDUP(E25/24;	23	=0,015504753*((('Wearou	t =0,015504753*(((*	N∈=0,015504753*(((We =0,015504753	*((("We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
26 22 =5	SUM('Wearout energy input'IAA3:AA8762)/		=ROUNDUP(E26/24;	24	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((("We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
27 23 =5	SUM('Wearout energy input'IAB3:AB8762)/		=ROUNDUP(E27/24;	25	=0,015504753*((('Wearou	t =0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((("We =0,015	5504753*((('We	=0,01550473	53*((('We=0	,015504753*	(('W∈=0,015504
28 24 =5	SUM('Wearout energy input'!AC3:AC8762)/		=ROUNDUP(E28/24;	26	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
29 25 =5	SUM('Wearout energy input'!AD3:AD8762)/		=ROUNDUP(E29/24;	27	=0,015504753*((('Wearou	t =0,015504753*((('	N∈=0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550473	53*((('We =0	,015504753*	(('We =0,015504'
30			=ROUNDUP(E30/24;	28	=0,015504753*((('Wearou	t =0,015504753*(((*	Ne=0,015504753*(((We =0,015504753	*((('We=0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
31			=ROUNDUP(E31/24;	29	=0,015504753*((('Wearou	t =0,015504753*((('	N∈=0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550473	53*((('We =0	,015504753*	(('W∈=0,015504'
32			=ROUNDUP(E32/24;	30	=0,015504753*((('Wearou	t =0,015504753*((('	We =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
33			=ROUNDUP(E33/24;	31	=0,015504753*((('Wearou	t =0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('W∈=0,015504'
34			=ROUNDUP(E34/24;	32	=0,015504753*((('Wearou	t =0,015504753*((('	Ne =0,015504753*(((We =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We =0	,015504753*	(('We =0,015504'
35			=ROUNDUP(E35/24;	33	=0,015504753*((('Wearou	t+=0,015504753*((('	Ne=0,015504753*(((We =0,015504753	"((('We =0,015	5504753*((('We	=0,01550473	53*((('We=0	,015504753*	(('We =0,015504'
36			=ROUNDUP(E36/24;	34	=0,015504753*((('Wearou	it =0,015504753*((('	W∈=0,015504753*(((W€ =0,015504753	*((('We =0,015	5504753*((('We	=0,01550475	53*((('We=0	,015504753*	(['We =0,015504] -
Random Failure rat	te Lightning Failure rate Icing Failur	e rate Overload Failur	e rate Wearout F	ailure rate YearsD	laysHours Wearout en	ergy input converter	Wearout energy	input Average	wind speed p	ber hour	(+)		: 4	F 1006

Overload failure rate

Autoline 🖅 🖶 😏 - 🕫 - 🕫		Rel	ability converter system using failure be	haviors (replacement) - Excel				
File Home Insert Page Layout	Formulas Data Review View Add-in:	s Help 🔎 Tell me what you wan						🖻 Share
fx Insert Function AutoSum Recently Financial Logical Function	Text Date & Lookup & Math & More	Create from Selection Define Name	ce Precedents Show Formulas ce Dependents 'Show Formulas move Arrows - @ Evaluate Formula Formula Auditing	Window Calculate Now Calculation Calculate Shee Calculation	¢.			~
<u>F5 ▼ : × √ fe</u> at	MAX(0;0,01171838*('Average wind speed per ho	our'IF5-3,5)*(0,5/13))						^
al D E	FG	н	J K	L L	M N	0	P (2 R ·
2 Days Hours	1 2 3	4	5 6	7 8	9	10 11	12	13
3 -POUNDUP(53/24-0) 1	-MAY/0-0 01171828* -MAY/0-0 01171828* -MA	AX/0-0 01171828* -MAAX/0-0 0117182	* -MAX/0-0 01171838* -MAX/0-0 (1171838* -MAY/0-0 01171828* -MAY/0	-0.01171928* -MAX(0-0.011	71828*-MAX(0:0.01171838*-h	4AX(0-0.01171828* -MAX(0-0	01171939* -MAX/0-0.01
4 =ROUNDUR(54/24/012	-MAX(0,0,01171030 -MAX(0,0,01171030 -MA	x/0-0 011718387 -446 X/0-0 0117183	-http://www.com/com/	1171838* -MAX VOO 01171838* -MAX VO	0.01171030 -MAX(0:0.011	719397 -MAX(0:0 011719397 -h	1000 01171838" -MAX/0:0	01171030 -MAX(0:0.01
5 -ROUNDUR(55/24-012	-MAX(0,0,01171838 -MAX(0,0,01171838 -M	AV(0-0 01171020" -AAAV(0-0 01171020		1171838 -MAX(0,0,01171838 -MAX(0	-0.01171828* -MAAY(0:0.011	71920* -MAX(0,0,01171920* -8	4AV(0-0 01171938" -MAX(0,0,	01171928* -MAX(0-0.01
5 -ROUNDUD(56/24,0) 5	-344 X (0,0,01171030 -344 X (0,0,01171030 -444	AN(0,0,01171030 -MAAX/0,0 0117103		1171830 -HAK(0,0,01171830 -HAK(0	0.01171030 -MAX(0,0,011	719397	14X10-0 01171030 -144X10-0	01171030 -MAX(0,0,01
7 -POUNDUD(57/24-0) 5	-MAX/0-0 01171939* -MAX/0-0 01171939* -MA	×V/0-0 01171929* -446 ×/0-0 01171929		1171920* -846 V/0-0 01171920* -846 V/0	-0.01171929* -MAX(0:0.011	71929* -545 V(0-0 01171929* -5	44X(0-0.01171928* - MAX(0-0	01171929* - MAX(0-0.01
8 =ROUNDUR(ER/24,0) 5	-MAX/0-0 01171838 -MAX/0-0 01171838 -MA	AX(0,0,01171838 -MAAX(0,0,0117183	* -544 X (0:0 01171838* -544 X (0:0,0	1171838* -MAX(0,0,01171838* -MAX(0	0.01171838* =MAX(0:0.011)	71838" -MAX(0;0,01171838" -h	AX(0,0,01171838" -MAX(0,0,	01171838* -MAX(0,0,01
9 -ROUNDUR(E9/24:0) 7	-MAX/0-0.01171838* -MAX/0-0.01171838* -MA	AX/0-0 01171838* -MAX/0-0 01171838	* -MAX(0:0 01171838* -MAX(0:0 0	1171838* -MAX(0:0 01171838* -MAX(0	0.01171838* -MAX(0:0.011)	71838* -MAX(0:0.01171838* -h	4AX(0:0.01171838"-MAX(0:0	01171838* -MAX(0:0.01
10 =ROUNDUP(E10/24-08	=MAX(0:0.01171838* =MAX(0:0.01171838* =MA	AX(0:0.01171838* =MAX(0:0.0117183	*=MAX(0:0.01171838*=MAX(0:0.0	1171838* =MAX(0-0.01171838* =MAX(0	0.01171838* =MAX(0:0.011)	71838* =MAX(0:0.01171838* =h	AX(0:0.01171838* =MAX(0:0	01171838* =MAX(0:0.01
11 =ROUNDUP(F11/24-19	=MAX(0:0.01171838*=MAX(0:0.01171838*=MA	AX(0:0.01171838*=MAX(0:0.0117183	*=MAX(0:0.01171838*=MAX(0:0.0	1171838*=MAX(0:0.01171838*=MAX(0	0.01171838* =MAX(0:0.011)	71838* =MAX(0:0.01171838* =N	AX(0:0.01171838*=MAX(0:0	01171838*=MAX(0:0.01
12 =ROUNDUP(E12/24:010	=MAX(0:0.01171838* =MAX(0:0.01171838* =MA	AX(0:0.01171838* =MAX(0:0.0117183	* =MAX(0:0.01171838* =MAX(0:0.0	1171838*=MAX(0:0.01171838*=MAX(0	0.01171838" =MAX(0:0.011)	71838" =MAX(0:0.01171838" =N	AX(0:0.01171838"=MAX(0:0.	01171838* =MAX(0:0.01
13 =ROUNDUP(E13/24:11	=MAX(0:0.01171838*=MAX(0:0.01171838*=MA	AX(0:0.01171838* = MAX(0:0.0117183	* =MAX(0:0.01171838* =MAX(0:0.0	1171838* =MAX(0:0.01171838* =MAX(0	0.01171838" =MAX(0:0.011)	71838" =MAX(0:0.01171838" =N	AX(0:0.01171838"=MAX(0:0.	01171838" =MAX(0:0.0
14 =ROUNDUP[E14/24:12	=MAX(0:0.01171838" =MAX(0:0.01171838" =MA	AX(0:0.01171838" =MAX(0:0.0117183	* =MAX(0:0.01171838* =MAX(0:0.0	1171838* =MAX(0:0.01171838* =MAX(0	0.01171838" =MAX(0:0.011)	71838" =MAX(0:0.01171838" =N	AX(0:0.01171838" =MAX(0:0.	01171838" =MAX(0:0.01
15 =ROUNDUP[E15/24:013	=MAX(0:0.01171838" =MAX(0:0.01171838" =MA	AX(0:0.01171838" =MAX(0:0.0117183	* =MAX(0:0.01171838* =MAX(0:0.0	1171838* =MAX(0:0.01171838* =MAX(0	0.01171838* =MAX(0:0.011)	71838* =MAX(0:0.01171838* =N	AX(0:0.01171838" =MAX(0:0.	01171838* =MAX(0:0.01
16 =ROUNDUP[E16/24;014	=MAX(0;0,01171838" =MAX(0;0,01171838" =M/	AX(0;0,01171838" =MAX(0;0,01171838	* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838* =MAX(0;0,011	71838" =MAX(0;0,01171838" =N	MAX(0:0,01171838" =MAX(0:0,	01171838* =MAX(0;0,01
17 =ROUNDUP(E17/24;015	=MAX(0;0,01171838*=MAX(0;0,01171838*=MA	AX(0;0,01171838* =MAX(0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	0,01171838*=MAX(0;0,011	71838* =MAX(0;0,01171838* =N	AX(0;0,01171838"=MAX(0;0,	01171838*=MAX(0;0,01
18 =ROUNDUP(E18/24;016	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838" =MAX(0;0,01171838	* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838" =MAX(0;0,011	71838* =MAX(0;0,01171838* =N	AX(0;0,01171838* =MAX(0;0,	01171838* =MAX(0;0,01
19 =ROUNDUP(E19/24;017	=MAX(0;0,01171838" =MAX(0;0,01171838" =MA	AX(0;0,01171838* =MAX(0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838"=MAX(0;0,011	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838"=MAX(0;0,	01171838*=MAX(0;0,01
20 =ROUNDUP(E20/24;018	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838* =MAX(0;0,01171838	* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838" =MAX(0;0,011)	71838*=MAX(0;0,01171838*=h	AX(0:0.01171838* =MAX(0:0.	01171838* =MAX(0;0,01
21 =ROUNDUP(E21/24;019	=MAX(0;0,01171838" =MAX(0;0,01171838" =MA	AX(0;0,01171838" =MAX(0;0,01171838	8" =MAX(0;0,01171838" =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838" =MAX(0;0,011)	71838* =MAX(0;0,01171838* =N	MAX(0;0,01171838"=MAX(0;0,	01171838* =MAX(0;0,01
22 =ROUNDUP(E22/24;020	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838" =MAX(0;0,01171838	* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838" =MAX(0;0,011)	71838" =MAX(0;0,01171838" =N	AX(0;0,01171838"=MAX(0;0,	01171838"=MAX(0;0,01
23 =ROUNDUP(E23/24;021	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838" =MAX(0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838* =MAX(0;0,011)	71838* =MAX(0;0,01171838* =N	MAX(0;0,01171838"=MAX(0;0,	01171838* =MAX(0;0,01
24 =ROUNDUP(E24/24;022	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838* =MAX(0;0,01171838	5* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	0,01171838"=MAX(0;0,011	71838* =MAX(0;0,01171838* =N	MAX(0;0,01171838"=MAX(0;0,	01171838*=MAX{0;0,01
25 =ROUNDUP(E25/24;023	=MAX(0;0,01171838*=MAX(0;0,01171838*=MA	AX(0;0,01171838* =MAX(0;0,01171838	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	;0,01171838*=MAX(0;0,0113	71838* =MAX(0;0,01171838* =N	MAX(0;0,01171838*=MAX(0;0,	01171838*=MAX(0;0,01
26 =ROUNDUP(E26/24;024	=MAX{0;0,01171838* =MAX{0;0,01171838* =M/	AX(0;0,01171838* =MAX(0;0,01171838	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	0,01171838"=MAX(0;0,011	71838* =MAX(0;0,01171838* =N	AX(0;0,01171838*=MAX(0;0,	01171838* =MAX(0;0,01
27 =ROUNDUP(E27/24;025	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838"=MAX(0;0,01171838	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	;0,01171838"=MAX(0;0,0117	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838*=MAX(0;0,	01171838"=MAX(0;0,01
28 =ROUNDUP(E28/24;026	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838"=MAX(0;0,0117183	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838*=MAX{0;0,01171838*=MAX{0	;0,01171838"=MAX(0;0,011	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838"=MAX(0;0,	01171838*=MAX(0;0,01
29 =ROUNDUP(E29/24;027	=MAX(0;0,01171838"=MAX(0;0,01171838"=MA	AX(0;0,01171838" =MAX(0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838*=MAX(0;0,011)	71838*=MAX(0;0,01171838*=N	AAX(0;0,01171838"=MAX(0;0,	01171838* =MAX(0;0,01
30 =ROUNDUP(E30/24;0 28	=MAX{0;0,01171838* =MAX{0;0,01171838* =MA	AX{0;0,01171838" =MAX{0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	;0,01171838* =MAX(0;0,011)	71838* =MAX(0;0,01171838* =N	MAX(0;0,01171838"=MAX(0;0,	01171838*=MAX(0;0,01
31 =ROUNDUP(E31/24;029	=MAX(0;0,01171838*=MAX(0;0,01171838*=MA	AX(0;0,01171838" =MAX(0;0,01171838	5* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838* =MAX(0;0,011)	71838" =MAX(0;0,01171838" =N	MAX(0;0,01171838*=MAX(0;0,	01171838* =MAX(0;0,01
32 =ROUNDUP(E32/24;0 30	=MAX{0;0,01171838* =MAX{0;0,01171838* =MA	AX{0;0,01171838" =MAX{0;0,01171838	5* =MAX(0;0,01171538* =MAX(0;0,0	1171838* =MAX{0;0,01171838* =MAX{0	;0,01171838*=MAX(0;0,0113	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838* =MAX(0;0,	01171838* =MAX(0;0,01
33 =ROUNDUP(E33/24;031	=MAX(0;0,01171838*=MAX(0;0,01171838*=MA	AX(0;0,01171838* =MAX(0;0,01171838	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838*=MAX(0;0,01171838*=MAX(0	;0,01171838*=MAX(0;0,0113	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838*=MAX{0;0,	01171838*=MAX(0;0,01
34 =ROUNDUP(E34/24;0 32	=MAX(0;0,01171838* =MAX(0;0,01171838* =MA	AX(0;0,01171838* =MAX(0;0,01171838	8*=MAX(0;0,01171838*=MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838* =MAX(0;0,0113	71838* =MAX(0;0,01171838* =N	MAX{0;0,01171838* =MAX{0;0,	01171838" =MAX(0;0,01
35 =ROUNDUP(E35/24;033	=MAX(0;0,01171838*=MAX(0;0,01171838*=MA	AX(0;0,01171838* =MAX(0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838*=MAX(0;0,011	71838*=MAX(0;0,01171838*=N	MAX{0;0,01171838*=MAX{0;0,	01171838*=MAX(0;0,01
36 =ROUNDUP(E36/24;034	=MAX{0;0,01171838* =MAX{0;0,01171838* =MA	AX{0;0,01171838* =MAX{0;0,01171838	8* =MAX(0;0,01171838* =MAX(0;0,0	1171838* =MAX(0;0,01171838* =MAX(0	;0,01171838*=MAX(0;0,011	71838*=MAX(0;0,01171838*=N	MAX(0;0,01171838"=MAX(0;0,	01171838*=MAX(0;0,01
Random Failure rate L Ready Calculate	ightning Failure rate Icing Failure rate Ov	verload Failure rate Wearout Failu	re rate VearsDaysHours W	earout energy input converter Wea	rout energy input Avera	ge wind speed per hour	· · · · · · · · · · · · · · · · · · ·	< > > + 100%

Icing failure rate

- Antology (10) 🔒 🍤 - C ⁰ - 🕫					Reliability converter syste	m using failure behavior	s (replacement) - Excel					fark Haring [23]	
File Home Insert Page Layout	Formulas Data		Add-ins Help	C Tell me what you w	ent to do								년 Share
fx Insert Function AutoSum Recently Financial Logica Used - Function	I Test Date & Lookup - Time - Reference ion Library	& Math & More	Name Manager Create Defined	e Name - Sei Formula - og e from Selection R, I Names	Trace Precedents 1995 Trace Dependents 1996 Remove Arrows 1997 (3) B Formula	Show Formulas Error Checking - Evaluate Formula Auditing	tch dow Options - Calculation Calculation	alculate Now alculate Sheet Ition					^
D1 * × √ fe													î
1 D E	F	G	н	1	1	к	L	м	N	0	P	Q	R =
1		12	12	12	10	12	1			123	1.02	1.2	Year
2 Days Hours	1	2	3	4	5	6	7	8	9	10	11	12	13
3 =ROUNDUP(E3/24;0)1	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
4 =ROUNDUP(E4/24;0) 2	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
5 =ROUNDUP(E5/24;0) 3	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
6 =ROUNDUP(E6/24;0)4	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
7 =ROUNDUP(E7/24;0)5	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
8 =ROUNDUP(E8/24;0) 6	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
9 =ROUNDUP(E9/24;0) 7	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/1)
10 =ROUNDUP(E10/24;08	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
11 =KOUNDUP[E11/24;09	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
12 =ROUNDOP(E12/24,010	=0,05*(0,3/10)	10,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05-(0,3/10)	=0,05-(0,3/10)	10,05*(0,3/10)	=0,05*(0,3/10)	10,05-(0,3/10
13 =ROUNDUP[E13/24;011	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/1
14 =ROUNDUP[E14/24;012	0,05*(0,3/10)	=0,05*(0,3/10)	0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
15 =ROUNDUP[E15/24;013	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
16 =ROUNDUP(E10/24),14	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,03*(0,3/10)	=0,03*(0,3/10)	=0,05 (0,3/10)	=0,05-(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
17 -ROUNDUP(E17/24,015	-0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,3/10)	-0,03 (0,3/10)	-0,03 (0,3/10)	-0,05 (0,5/10)	-0,03 (0,3/10)	-0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,5/10
18 =ROUNDUP[E18/24,16	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	-0.05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	-0.05*(0,3/10)	=0,05*(0,3/10)	-0.05*(0,3/1)
19 =ROUNDUP[E19/24,17	-0.05*(0,3/10)	=0,05 (0,5/10)	-0.051(0,3/10)	=0,05 (0,3/10)	-0.05 (0,3/10)	=0,05 (0,5/10)	-0,05 (0,5/10)	-0,05 (0,5/10)	=0,03 (0,3/10)	-0.057(0,3/10)	-0,05*(0,3/10)	-0,03 (0,3/10)	-0.05 (0,5/10
20 =ROUNDUP(E20/24)018	0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/1)
21 =ROUNDUP[E21/24,15	=0,05 (0,3/10)	=0,05*(0,3/10)	=0,05 (0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05 (0,3/10)	=0,05 (0,3/10)	=0,05 (0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,03*(0,3/10)	=0,05 (0,3/1)
22 =ROUNDUD(522/24, 20	-0.05*(0.3/10)	=0,05 (0,3/10)	=0,05 (0,3/10)	=0,05 (0,3/10)	=0,03 (0,3/10)	=0,05*(0,3/10)	=0,05 (0,3/10)	=0,05 (0,3/10)	=0.05*(0.3/10)	=0,05 (0,3/10)	=0,05*(0,3/10)	=0,05 (0,3/10)	-0.05*(0.3/10
24 -POUNDUP(524/24/22)	-0.05*(0.3/10)	=0,05*(0,3/10)	=0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10
25 -POUNDUD(525/24/22	-0.05 (0,3/10)	-0.051(0.3/10)	-0.057(0,3/10)	-0.05*(0,3/10)	-0.057(0.3/10)	-0.057(0.3/10)	-0.057(0,3/10)	-0.057(0.3/10)	-0.057(0,3/10)	-0.057(0.2/10)	-0.057(0,3/10)	-0.057(0,3/10)	-0.0510.2/10
26 -POUNDUD(2)/24,025	-0,05*(0,3/10)	=0,05 (0,3/10)	=0,05 (0,5/10)	-0,05 (0,3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0,05 (0,3/10)	-0,05 (0,3/10)	-0,05*(0,3/10)	-0,05 (0,3/10)	-0,05 (0,3/10)	-0,05*(0,3/1)
27 -POUNDUD(527/24-25	-0.05*/0.2/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.2/10)	-0.05*(0.2/10)	-0.05*(0.2/10)	-0.05*(0.2/10)	-0.05*(0.2/1/
28 =ROUNDUP(E28/24-026	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
29 =BOUNDUP(F29/24-127	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/1(
30 -ROUNDUR(E30/24/028	-0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	=0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	=0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10
31 -ROUNDUP(E31/24/029	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10
32 =ROUNDUP(E32/24:030	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
33 =ROUNDUP(E33/24:031	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
34 =ROUNDUP(E34/24:032	=0,05*(0,3/10)	=0.05*(0.3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0.05*(0,3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
35 =ROUNDUP(E35/24:033	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
36 =ROUNDUP(E36/24:034	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
	Lightning Failure rate	Icing Failure rate	Overload Failure	rate Wearout Fa	ailure rate YearsDa	aysHours Wearou	t energy input convert	er Wearout ener	gy input Average	wind speed per hour	()		+ 1096

Lightning failure rate

Autoliare 💷 🔒 🍤 - 🕐 - 🕫					leliability converter syste	m using failure behavio	ars (replacement) - Excel					fark Haring 🖽	- a ×
File Home Insert Page Layout	t Formulas Data	Review View	Add-ins Help 🔎	Tell me what you w	ent to do								
fx Insert Function AutoSum Recently Financial Logic Used	al Test Date & Lookup - Time - Reference tion Library	€ Interest of the second seco	Name Manager Define Na Define Na Create fro Defined Nam	me - Bot mula - og m Selection R F	Frace Precedents 1925 Frace Dependents 1946 Remove Arrows - (A) B Formula	how Formulas irror Checking - W ivaluate Formula Auditing	Attch indow Calculation Options - Calculatio	culate Now culate Sheet in					^
01 * X ~ fr													~
J D E	F	G	н	1	J	κ	Formula Bar	M	N	0	P	Q	R a
1			12		12		-			122			Year
Z Days Hours	1	2	3	4	5	0	/	8	9	10	11	12	13
3 =ROUNDUP(E3/24;0) 1	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
4 =ROUNDUP(E4/24;0) 2	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
5 =ROUNDUP(E5/24;0) 3	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
6 =ROUNDUP(E0/24;0)4	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/1(
2 =ROUNDUP(E//24,0) 5	=0,05 (0,5/10)	=0,05 (0,3/10)	-0.051(0,3/10)	=0,05 (0,5/10)	-0.05 (0,3/10)	=0,05*(0,3/10)	-0.05 (0,5/10)	-0.05*(0,3/10)	-0.051(0,3/10)	-0.05*(0.3/10)	-0,05 (0,3/10)	=0,05*(0,3/10)	-0.051(0.3/1)
0 =ROUNDUD(58/24,0) 0	-0.05*(0.3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05 (0,3/10)	=0,05*(0,3/10)	-0.05*(0.3/10)	=0,05*(0,3/10)	-0.05*(0.3/10)	=0,05 (0,3/10)	-0.05*(0.3/10)	=0,05 (0,5/10)	-0.05*(0.3/10)	-0.05*(0.3/10
10 =ROUNDUP(E10/34-08	=0.05*(0.3/10)	=0,05*(0,3/10)	-0,05*(0,3/10)	=0,05 (0,3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0,05*(0,3/10)	-0,05 (0,3/10)	=0.05*(0.3/10)	=0,05*(0,3/10)	=0.05*(0.3/10)	-0,05*(0,3/10)	-0,05*(0,3/10
11 -ROUNDUR(E11/34-19	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.3/10)	-0.05*(0.2/10)	-0.05*(0.3/10
12 =ROUNDUP(E12/24-110	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
13 =ROUNDUP(E13/24-11)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/1
14 =ROUNDUP[F14/24:112	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/1
15 =ROUNDUP[E15/24:13	0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10)	=0.05*(0.3/10
16 =ROUNDUP(E16/24:014	=0.05*(0.3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0.05*(0.3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
17 =ROUNDUP(E17/24;015	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
18 =ROUNDUP(E18/24;016	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
19 =ROUNDUP(E19/24;017	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
20 =ROUNDUP(E20/24;018	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0.05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
21 =ROUNDUP(E21/24;019	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
22 =ROUNDUP(E22/24;0 20	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
23 =ROUNDUP(E23/24;021	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
24 =ROUNDUP(E24/24;022	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
25 =ROUNDUP(E25/24;023	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
26 =ROUNDUP(E26/24;0 24	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
27 =ROUNDUP(E27/24;025	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/1(
28 =ROUNDUP(E28/24;026	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
29 =ROUNDUP(E29/24;027	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
30 =ROUNDUP(E30/24;0 28	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
31 =ROUNDUP(E31/24;029	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
32 =ROUNDUP(E32/24;030	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
33 =ROUNDUP(E33/24;031	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
34 =ROUNDUP(E34/24;032	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
35 =ROUNDUP[E35/24;033	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10
30 =ROUNDUP(E36/24;034	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/10)	=0,05*(0,3/1(
Kandom Failure rate Ready Calculate	Lightning Failure rate	Icing Failure rate	Overload Failure rat	e Wearout Fa	ilure rate YearsDa	ysHours Wearo	ut energy input converter	Wearout energy	y input Average	wind speed per hour	•		 + 100%

Random failure rate

Autolieve (1999) 🔒 🍤 - 🕫 - 🕫		an an an an a		1	leliability converter syste	m using failure behavior	s (replacement) - Excel				,	fark Haring 🖽	- a x
File Home Insert Page Layout	Formulas Data	Review View	Add-ins Help	○ Tell me what you w	ent to do								년 Share
fx Insert Function AutoSum Recently Financial Logical Used	Text Date & Lookup Time ~ Reference an Library	& Math & More e * Trig * Functions	Name Create Manager Create Defined I	e Name ~ \$0 Formula - og e from Selection \$5 Names	Trace Precedents (5) Trace Dependents (6) Remove Arrows - (6) Formula	Show Formulas Error Checking – Wa Evaluate Formula Win Auditing	etch dow Options - Calculation Calculat	alculate Now alculate Sheet tion					^
A V X 1 * 11	=0,08*(0,2/16)												^
a D E	E	G	н	1.11	J	к	L	м	N	0	p	Q	R 🖷
1		2	2	1			7	0	0	10		12	Year
2 Days Hours	4	-0.08*/0.2/16)	3	4	2 -0.02*(0.2(16)	0	-0.08*(0.3/16)	0 0000 0 2/16)	9	-0.02*(0.2/16)	-0.092(0.2/16)	-0.00*(0.3/16)	-0.08*(0.2/14
4 =ROUNDUP[63/24,0]1	-0,08 (0,2/16)	-0.08*(0,2/16)	=0,08 (0,2/16)	=0,08 (0,2/16)	=0,08 (0,2/16)	=0,08*(0,2/16)	=0,08 (0,2/16)	-0,08 (0,2/10)	=0,08 (0,2/10)	-0,08 (0,2/10)	-0,08 (0,2/10)	-0,08 (0,2/10)	-0,08 (0,2/10
5 -ROUNDUD(ES/24-0)2	-0.08*(0,2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0,2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/16)	-0.08*(0.2/1/
6 =ROUNDUR(E6/24:0) 4	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/1/
7 =ROUNDUP[F7/24-015	=0.08*/0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/1)
8 =ROUNDUP[E8/24:0] 6	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/10
9 =ROUNDUP(E9/24:0) 7	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16
10 =ROUNDUP(E10/24:08	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16
11 =ROUNDUP[E11/24:09	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16
12 =ROUNDUP[E12/24:010	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16
13 =ROUNDUP(E13/24:011	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/16)	=0.08*(0.2/1
14 =ROUNDUP(E14/24;12	=0.08*(0.2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0.08*(0.2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
15 =ROUNDUP(E15/24;013	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
16 =ROUNDUP(E16/24;014	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
17 =ROUNDUP(E17/24;015	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
18 =ROUNDUP(E18/24;016	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
19 =ROUNDUP(E19/24;017	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
20 =ROUNDUP(E20/24;018	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
21 =ROUNDUP(E21/24;019	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
22 =ROUNDUP(E22/24;0 20	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
23 =ROUNDUP(E23/24;021	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
24 =ROUNDUP(E24/24;0 22	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
25 =ROUNDUP(E25/24;023	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
26 =ROUNDUP(E26/24;0 24	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/18
27 =ROUNDUP(E27/24;025	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
28 =ROUNDUP(E28/24;0 26	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
29 =ROUNDUP(E29/24;027	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
30 =ROUNDUP(E30/24;0 28	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
31 =ROUNDUP(E31/24;029	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
32 =ROUNDUP(E32/24;0 30	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
33 =ROUNDUP(E33/24;031	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
34 =ROUNDUP(E34/24;032	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16
35 =ROUNDUP[E35/24;033	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10
36 =ROUNDUP[E36/24;034	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/16)	=0,08*(0,2/10 +
Random Failure rate Ready Calculate	Lightning Failure rate	Icing Failure rate	Overload Failure	rate Wearout Fa	ailure rate YearsDa	aysHours Wearou	t energy input converte	er Wearout ener	gy input Average	wind speed per hour	•		 + 100%

Early failure rate

Autolieve (000) 🔒 🍤 - 🕫 - 🔹	Reliability-converter system using laluer behaviors (replacement) - Incel Mark Haring 🖽 — 🔿 🗙
File Home Insert Page Layout	Formulas Data Review View Addrins Help // Tell meshat you want to do
fx Insert Function AutoSum Recently Financial Logical Function	A Image: Construction of the sector of
F3 * × √ fc =0	. 16668*((YearDaysHours)153/1, 7999)~0, 7,71(0,3/12)
J D E	F G H I J K L M N O P Q R
2 Davs Hours	Year
a pourpupira (a cel r	
3 =ROUNDUP[E3/24;0]1	
4 =ROUNDUP(E4/24;0) 2	10,0606* ((YearSDay =0,1606* ((YearSDay =0,1606* ((YearSDay =0,1606* (YearSDay =0,1606* (
5 =ROUNDUP(E5/24)0)3	=0,10006,(Ideal203A)=0'10002,(Ideal203A)=0'10002,(Ideal203A)=0'10002,(Ideal203A)=0'10002,(Ideal203A)=0'10002,(Ideal203A)=0'10002,(Ideal20A)=0'1002,(Ideal20A)=0'1000,(Ideal20A)=0'1000,(Ideal20A)=0'10000,(
6 =ROUNDOP(E6/24;0)4	an'10006. (Liseazona) an'10002. (Liseazona) an'1002. (Liseazona) an'1002. (Liseazona) an'1002. (Liseazona) an'
7 =ROUNDUP(E//24,0)5	au,10000c ((rearSDa) = 0,1000c
8 =ROUNDUP(E8/24;0)6	au/1000cs ((Leat2DA) 4/0.1000cs ((Leat2DA) 4/0.1000cs (Leat2DA) 4/0.1000
9 =ROUNDUP(E)/24,017	
11 -ROUNDUP(E10/24,68	
12 -ROUNDUD(512/24,45	-0.70000 (Leaizota)-0.7000 (Le
12 =ROUNDUP[E12/24,010	au, Loode ((Tear)Luky vol, Loode ((Tear)Luky vol, Loode) (Tear)Luky vol, Loode)
14 -ROUNDUP[E13/24,011	100000 (Teas Duby ~ 0,10000 (Teas Duby ~ 0,10000 (Teas Duby ~ 0,10000)) (Teas Duby ~ 0,10000) (Teas Duby ~ 0,1
14 -ROOMDOP[E14/24,412	0.20000 [Lice12704] 4/10000 [Lice12704] 4/1000
15 -ROUNDUP[E15/24/15	
17 -ROUNDUD(E17/34-15	
18 -ROUNDUR(E18/34-16	
19 -ROUNDUP(E19/24-117	
20 =ROUNDUP(E20/24-118	Control (1998) 2012 (Second) (Second) (Second) 2012 (Se
21 -POUNDUD(21/34-19	
22 =ROUNDUP(E22/24/20	
22 =ROUNDUD(522/34-01	
24 =ROUNDUR(E24/34-622	1 16662*1/CarrDau d) 16662*1/Car
25 -ROUNDUR(E25/34-023	-D 15656*/(VaarcDav-D 16656*/(VaarcDav-D 16656*/(Va
26 =ROUNDUP(E26/24/24	1 16668*/(Vaarchau d) 1668*/(Vaarchau d)
27 =ROUNDUP(F27/24-125	1 16668*/(VearClay =1 1668*/(VearClay =1 1668*/(VearClay =1 16668*/(VearClay =1 16668*
28 =ROUNDUP[E28/24:026	0.16668*(/YearsDay =0.16668*(/YearsDay =0.1666
29 =ROUNDUP[E29/24:027	0.16668*((YearsDay =0.16668*((YearsDay =0.16668*((YearsDay =0.16668*((YearsDay =0.16668*((YearsDay =0.16668*(YearsDay =0.16668*
30 =ROUNDUP[E30/24:028	0.16668*((YearsDay = 0.16668*((YearsDay = 0.16668*(YearsDay = 0.16
31 =ROUNDUP[E31/24:129	D. 16668*1(YearCDay id: 16668*
32 =ROUNDUP(E32/24:030	10,16665*1(YearsDay = 0,16665*1(YearsDay = 0,16665*
33 =ROUNDUP[E33/24:031	-0.16668*(/YearsDay -0.1668*(/YearsDay -0.16668*(/YearsDay -0.16668*(/YearsDay -0.1666
34 =ROUNDUP(E34/24:032	10,16668*((YearsDay = 0,16668*((YearsDay = 0,16668*((YearsDay = 0,16668*((YearsDay = 0,16668*((YearsDay = 0,16668*(YearsDay = 0,1668*(YearsDay = 0,16668*(YearsDay = 0,16668*(YearsDay = 0
35 =ROUNDUP(E35/24:033	=0,16668" ((YearsDay =0,16668" ((YearsDay =0,16668") ((YaarsDay =0,1668") ((YaarsDay =0,1668") ((YaarsDay =0,1668") ((YaarsDay =0,16668") ((YaarsDay =0,16
36 =ROUNDUP(E36/24;034	=0,16668* {(YearsDay =0,16668* ((YearsDay =0,16668*)(YearsDay =0,16668*)(YearsDay =0,16668*)(YearsDay =0,16668*) = 0,16668* ((YearsDay =0,16668*) = 0,16668*) = 0,16668* = 0,1668* = 0,1668* = 0,16668* = 0,16668* = 0,16668* = 0,16668* = 0,16668* = 0,16668* = 0,16668* = 0,16668
Total Reliability Chart I Ready Calculate	allwre rates Aging Failure rate Early Failure rate Random Failure rate Uigthning Failure rate long Failure rate Overload Failure rate Wearout Failure rate VearsDaysHours Wearout energy input conve () : ()

Aging failure rate

Autolieve 🖅 🔒 🖘 🖑 - 🕫		Reliability converter system using failure behaviors (rep	lacement) - Excel		Mark Haring 🖽 — 🗗 🗙
File Home Insert Page Layout	Formulas Data Review View Add-ins Help 🔎 Tell me what				
fx Insert Function AutoSum Recently Financial Logical Used v Function	Ted Data & Lookup & Math & More Teme - Reference - Trig - Functions - Usary	Bo Trace Precedents 22 Show Formulas Car Trace Dependents 4 Forror Checking - Remove Arrows - & Evaluate Formula Formula Auditing	Calculate Now Calculation		^
F3 • × √ fe a),1884875*((YearsDaysHours1F3/13,2635)^1_5)*(0,7/17)				^
a D E	F G H	I J K	E .	MN	0 P Q .
2 Days Hours	1 2 3 4	5 6	7 8	9 10	11 12 13
4 #GOLNOUPE(4245) 2 ± 4 #GOLNOUPE(5245) 4 ± ± ± GOLNOUPE(5245) 4 ± <td< td=""><td>Light Straff (Intersity Control (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994)</td><td>84435°(Teard D. 4).884475°(Teard D. 4).884475°</td><td>16 Clearly 17 578482, 0 Gran 16 Clearly 17 578482, 0 Gran 16 Clearly 17 578442, 0 Gran 16 Clearly 17 578443, 0 Gran 16 Clearly 17 578444, 0 Gran 16 Clearly 17 57844, 0 Gran 17 Clearly 17 Clea</td><td>48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 1848475 48475 (Weard O., 1848475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 184445) 48475 (Wear</td><td>139 (YearD - L1848 7) YearD - L1848 7) YearD - L1847 7) Y</td></td<>	Light Straff (Intersity Control (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994) (1994) (1994) (1994) (1994) (1994) (1994) Light Straff (Intersity Control (1994)	84435°(Teard D. 4).884475°(Teard D. 4).884475°	16 Clearly 17 578482, 0 Gran 16 Clearly 17 578482, 0 Gran 16 Clearly 17 578442, 0 Gran 16 Clearly 17 578443, 0 Gran 16 Clearly 17 578444, 0 Gran 16 Clearly 17 57844, 0 Gran 17 Clearly 17 Clea	48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 184847 48475 (Weard O., 1848475 (Weard O., 1848475 48475 (Weard O., 1848475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 1848475) 48475 (Weard O., 184475 (Weard O., 184445) 48475 (Wear	139 (YearD - L1848 7) YearD - L1848 7) YearD - L1847 7) Y
Total Reliability Chart	Failure rates Aging Failure rate Early Failure rate Random Failure r	ate Lightning Failure rate Icing Failure rate	Overload Failure rate Wea	arout Failure rate VearsDaysHours Wea	arout energy input conve (+)
Ready Calculate					田田田一一 + 10%

Reliability converter system

File Nume Num Nume Nume Nume	දුරු 50ar රංක රංක
JAX Image Image <t< th=""><th></th></t<>	
1 * <th></th>	
2	•
م المراجع من المراجع	TF
18.11. ListMinderFaurueritistIStICCC6:URLT_arf AuruerItistIStICCC201.INVC68 ListMinderFauruerItistIStICCC201.INVC61 ListMinderFauruerItistIStICC201.INVC61 ListMinderFauruerItistIStICC201.INVC61 ListMinderFauruerItistIStICC201.INVC61 ListMinderFauruerItistIStICC201.INVC61 ListMinderFau	e (+) : () +

Overview PM and CM costs

				R	PM (CO)										CM (C1)	
Total vessel cost [€] 516500	0	Number of T technicians		eplacement	Vaintenance ii action		906000	Total vessel cost [€]			2	Number of technicians		Replacement	Maintenance action	
Total downtime cost [€] 11713,5	0	ravel time (h)		0	Pre- nspections CTV [€]		309894	Total downtime cost [€]				Travel time (h)		3500	Pre- inspections CTV [€]	
Fotal PM cost Total labor cost [€] 0	0	Inspection time (h)	Labor cost	UL	Vessel type		1760	Total labor cost [€]	Total PM co:		00	Inspection time (h)	Labor cost	JU	Vessel type	
Total spare- part cost [€] 13000	80	Hourly rate (€/h)		0	Mobilization time (h)		13000	Total spare- part cost [€]	54		80	Hourly rate (€/h)		1440	Mobilization time (h)	
Total PM cost [€] 541213,5	0	Total labor cost [€]		114000	Mobilization costs [€]	Vessel cos	1230654	Total PM cost [€]			1760	Total labor cost [€]		500000	Mobilization costs [€]	Vessel
				140000	Day rate [€]	Ŧ								140000	Day rate [€]	cost
	converter system	Component	Spare-I	5833,33333	Hourly rate [€]						converter system	Component	Spare par	5833,333333	Hourly rate [€]	
	13000	Cost [€]	part cost	57	Maintenance time (h)						13000	Cost	t cost [€]	57	Maintenance time (h)	
				12	Travel time (h)									12	Travel time (h)	
				516500	Total vessel cost [€]									906000	Total vessel cost [€]	
					N											
				1	mber of f WT										Number o WT	
				0,1	·eed-in price (€ kW/h)									0,	f Feed-in price (€ kW/h)	
				5000	Turbine size (kW) t									1 500	e Turbine size (kW)	
				0,411	Capacity factor (%)									0 0,41:	Capacity factor (%)	
				0	Downtime travel time CTV (h)	Downtime co:								3	Downtime travel time CTV (h)	Downtime (
				0	Downtime pre- inspection (h)	st								8	Downtime pre- inspection (h)	ost
				0	Downtime mobilization time (h)									1440	Downtime mobilization time (h)	
				57	Duration maintenance (h)									57	Duration maintenance (h)	
				11713,5	Total downtime cost [€]									309894	Total downtime cost [€]	

Optimal fixed operational time

AutoSav	e 🖲 CH 🖥	∃ 5-∂-∓			Optimal policy and costs	of converter system (replac	ment) - Excel		Mark Ha	ing 🖽 —	a x
	Home In	nert Page Largert Formula	Data Review View Ad	duine Helo C	Tell me what you want to do						LC2 Share
		ioent ruge cayour ronnicitor		and nop y	Terme mat you man to up						
fx Insert Function	AutoSum Rec v Usi	tently Financial Logical Text Date 8 ed Financial Logical Text Date 8 Function Library	Lookup & Math & More Reference ~ Trig ~ Functions ~	Name State for the Create for the Cr	Iame ~ Bo Trace Precedents 202 Sho ormula - of Trace Dependents for Error rom Selection % Remove Arrows ~ (2) Eva mes Formula Au	ov Formulas or Checking ~ eluate Formula window	Calculation				~
G106	•	× ✓ <i>J</i> ==('Costs PM'!:	\$G\$14*C106}+{'Costs CM'!\$G\$14'	'D106)							Â
	A	В	с	D	E	F	G	н	1	1	
1											
2											
3											
						$(\overline{R}(t)) dt$					
•		G(*)	Supplied B(T)	1.P/T)	Transported rule	J ₀ ((t) (t)	FCC	G(t) Replacement	Min		
6		0	=\C\Lisers\Liser\Drophox\Pr	=1.06	0.00000001	-66	=//Costs PM/160\$14*C61+//	Generation Construction	-IE(H6-MIN(\$H\$6-\$H\$106)-		
7		0.25	= Cr(Users)User\Dropbox(P	-1-C7	=AVERAGE(C6*C7)*\$8\$7	-SUM(\$E\$6:E7)	=('Costs PM' \$G\$14*C7)+('	-67/57	-IE(H7-MIN(\$H\$6;\$H\$106);		_
0		0.5	-'C:\Users\User\Dropbox\P	-1.08	-AVERAGE(C7:C8)*\$8\$7	-SUM(\$E\$6-E8)	-('Costs PM4'(\$G\$14*CR)+/'		-IE(HR-MIN(\$H\$6-\$H\$106))		2000000
0		0.75	-'C:\Users\User\Dronbox\Pr	-1-C9	-AVERAGE(CR:C9)*\$8\$7	-SLIM(\$E\$6-E9)	-//Costs PM/190914 C0)+/	-69/59	-IE(NO-MIN(\$155-\$15100))		
10		1	='C:\Users\User\Dronbox\Pr	=1-C10	=AVERAGE(C9:C10)*\$8\$7	=SUM(\$E\$6:E10)	=('Costs PM' \$G\$14*C10)+	(=G10/E10	=IE(H10=MIN(\$H\$6;\$H\$106		1800000
11		1.25	='C-\Users\User\Dropbox\Pr	=1-C11	=AVERAGE(C10:C11)*\$8\$7	=SUM(\$E\$6:E11)	=['Costs PM'!\$G\$14*C11)+	=611/F11	=IF(H11=MIN(\$H\$6:\$H\$106		1600000
12		15	='C'\Users\User\Drophox\Pr	=1.012	=AVERAGE(C11:C12)*\$8\$7	=SUM(\$E\$6-E12)	=('Costs PM'ISG\$14*C12)+	=612/F12	=IE(H12=MIN(\$H\$6:\$H\$106		1400000
13		1.75	='C'\Users\User\Drophox\Pr	=1.013	=AVERAGE(C12:C13)*\$B\$7	=SUM(\$E\$6-E13)	=['Costs PM'I\$G\$14*C13]+	G13/F13	=IE(H13=MIN(\$H\$6:\$H\$106		
4		2	='C:\User\User\Dropbox\Pr	=1-C14	=AVFRAGE(C13:C14)*\$8\$7	=SUM(SES6:E14)	=('Costs PM'156514*C14)+	=614/F14	=IF(H14=MIN(SH56:SH5106		120000
15		2.25	='C:\Users\User\Dropbox\Pr	=1-C15	=AVERAGE(C14:C15)*\$B\$7	=SUM(\$E\$6:E15)	=('Costs PM' \$6\$14*C15)+	=G15/F15	=IF(H15=MIN(\$H\$6:\$H\$106		1000000
16		2.5	='C:\Users\User\Dropbox\Pr	=1-C16	=AVERAGE(C15:C16)*\$B\$7	=SUM(\$E\$6:E16)	=('Costs PM'!\$G\$14*C16)+	=G16/F16	=IF(H16=MIN(\$H\$6:\$H\$106		800000
17		2.75	='C:\Users\User\Dropbox\Pr	=1-C17	=AVERAGE(C16:C17)*SB\$7	=SUM(\$E\$6:E17)	=['Costs PM'!\$G\$14*C17)+	=G17/F17	=IF(H17=MIN(SH\$6:SH\$106		
8		3	='C:\Users\User\Dropbox\Pr	=1-C18	=AVERAGE(C17:C18)*\$B\$7	=SUM(\$E\$6:E18)	=('Costs PM'!\$G\$14*C18)+	G18/F18	=IF(H18=MIN(SHS6:SHS106		600000
19		3.25	='C:\Users\User\Dropbox\Pr	=1-C19	=AVERAGE(C18:C19)*\$B\$7	=SUM(\$E\$6:E19)	=['Costs PM'!\$G\$14*C19]+	=G19/F19	=IF(H19=MIN(SH\$6:SH\$106		400000
20		3.5	='C:\Users\User\Dropbox\Pr	=1-C20	=AVERAGE(C19:C20)*\$B\$7	=SUM(\$E\$6:E20)	=('Costs PM'I\$G\$14*C20)+	=G20/F20	=IF(H20=MIN(\$H\$6:\$H\$106		200000
21		3.75	='C:\Users\User\Dropbox\Pr	=1-C21	#AVERAGE(C20:C21)*\$B\$7	=SUM(\$E\$6:E21)	=('Costs PM'ISG\$14*C21)+	=G21/F21	=IF(H21=MIN(\$H\$6:\$H\$106		200000
2		4	='C:\Users\User\Dropbox\Pr	=1-C22	=AVERAGE(C21:C22)*SB\$7	=SUM(\$E\$6:E22)	=('Costs PM'!\$G\$14*C22)+	=G22/F22	=IF(H22=MIN(\$H\$6;\$H\$106		c
23		4,25	='C:\Users\User\Dropbox\Pr	=1-C23	=AVERAGE(C22:C23)*\$B\$7	=SUM(\$E\$6:E23)	=('Costs PM' \$G\$14*C23)+	G23/F23	=IF(H23=MIN(\$H\$6;\$H\$106		
24		4,5	='C:\Users\User\Dropbox\Pr	=1-C24	=AVERAGE(C23:C24)*\$B\$7	=SUM(\$E\$6:E24)	=('Costs PM'!\$G\$14*C24)+	'=G24/F24	=IF(H24=MIN(\$H\$6:\$H\$106		
25		4,75	='C:\Users\User\Dropbox\Pr	=1-C25	=AVERAGE(C24:C25)*\$B\$7	=SUM(\$E\$6:E25)	=('Costs PM'!\$G\$14*C25)+	=G25/F25	=IF(H25=MIN(\$H\$6:\$H\$106		
26		5	='C:\Users\User\Dropbox\Pr	=1-C26	=AVERAGE(C25:C26)*\$B\$7	=SUM(\$E\$6:E26)	=('Costs PM'!\$G\$14*C26)+	=G26/F26	=IF(H26=MIN(\$H\$6:\$H\$106		
27		5,25	='C:\Users\User\Dropbox\Pr	=1-C27	=AVERAGE(C26:C27)*\$B\$7	=SUM(\$E\$6:E27)	=('Costs PM'!\$G\$14*C27)+	=G27/F27	=IF(H27=MIN(\$H\$6:\$H\$106		
28		5,5	='C:\Users\User\Dropbox\Pr	=1-C28	=AVERAGE(C27:C28)*\$B\$7	=SUM(\$E\$6:E28)	=('Costs PM'!\$G\$14*C28)+	G28/F28	=IF(H28=MIN(\$H\$6:\$H\$106		
29		5,75	='C:\Users\User\Dropbox\Pr	=1-C29	=AVERAGE(C28:C29)*\$B\$7	=SUM(\$E\$6:E29)	=('Costs PM'l\$G\$14*C29)+	(=G29/F29	=IF(H29=MIN(\$H\$6:\$H\$106		
30		6	='C:\Users\User\Dropbox\Pr	=1-C30	=AVERAGE(C29:C30)*\$B\$7	=SUM(\$E\$6:E30)	=('Costs PM'!\$G\$14*C30)+	(*=G30/F30	=IF(H30=MIN(\$H\$6:\$H\$106		
31		6,25	='C:\Users\User\Dropbox\Pr	=1-C31	=AVERAGE(C30:C31)*\$B\$7	=SUM(\$E\$6:E31)	=('Costs PM'!\$G\$14*C31)+	('=G31/F31	=IF(H31=MIN(\$H\$6:\$H\$106		
32		6,5	='C:\Users\User\Dropbox\Pr	=1-C32	=AVERAGE(C31:C32)*\$B\$7	=SUM(\$E\$6:E32)	=('Costs PM'!\$G\$14*C32)+	(*=G32/F32	=IF(H32=MIN(\$H\$6:\$H\$106		
33		6,75	='C:\Users\User\Dropbox\Pr	=1-C33	=AVERAGE(C32:C33)*\$B\$7	=SUM(\$E\$6:E33)	=('Costs PM'!\$G\$14*C33)+	(*=G33/F33	=IF(H33=MIN(\$H\$6:\$H\$106		
34		7	='C:\Users\User\Dropbox\Pr	=1-C34	=AVERAGE(C33:C34)*\$B\$7	=SUM(\$E\$6:E34)	=('Costs PM'!\$G\$14*C34)+	(*=G34/F34	=IF(H34=MIN(\$H\$6:\$H\$106		
35		7,25	='C:\Users\User\Dropbox\Pr	=1-C35	=AVERAGE(C34:C35)*\$B\$7	=SUM(\$E\$6:E35)	=('Costs PM'!\$G\$14*C35)+	(*=G35/F35	=IF(H35=MIN(\$H\$6:\$H\$106)		
36		7,5	='C:\Users\User\Dropbox\Pr	=1-C36	=AVERAGE(C35:C36)*\$B\$7	=SUM(\$E\$6:E36)	=('Costs PM'I\$G\$14*C36)+	(¹ =G36/F36	=IF(H36=MIN(\$H\$6:\$H\$106		
	Opti	mal g(T) Costs PM Costs C	M (+)				: •				Þ
Ready										E	+ 100%

ECC different CM/PM cost ratios

AutoSave 💽 🖬 🕤 - 🕫 -			Sensitive	analysis (ratio maintenance costs) rep	placement converter system - E	xcel			Mark Haring 🖽 —	a x
	eyout Formulas Data I	Review View Add-in:								🖻 Shan
fx Insert Function	ogical Text Date & Lookup & Time - Reference	Math & More A	Define Name So Trace Ruse in Formula - Sector Selection Defined Names	Precedents Show Formulas Dependents Fror Checking we Arrows - & Evaluate Formula Formula Auditing	Watch Window Calculation Calculation Calculation	alculate Now alculate Sheet ition				~
66 * × √ β	=(C6**Cost inputs config	urations'!\$G\$5)+('ECL & E	CC for 9 categories'ID6*'Cost inputs co	nfigurations'1\$H\$5)						~
al B	с	D	E	F	G	н	1	J J	к	
1										
2										
3										
4				$\int_{0}^{T} R(t) dt$						
5 G(t)	Survival, R(T)	1-R(T)	Trapezoidal rule	ECL	Ratio 2	Ratio 2,5	Ratio 3	Ratio 3,5	Ratio 4	R
5 0	='C:\Users\User\Dropbox\P	=1-C6	0,000000001	=E6	=(C6*'Cost inputs configura	=(C6*'Cost inputs configure	=(C6*'Cost inputs configur	ra=(C6*'Cost inputs configur	a=(C6*'Cost inputs configu	ura=(C6*'Cost
7 0,25	='C:\Users\User\Dropbox\Pr	=1-C7	=AVERAGE(C6:C7)*\$B\$7	=SUM(\$E\$6:E7)	=(C7*'Cost inputs configura	=(C7*'Cost inputs configure	=(C7*'Cost inputs configur	a =(C7*'Cost inputs configur	=(C7*'Cost inputs configu	ura=(C7*'Cost
8 0,5	='C:\Users\User\Dropbox\Pr	=1-C8	=AVERAGE(C7:C8)*\$B\$7	=SUM(\$E\$6:E8)	=(C8*'Cost inputs configura	=(C8*'Cost inputs configure	=(C8*'Cost inputs configur	ra=(C8*'Cost inputs configur	a ={C8*'Cost inputs configu	ura=(C8*'Cost
0,75	='C:\Users\User\Dropbox\Pr	=1-C9	=AVERAGE(C8:C9)*\$8\$7	=SUM(\$E\$6:E9)	=(C9*'Cost inputs configura	=(C9*'Cost inputs configura	=(C9*'Cost inputs configur	ra=(C9*'Cost inputs configur	a=(C9*'Cost inputs configu	ura=(C9*'Cost
0 1	='C:\Users\User\Dropbox\Pr	=1-C10	=AVERAGE(C9:C10)*\$B\$7	=SUM(\$E\$6:E10)	=(C10*'Cost inputs configu	=(C10*'Cost inputs configu	=(C10*'Cost inputs configu	={C10*'Cost inputs configu	={C10*'Cost inputs config	gut=(C10*'Cos
1 1,25	='C:\Users\User\Dropbox\Pr	=1-C11	=AVERAGE(C10:C11)*\$B\$7	=SUM(\$E\$6:E11)	=(C11*'Cost inputs configu	=(C11*'Cost inputs configu	=(C11*'Cost inputs configu	=(C11*'Cost inputs configu	=(C11*'Cost inputs config	gut=(C11*'Co1
2 1,5	='C:\Users\User\Dropbox\Pr	=1-C12	=AVERAGE(C11:C12)*\$B\$7	=SUM(\$E\$6:E12)	=(C12*'Cost inputs configu	=(C12*'Cost inputs configu	=(C12*'Cost inputs configu	<pre>u =(C12**Cost inputs configu</pre>	={C12*'Cost inputs config	gur=(C12*'Cos
3 1,75	='C:\Users\User\Dropbox\Pr	=1-C13	=AVERAGE(C12:C13)*\$B\$7	=SUM(\$E\$6:E13)	=(C13*'Cost inputs configu	=(C13*'Cost inputs configu	=(C13*'Cost inputs configu	<pre>// =(C13*'Cost inputs configu</pre>	=(C13*'Cost inputs config	gur=(C13*'Cd
4 2	='C:\Users\User\Dropbox\Pr	=1-C14	=AVERAGE(C13:C14)*\$B\$7	=SUM(\$E\$6:E14)	=(C14*'Cost inputs configu	=(C14*'Cost inputs configu	=(C14*'Cost inputs configu	u =(C14*'Cost inputs configu	a ={C14*'Cost inputs config	gut=(C14*'Cos
5 2,25	='C:\Users\User\Dropbox\Pr	=1-C15	=AVERAGE(C14:C15)*\$B\$7	=SUM(\$E\$6:E15)	=(C15*'Cost inputs configu	=(C15*'Cost inputs configu	=(C15*'Cost inputs configu	=(C15*'Cost inputs configu	=(C15*'Cost inputs config	gut=(C15*'Cos
6 2,5	='C:\Users\User\Dropbox\Pr	=1-C16	=AVERAGE(C15:C16)*\$B\$7	=SUM(\$E\$6:E16)	=(C16*'Cost inputs configu	=(C16*'Cost inputs configu	=(C16*'Cost inputs configu	=(C16*'Cost inputs configu	={C16*'Cost inputs config	gut=(C16*'Cos
7 2,75	='C:\Users\User\Dropbox\Pr	=1-C17	=AVERAGE(C16:C17)*\$B\$7	=SUM(\$E\$6:E17)	=(C17*'Cost inputs configu	=(C17*'Cost inputs configu	=(C17*'Cost inputs configu	=(C17*'Cost inputs configu	=(C17*'Cost inputs config	gut=(C17*'Cos
8 3	='C:\Users\User\Dropbox\Pr	=1-C18	=AVERAGE(C17:C18)*\$B\$7	=SUM(\$E\$6:E18)	=(C18*'Cost inputs configu	=(C18*'Cost inputs configu	=(C18"Cost inputs configu	={C18**Cost inputs configu	=[C18"Cost inputs config	gut=(C18*'Cos
9 3,25	='C:\Users\User\Dropbox\Pr	=1-C19	=AVERAGE(C18:C19)*\$B\$7	=SUM(SES6:E19)	=(C19*'Cost inputs configu	=(C19*'Cost inputs configu	=(C19*Cost inputs configu	=(C19"Cost inputs configu	=(C19"Cost inputs config	gut=(C19*'Cos
0 3,5	='C:\Users\User\Dropbox\Pr	=1-C20	=AVERAGE(C19:C20)*SB\$7	=SUM(SES6:E20)	=(C20**Cost inputs configu	=(C20* Cost inputs configu	=(C20* Cost inputs configu	=(C20*Cost inputs configu	=(C20" Cost inputs config	gut=(C20*Cos
3,75	=C:\Users\User\Uropbox\Pr	=1-C21	=AVERAGE(C20:C21)*5857	=SUM(SES6:E21)	=(C21* Cost inputs configu	=(C21* Cost inputs configu	=(C21* Cost inputs configu	(C21-Cost inputs configu	=(C21=Cost inputs config	gut=(C21*Cos
2 4	=C:\Users\User\Dropbox\Pr	=1-C22	=AVERAGE(C21:C22)-5857	=SUM(5E56:E22)	=(C22* Cost inputs configu	=(C22* Cost Inputs configu	=(C22* Cost inputs configu	(C22*Cost inputs configu	(C22* Cost inputs config	gut=(C22*C05
4,23	= C: (Users) User(Dropbox (Pr	=1-C23	=AVERAGE(C22:C23) 5B57	=5010(\$5\$6:524)	=(C23* Cost inputs configu	=(C23* Cost inputs configu	=(C23* Cost inputs configu	=(C24*Cost inputs configu	=[C23* Cost inputs config	gur=(C23*C05
5 4 75	-'C-\Lisers\Liser\Dropbox\Pr	-1.025	-AVERAGE(C24-C25)*\$8\$7	-SUM(\$E\$6-E25)	=(C25*'Cost inputs configu	-IC25*'Cost inputs configu	-IC25*'Cost inputs configu	-IC25*'Cost inputs configu	-(C25*'Cost inputs config	aut=(C25*'Cos
6 5	='C:\Users\User\Dronbox\Pr	=1.026	=AVERAGE(C25*C26)*\$8\$7	=SUM(\$E\$6-E26)	=(C26*'Cost inputs configu	=(C26*'Cost inputs configu	=(C26*'Cost inputs configu	=[C26*'Cost inputs configu	=[C26*'Cost inputs config	aut=(C26*'Cot
7 5.25	='C:\Users\User\Dropbox\Pr	=1.C27	=AVERAGE(C26:C27)*\$8\$7	=SUM(\$E\$6-E27)	=(C27*'Cost inputs configu	=(C27*'Cost inputs configu	=(C27*'Cost inputs configu	a =(C27*'Cost inputs configu	=[C27*'Cost inputs config	gut=[C27*'Cot
8 5.5	='C:\Users\User\Dropbox\Pr	=1-C28	=AVERAGE(C27:C28)*\$B\$7	=SUM(SES6:E28)	#(C28*'Cost inputs configu	=(C28*'Cost inputs configu	=(C28*'Cost inputs configu	a =(C28*'Cost inputs configu	=[C28*'Cost inputs config	rut =(C28*'Cos
9 5.75	='C:\Users\User\Dropbox\Pr	=1-C29	=AVERAGE(C28:C29)*\$B\$7	=SUM(\$E\$6:E29)	#(C29*'Cost inputs configu	=(C29*'Cost inputs configu	=(C29*'Cost inputs configu	=(C29*'Cost inputs configu	=(C29*'Cost inputs config	rut=(C29*'Cos
0 6	='C:\Users\User\Dropbox\Pr	=1-C30	=AVERAGE(C29:C30)*\$B\$7	=SUM(\$E\$6:E30)	=(C30*'Cost inputs configu	=(C30*'Cost inputs configu	=(C30*'Cost inputs configu	=(C30*'Cost inputs configu	=(C30*'Cost inputs config	gur=(C30*'Cos
1 6.25	='C:\Users\User\Dropbox\Pr	=1-C31	=AVERAGE(C30:C31)*\$B\$7	=SUM(\$E\$6:E31)	=(C31*'Cost inputs configu	=(C31*'Cost inputs configu	=(C31*'Cost inputs configu	=(C31*'Cost inputs configu	=(C31*'Cost inputs config	gue=(C31*'Cos
2 6,5	='C:\Users\User\Dropbox\Pr	=1-C32	=AVERAGE(C31:C32)*\$B\$7	=SUM(\$E\$6:E32)	=(C32*'Cost inputs configu	=(C32*'Cost inputs configu	=(C32*'Cost inputs configu	=(C32*'Cost inputs configu	=(C32*'Cost inputs config	gut=(C32*'Cos
3 6,75	='C:\Users\User\Dropbox\Pr	=1-C33	=AVERAGE(C32:C33)*\$B\$7	=SUM(\$E\$6:E33)	=(C33*'Cost inputs configu	=(C33*'Cost inputs configu	=(C33*'Cost inputs configu	=(C33*'Cost inputs configu	=(C33*'Cost inputs config	gut=(C33*'Cos
4 7	='C:\Users\User\Dropbox\Pr	=1-C34	=AVERAGE(C33:C34)*\$B\$7	=SUM(\$E\$6:E34)	=(C34*'Cost inputs configu	=(C34*'Cost inputs configu	=(C34*'Cost inputs configu	=(C34*'Cost inputs configu	=(C34*'Cost inputs config	gut=(C34*'Cos
35 7,25	='C:\Users\User\Dropbox\Pr	=1-C35	=AVERAGE(C34:C35)*\$B\$7	=SUM(\$E\$6:E35)	=(C35*'Cost inputs configu	=(C35*'Cost inputs configu	=(C35*'Cost inputs configu	=(C35*'Cost inputs configu	=(C35*'Cost inputs config	gut=(C35*'Cos
36 7,5	='C:\Users\User\Dropbox\Pr	=1-C36	=AVERAGE(C35:C36)*\$B\$7	=SUM(\$E\$6:E36)	=(C36*'Cost inputs configu	=(C36*'Cost inputs configu	=(C36*'Cost inputs configu	=(C36*'Cost inputs configu	=(C36*'Cost inputs config	gut=(C36*'Cos
ECL & ECC for 9 cat	tegories Optimal G(t)	Cost inputs configurations	Costs PM Costs CM 🛞			•				
leady								H	I III	+ 1009

Optimal G(t) different CM/PM cost ratios

Autobere (* 08	B 5-0-+		Sensitive analysis (ratio maintenance costs) replacement converter system - Excel		Mark Haring 🖽 — 🗗 🗙
	(and a second second				100
rile nome	insert. Page Layout	Pormolasi Data neview view Add-ins Help > Lei me what	ou want to do		E3 pune
$\int_{\text{Insert}} \sum_{\text{AutoSun}}$	Recently Financial Logical	Image Image <th< th=""><th>Tace Precedents Softwor Formulas Tace Dependents Calculate Now Calculate Now Calculate Now Calculate Sheet Calcul</th><th></th><th></th></th<>	Tace Precedents Softwor Formulas Tace Dependents Calculate Now Calculate Now Calculate Now Calculate Sheet Calcul		
	Furthered	total territoria	Canonicon		
a •	te <u>A</u> ∨ × []	ECL & ECC for 9 categories'IG6/ECL & ECC for 9 categories'IF6	F G H	1 1 1 1	·
1	-		Optimal G(t)		
2	G(t)	Ratio 2 Ratio 2.5 Ratio 3	Ratio 3.5 Ratio 4 Ratio 4.5	Ratio 5	Ratio 6
3	0	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie1='ECL & ECC for 9 ca	egoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categori	iet="ECL & ECC for 9 categoriet="ECL & ECC for 9 categori	er='ECL & ECC for 9 categories
4 5 6	0,25 0,5	"ECL & ECC for 9 categorie ="ECL & ECC for 9 categorie = "ECL & ECC & ECC for 9 categorie = "ECL & ECC	egorie ="ECL & ECC for 9 categorie = "ECL & ECC for	e =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categorie =ECL & ECC for 9 categories	e ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie
7	1	="FCL & ECC for 9 categories ="FCL & ECC for 9 categories ="FCL & ECC for 9 ca	egories="FCL & ECC for 9 categories" FCL & ECC for 9 categories="FCL & ECC for 9 categories	HE = FCL & ECC for 9 categorie = FCL & ECC for 9 categori	et ='FCL & FCC for 9 categories
8	1,25	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egories='ECL & ECC for 9 categories='ECL & ECC for 9 categories='ECL & ECC for 9 categories	te = "ECL & ECC for 9 categorie = "ECL & ECC for 9 categori	et ='ECL & ECC for 9 categories
9	1,5	='ECL & ECC for 9 categories='ECL & ECC for 9 categories='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	te ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	et='ECL & ECC for 9 categories
10	1,75	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	e ='ECL & ECC for 9 categorie:
11	2	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	et='ECL & ECC for 9 categorie:
12	2,25	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie	e ='ECL & ECC for 9 categorie
13	2,5	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie	er='ECL & ECC for 9 categories
14	2,75	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie = ECL & ECC for 9 categorie = ECL & ECC for 9 categorie	e ='ECL & ECC for 9 categorie:
15	3	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egories = 'ECL & ECC for 9 categories = 'ECL & ECC for 9 categories = 'ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie	e:='ECL & ECC for 9 categorie:
16	3,25	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie = 'ECL & ECC for 9 categorie = 'ECL & ECC for 9 categorie = 'ECL & ECC for 9 categori	Ie = ECL & ECC for 9 categorie = ECL & ECC for 9 categorie	et='ECL & ECC for 9 categorie
17	3,5	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egories='ECL & ECC for 9 categories='ECL & ECC for 9 categories='ECL & ECC for 9 categori	let="ECL & ECC for 9 categoriet="ECL & ECC for 9 categorie	et='ECL & ECC for 9 categories
18	3,75	="ECL & ECC for 9 categories ="ECL & ECC for 9 categories ="ECL & ECC for 9 ca	egorier="ECL & ECC for 9 categorier="ECL & ECC &	let=ECL & ECC for 9 categoriet=ECL & ECC for 9 categorie	et='ECL & ECC for 9 categories
19	4	= ECL & ECC for 9 categories = ECL & ECC for 9 categories = ECL & ECC for 9 ca	egoner=ELL & ELC for 9 categorier=ELL & ELC for 9 categorier=ELL & ELC for 9 categorier	Tel = ECL & ECC for 9 categoriel = ECL & ECC for 9 categorie	er=ECL & ECL for 9 categories
20	4,23	= ECL & ECC for 9 categoriel = ECL & ECC & ECC for 9 categoriel = ECL & ECC &	egoriel = ECL & ECC for 9 categoriel = ECL & ECC for 9 categoriel = ECL & ECC for 9 categoriel	ter = ECL & ECC for 9 categoriel = ECL & ECC for 9 categorie	er=ECL & ECC for 9 categorie
22	4,5	= ECL & ECC for 9 categories = ECL & ECC for 9 categories = ECL & ECC for 9 categories	egones - ECL & ECC for 9 categories - ECL & ECC for 9 categories - ECL & ECC for 9 categories	res = ECL & ECC for 9 categories = ECL & ECC for 9 categories	et="ECL & ECC for 9 categories
22	4,73 E	For a second stategories = Second Se Second Second Seco	egories - ECC & ECC for 9 categories - ECC & ECC for 9 categories - ECC & ECC for 9 categories	res - ECC & ECC for 9 categories - ECC & ECC for 9 categories	er= ECC & ECC for 9 categorie
24	5.25	"FCI & ECC for 9 categories "ECI & ECC for 9 categories "ECI & ECC for 9 categories	egories = FCL & ECC for 9 categories = FCL & ECC for 9 categories = FCL & ECC for 9 categories	iet="FCL & ECC for 9 categoriet="FCL & ECC for 9 categoriet	et="FCL & ECC for 9 categories
25	5.5	='FCI & FCC for 9 categoriet='FCI & FCC for 9 categoriet='FCI & FCC for 9 ca	egorie = FCI & FCC for 9 categorie = FCI & FCC for 9 categorie = FCI & FCC for 9 categorie	let='FCI & FCC for 9 categorie ='FCI & FCC for 9 categori	er='FCI & FCC for 9 categories
26	5.75	='ECL & ECC for 9 categoriel ='ECL & ECC for 9 categoriel ='ECL & ECC for 9 ca	egoriet='ECL & ECC for 9 categoriet='ECL & ECC for 9 categoriet='ECL & ECC for 9 categoriet	iet="ECL & ECC for 9 categoriet="ECL & ECC for 9 categori	et='ECL & ECC for 9 categorie
27	6	='ECL & ECC for 9 categoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 ca	egoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categoriel	ies='ECL & ECC for 9 categories='ECL & ECC for 9 categori	et='ECL & ECC for 9 categories
28	6.25	='ECL & ECC for 9 categoriel ='ECL & ECC for 9 categoriel ='ECL & ECC for 9 ca	egoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categoriel='ECL & ECC for 9 categori	ter='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	et='ECL & ECC for 9 categorie
29	6.5	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egorie = 'ECL & ECC for 9 categorie = 'ECL & ECC for 9 categorie = 'ECL & ECC for 9 categorie	ter="ECL & ECC for 9 categories="ECL & ECC for 9 categories	et='ECL & ECC for 9 categories
30	6,75	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egories='ECL & ECC for 9 categories='ECL & ECC for 9 categories='ECL & ECC for 9 categories	ter="ECL & ECC for 9 categorie ="ECL & ECC for 9 categori	et ='ECL & ECC for 9 categories
31	7	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categories ='ECL & ECC for 9 categori	ies='ECL & ECC for 9 categories='ECL & ECC for 9 categories	e ='ECL & ECC for 9 categorie
32	7,25	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ies='ECL & ECC for 9 categories='ECL & ECC for 9 categories	e ='ECL & ECC for 9 categorie:
33	7,5	='ECL & ECC for 9 categories ='ECL & ECC for 9 categories ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	e ='ECL & ECC for 9 categorie:
34	7,75	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie	e ='ECL & ECC for 9 categorie
35	8	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie	e= = 'ECL & ECC for 9 categorie:
36	8,25	='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 ca	egorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categorie ='ECL & ECC for 9 categori	ie = ECL & ECC for 9 categorie = ECL & ECC for 9 categorie	er='ECL & ECC for 9 categorie
	ECL & ECC for 9 categories	Optimal G(t) Cost inputs configurations Costs PM Costs CM			•